A DESCRIPTIVE GRAMMAR OF DESANO

by

Wilson de Lima Silva

A dissertation submitted to the faculty of The University of Utah in partial fullfillment of the requirements for the degree of

Doctor of Philosophy

Department of Linguistics

University of Utah

August 2012

Copyright © Wilson de Lima Silva 2012

All Rights Reserved

The University of Utah Graduate School

STATEMENT OF DISSERTATION APPROVAL

The dissertation of	Wilson de Lima Silva	
has been approved by the following supervisory committee members:		
Lyle Campbell	, Chair	5/29/2012
		Due Approved
Mary Ann Christison	, Member	6/14/2012
		Date Approved
Edward Rubin	Member	5/8/2012
	,	Date Approved
Snike Gildea	Member	5/24/2012
Spine Onder	,	Date Approved
Kristine Stenzel	. Member	5/7/2012
		Date Approved
and by Edwar	rd Rubin	, Chair of
the Department of Linguistics		

and by Charles A. Wight, Dean of The Graduate School.

ABSTRACT

This dissertation provides a linguistic description and analysis of Desano, an endangered Tukanoan language of the Vaupes region of Brazil. Much valuable knowledge would be lost if this language were to become extinct without documentation. Several of the Tukanoan languages in the upper Amazon are highly endangered. The Eastern Tukanoan people are famous for their linguistic exogamy and 'obligatory' multilingualism; there are some twenty languages in the region whose speakers must marry someone who speaks a different language.

There are a number of linguistic traits in Desano described in this dissertation which are of particular interest to linguists in general, because they are rare in the world's languages, and they stand to contribute much to our understanding of the full range of possibilities in human grammar (and consequently also of some of the limits of human cognition). These include: nasal harmony (in phonology), the noun classifier system (in morphology), and the evidential system (in the interface of morphology-syntaxsemantics). The dissertation begins with an introduction of the Desano people and their language; including sociolinguistic information and some historical background. The second chapter presents a phonological description. It then proceeds with a description of the parts of speech in Desano and the characterization of the 'word' in Desano, in Chapter 3. The bulk of the dissertation is devoted to the morphology. This dissertation provides a reasonably comprehensive description and documentation of Desano, one of the most endangered Tukanoan languages. The descriptions of unusual typological traits (i.e., nasal harmony, evidentiality, verb serialization, etc.) contribute to the general linguistic scholarship in significant ways, since accurate accounts of the traits mentioned are of considerable interest for linguistic typology and theory generally. To my parents Wilson and Socorro with love.

TABLE OF CONTENTS

ABS	TRACT	iii
LIST	OF FIGURES	viii
LIST	OF TABLES	ix
ABB	REVIATION LIST	xi
ACK	NOWLEDGMENTS	xiv
Chap	oters	
1	THE DESANO PEOPLE AND THEIR LANGUAGE	1
	 1.1 Goals of the Dissertation. 1.2 A Brief History of the Research. 1.3 Methodology and Corpus. 1.4 Typological Profile of Desano. 1.5 A Brief History of Contact. 1.6 The Vaupés Region of Northwest Amazonia. 1.7 The Languages of the Vaupés and the Language Contact Situation. 1.8 The Tukanoan Languages. 1.9 The Desano People. 1.10 The Desano Language: Sociolinguistic Situation. 1.11 Previous Work on Desano. 1.12 Language Maintenance and Educational Projects. 	1 2 4 5 7 9 12 14 15 18 20 27
2	PHONETICS AND PHONOLOGY	30
	 2.1 Introduction. 2.2 Segmental Phonology. 2.3 Suprasegmental Phonology. 2.4 Morphophonemic Processes. 2.5 Notes on Orthography. 	30 31 57 79 83
	2.6 Summary	

3	PARTS-OF-SPEECH	85
	3.1 Introduction	85
	3.2 The 'Word' in Desano	86
	3.3 Parts of Words: Roots and Formatives	88
	3.4 Parts-of-Speech: An Overview	89
	3.5 Summary	103
4	NOMINAL MORPHOLOGY	104
	4.1 Introduction	104
	4.2 Structure of Noun Roots	104
	4.3 Types of Nouns	105
	4.4 Pronouns and other Pro-Forms	139
	4.5 Other Nominal Morphemes	154
	4.6 Noun Phrase Structure	175
	4.7 Summary	201
5	VERBAL MORPHOLOGY	203
	5.1 Introduction	203
	5.2 Structure of Verb Roots	203
	5.3 Verb Classes	206
	5.4 Serial Verb Construction	234
	5.5 The Verb Morphology	244
	5.6 Summary	269
CON	NCLUSION	271
App	endices	
A: S	URVEY OF LANGUAGES SPOKEN IN THE DESANO COMMUNITIES	273
B: T	EXT 1 - THE HUNTER AND THE MONKEY	279
C: T	EXT 2 - COLLECTING MATERIALS TO MAKE ORNAMENTS	286
REF	ERENCES	290

LIST OF FIGURES

1.1	The Vaupés Region in the Brazil-Colombia border1	1
2.1	Formant plots of the six vowels of Desano	4
2.2	Spectrograms for the words [a ^h siri] 'hot' and [we ^h ko] 'parrot'4	.8
2.3	Spectrograms for the words [gobe] 'hole' and [yeba] 'ground'	.9
2.4	Spectrograms of [kore] 'before' and [ko ² re] 'to wait'	3
2.5	Spectrograms for (C) V_1V_2 sequences	9
2.6.	Spectrograms for (C) V_1V_1 sequences	0
2.7	Prominent pitch movement	6
2.8	Vowel length in Desano words	8
4.1	General class markers	06

LIST OF TABLES

1.1	Languages of the Vaupés linguistic area
1.2	Desano speakers in the communities in Brazil
2.1	Desano vowel inventory
2.2	Desano consonant inventory
2.3	Desano vowel phonemes
2.4.	Illustration of desano vowels and formant measurements
2.5.	Desano vowel contrasts in oral contexts
2.6.	Desano vowel contrasts in nasal contexts
2.7	Vowel combinations
2.8	Plain and laryngealized roots
2.9	Nasal allophones of consonants and vowels
2.10	Vowel length in Desano words in seconds
2.11	Desano phonemes and nasal realization74
2.12	Desano phonemes and graphemes
4.1	Subcategories of animate nouns
4.2.	Hierarchy of animates
4.3	Unsuffixed inanimate noun roots
4.4	Grammaticalization continuum of class markers
4.5	Nominalization resulting in animate nouns

4.6	Derived inanimate nouns	
4.7	Desano personal pronouns	140
4.8	Interrogative pro-forms in Desano	149
4.9	Nominal morphology	
4.10	Possessive pronouns	
5.1	Template of verbal morphology	
5.2	Clause modality	
5.3	Desano evidentials	
5.4	Types of evidence	
5.5	Markers of personal agreement	

ABBREVIATION LIST

1	first person
2	second person
3	third person
ABSTR	abstract
ADD	additive
ADMON	admonition
ADV	adverbializer
ADVER	adversative
AN	animate
ASP	aspectual
AUG	augmentative
CEL	celerative
CLS	classifier
COM	comitative
COMPL	completive
CONCR	concrete
CONJ	conjunction
CONT	continuative
CONTR	contrary
DEIC	deictic
DEM	demonstrative
DEON	deontic
DES	desiderative
DIM	diminutive
DIREC	direction
DIST	distal

DUB	dubidative
DUR	durative
EMPH	emphatic
EVID	evidential
EXCL	exclusive
EXIST	existential
EXRT	exhortative
F	feminine
FOLK	folklore
FRUST	frustrative
HSAY	hearsay
IMP	imperative
IMPERF	imperfect
IN	inanimate
INCEPT	inceptive
INCL	inclusive
IND	individualizer
INSTR	instrument
INTER	interrogative
INTERJ	interjection
LOC	locative
M	masculine
MAN	manner
MOD	modality
MODIF	modifier
МОТ	motion
MOV	movement
N	noun
NEG	negation
NP	noun phrase
NOM	nominalizer

NON	non-
NUM	number
OBJ	object
PARTIC	participial
PERF	perfect
PERM	permission
PL	plural
POSS	possessive
PREDIC	prediction
PROX	proximal
QUANT	quantifier
QUOT	quotative
RC	relative clause
REAS	reason
REF	referential
RESUL	resultative
SG	singular
SOL	solitary
SPECUL	speculative
SUBJ	subject
ТМР	temporal
TOTAL	totalizer
VBLZ	verbalizer

ACKNOWLEDGMENTS

This dissertation would not be possible without the help of so many people in so many ways. First, I want to thank the Desano people for teaching me their language, for sharing their knowledge and for having me in their homes and in their communities (special thanks to Frank C. Matos, Jacinto Castilho and Macelino Massa).

There have been many people who have helped shape my academic career. First, I extend my deepest gratitude to my adviser, Professor Lyle Campbell, who put a tremendous amount of work into advising me and helping me in many ways. Not only was he readily available for me - as he so generously is for all of his students - but he always read and responded to the drafts of each chapter of my work more quickly than I could have hoped. His continuous advice, encouragement, patience, and friendship were fundamental in sustaining me through this process. I would also like to thank the other members of my committee. Mary Ann Christison, Edward Rubin, Spike Gildea and Kristine Stenzel have all made extensive comments on this work and gave supportive suggestions. I would like to acknowledge that Kristine Stenzel also works on Tukanoan languages and, besides providing comments and suggestion on this work, has also shared with me her own fieldwork experiences in the Vaupés Region, which helped me to carry out successful fieldwork with the Desano people. The participation of each of my committee members has led to great improvements in this dissertation.

Thank you also to Joyce McDonough, at the University of Rochester, who helped me as much as my official committee members with detailed feedback on many aspects of this work. I also would like to thank Keren Rice, who received me as a visiting student for one semester at the University of Toronto, and whose seminar was extremely valuable for me to understand some aspects of the Desano phonology. Of course, despite all the assistance provided by these professors, I alone remain responsible for the content of this dissertation, including any errors or omissions which may unwittingly remain.

I am also thankful to people at other institutions. Professor Dulce Francesquini, for sparking my interest on the study of Amazonian languages when I was an undergraduate student at the Federal University of Amazonas; to Professor Paulo Renan da Silva, my undergraduate advisor, who encouraged me to pursue a career in linguistics; and, to Professor Aryon Rodrigues, from the National University of Brasília (UnB), who suggested and encouraged me to document a language in the Upper Rio Negro Region. Thanks are also due to the friends and faculty of the Department of Linguistics at the University of Rochester, where I 'officially' started my career in linguistics.

The field research and graduate training that resulted in this work would not be possible without financial assistance from several sources. I am grateful to the University of Utah for subsidizing my studies with the following scholarships: The Department of Linguistics Wick R. Miller Scholarship and Cathy Miller Horiuchi Scholarship; the College of Humanities Steffensen Cannon Scholarship; and the University of Utah Graduate Research Fellowship. I would like to thank the Endangered Language Fund for its financial support in the initial stages of my research. I am also grateful to The Hans Rausing Endangered Language Documentation Project (Field Trip Grant 155); the

xv

National Science Foundation (Doctoral Dissertation Improvement Grant, NSF/BCS-0756067); and to UNESCO/Museu do Índio for providing funding for the main fieldwork trips of the last five years. I also want to thank FOIRN (Federação das Organizações Indígenas do Rio Negro) for the logistic support to travel in the area.

Many friends and colleagues have played important role in my making it successfully through graduate school. Carolina Coelho, Katherine Matsumoto-Gray, Randall Gess, and Rosemary Bean De Azcona have been true friends and colleagues with whom I've discussed many aspects of my work. Many other 'nonlinguist' friends have been supportive during my graduate studies: David Loudon, Michele Villalobos, Justin Naylor, and Tim Hunger have been great friends and listeners; and my friend Thomas McBurney, for hosting me for while I was in Toronto for one semester. I also want to express my gratitude to the staff of the Department of Linguistics at the University of Utah, especially Shantel Baldonado de Arraiz and Kacey Campbell, who were extremely helpful and patient while guiding me through the bureaucratic aspects of graduate school.

I would like to thank family. My mom Socorro and dad Wilson for all the support and inspiration they have given me over the years. Benjamin Lawrance, for the endless patience and support you have given me.

xvi

CHAPTER 1

THE DESANO PEOPLE AND THEIR LANGUAGE

1.1 Goal of the Dissertation

The goal of this dissertation is to provide an in-depth linguistic description and analysis of Desano, an Eastern Tukanoan (ET) language of the Vaupés Region of Brazil. Study of ET languages is important for advancing understanding of various aspects of linguistic theory. The ET languages exhibit important typological traits little known elsewhere in the world, such as a complex system of noun classifiers, a unique evidentiality system, and nasal harmony.

This dissertation contributes to the linguistic scholarship in three areas: (1) it provides a reasonably comprehensive description and documentation of Desano, one of the most endangered ET languages of the region; (2) it contributes to the language community by providing material to aid in the communities' language revitalization efforts; and, (3) it contributes to general linguistic scholarship in significant ways, by furthering understanding of: (i) typological understanding of languages; (ii) theoretical linguistics, providing evidence on the basis of which theoretical claims will be tested; (iii) historical linguistics, providing material needed for the study of language change, language contact, and the genetic relationships among languages of the Vaupés area; and (iv) the study of ET languages, contributing to a growing body of work on languages of the Vaupés area.

1.2 A Brief History of the Reseach

The present work is part of my ongoing investigation of Desano. I first made contact with the Desanos during the summer of 2007.¹ Desano leaders and community members had long searched for a linguist to help them with language documentation and revitalization activities. I was received enthusiastically by the group. This first field trip was important in establishing a relationship with the leaders of the group and discussing the viability of conducting linguistic research with them. The Desano leaders, the directors of FOIRN (*Federação das Organizações Indígenas do Rio Negro*), the official indigenous organization with jurisdiction in the region, and I agreed to a long-term commitment between myself and the Desano group. On the same occasion, FOIRN asked me to supervise their linguistic efforts with one of their projects. One of the projects for which I was the linguistic superviser was the '*Projeto Bayawi*'. The goal of this project was to provide activities in which the young generation of Desanos would be exposed (immersed) to traditional activities of the group.

In 2008, I returned to the field and organized the first meeting with Desano teachers, community leaders and community members to discuss a plan in order to start a collaborative project for the documentation and revitalization of the language. During this fieldwork, I also did preliminary linguistic research to ascertain the viability of documentation, as well as a project to undertake a linguistic analysis of the language. I

¹ The field trip was supported by a small grant from *The Endangered Language Fund*.

organized multiple linguistic workshops in the communities in order to develop an orthography for use by the Desano. During these workshops, participants also received preliminary training in language documentation. Fieldwork was conducted in the Desano communities on the Brazilian side of the Papuri River. Community members also created the "Desano Committee," comprised of six Desanos who would advocate on behalf of the group and serve as intermediaries between me and the community members in order to ascertain that the project activities in the Desano Committee chose two Desano speakers to work with me as 'Indigenous Research Assistants' (IRA). These IRAs have received training in language documentation and basic linguistic analysis, and they have worked and continue to work closely with me and with the communities.²

Since 2010, the two IRAs have been actively working on transcription, translation and collection of material in the communities – these materials have been parsed and translated. Museu do Índio and UNESCO secured support for the work with the Desano communities in Brazil for the years 2011 and 2012. During three years of work in the Desano communities in Brazil, the project team has recorded, transcribed, and translated forty-five texts in a variety of genres (stories, tales, myths, dialogues, songs, poems, etc.), comprising a total of more than twenty-five hours of audio and video recorded materials. In addition, the project team has made audio recordings of all items for a digital Desano dictionary. It has prepared booklets to be used in the schools as aids for the communities' language revitalization efforts.

² The work conducted during the years of 2008 and 2009 was funded by NSF/DEL (Doctoral Dissertation Improvement Grant, BCS-0756067) and *The Hans Rausing Endangered Languages Project* (ELDP Field Trip Grant, FTG 055)

One of the most significant results of the project, however, is this dissertation, which will be translated into Portuguese. Other important results involve consultation and active participation of community members regarding the goals and plan of work for the ongoing documentation project among the Desano communities in Brazil with the support of Museu do Indio. The 'Desano Committee' and I have continued to organize annual meetings and linguistic workshops with Desanos from communities on the Tiquié and Papurí rivers in Brazil. These meetings and workshops have always been well attended. In the workshops, we discuss issues regarding language revitalization and language policy, we also define the goals and discuss the methodologies of the ongoing project activities in the communities. Concrete accomplishments include resolving a number of orthographic controversies (particularly the representation of intervocalic flapping of /d/ and the high central vowel /i/, and a sociolinguistic survey of all the Desano communities in Brazil, which showed that the number of fluent speakers of Desano is very small: only 150 individuals speak the language fluently in this region (although the number of fluent speakers in Colombia is assumed to be larger).

1.3 Methodology and Corpus

Data for this dissertation were collected during my own fieldwork with native speakers of Desano. This fieldwork employed traditional methods: elicitation of data; audio and video recording of elicitation and of naturally occurring speech in numerous genres (narratives, including oral history, traditional tales, dialogue, conversation, recipes and direction-giving, jokes, songs, etc.); and transcription and analysis of these data in the field with the assistance of native speakers. The majority of the recordings of naturally occurring speech were collected largely by the trained community members to ensure the "naturalness" (i.e., authenticity) of the data.

Linguistic questionnaires (lexical and morphosyntactic) were also employed.³ One of the questionnaires used was a lexical questionnaire of about 1,000 words based on Kaufman and Berlin's (1987) questionnaire for South American languages and on Kaufman's (1967) questionnaire for Mayan and other Mesoamerican languages.

The approach I took to the description and analysis was worked out inductively, based on texts of natural occurring speech and from elicited utterances (based on these texts). Thus, my goal was to describe Desano in its own terms, in a user-friendly fashion, employing terminology from linguistic typology.

1.4 Typological Profile of Desano

The basic (or preferred) constituent order in Desano is SV for intransitive sentences and SOV for transitive utterences. The language exhibits a nominative-accusative alignment system; generally the accusative is marked by the suffix *-re* (this suffix has multiple functions and its cognate is a current topic of investigation in other ET languages, see Stenzel 2008). Both pronouns and other referential NPs can be omitted in discourse when they are recoverable from contexts (usually through agreement suffixes attached to the verb).

Desano is agglutinative and exclusively suffixing (except for the possessive marker *yaa*, which functions as a clitic and can be attached at the beginning of nouns). Verbs and nouns are the two major lexical classes (parts of speech). Adjectival notions

³ These included personal/social information from each participant (all relevant metadata).

are generally coded by either qualitative verbs or nominal roots. The verbal morphology is highly complex, with verb suffixes encoding tense, number, person, gender, mood, and evidentialilty. The language also has complex verb serialization, which can take up to four verbal roots. Like other Tukanoan languages (cf. Gomez-Imbert 2007), Desano exhibits extensive nominal classification, including noun classes and noun classifiers. All Desano nouns fall into one of four classes: animate plural, masculine singular animate, feminine singular animate, and inanimate. Membership is generally predictable on semantic grounds, except for the split of animals between the two animate genders (i.e., some animals are treated as human-like nouns). The evidential system of Desano is highly complex. In Desano, evidentiality is a grammatical category that is obligatorily marked on the verb root. In a 'declarative' sentence in Desano, besides including information such as 'tense/aspect', 'person', 'gender', and 'number', the speaker has to indicate how he or she knows the information.

Phonologically, Desano exhibits a classic Amazonian six-vowel system (including a central high vowel), with contrastive nasality, and an inventory of nine consonants (the status of the glottal stop (?) and the glottal fricative (*h*) as full consonant segments is in question and is under investigation (Silva, *to appear*). Desano also exhibits nasal harmony that occurs on the word level (i.e., nasality can spread from the root morpheme to affixes). This suprasegmental nasalization is common in many Tukanoan languages (Barnes 1999: 211-212; Barnes 1996; Gomez Imbert and Kenstowicz, 2000; Stenzel, 2007). Desano exhibits a two-level tone system. Desano verbal and nominal roots may exhibit either HH, HL, or LL tone patterns, and a number of verbal affixes bear lexical high tone. Tone spreading seems to be limited, affecting only word-final inflectional suffixes. Desano shows multiple lexical and grammatical homonymy.

1.5 A Brief History of Contact

It is believed that before the arrival of the Europeans conquerers in Brazil in 1500 the number of indigenous languages spoken in Brazil was at least double of what are spoken today (Rodrigues 2002). As for the number of languages spoken today, figures vary somewhat. Rodrigues (2005) lists approximately 180; ISA (Instituto Socioambiental) lists more than 180; and Moore (2006) talks about the existence of about 160 indigenous languages (some consider Moore 2006 the more accurate count, cf. Stenzel 2005). The count of the indigenous groups in Brazil that speak these languages also varies somewhat; Moore (2006) counts 220 groups, ISA (Instituto Socioambiental) lists 222, and CIMI (Conselho Indigenista Missionário) gives 235.⁴ Nearly all that is known about Desano history is based on oral traditions of the group. Buchillet (1990b) and Wright (1987, 1991) present some history about the first contacts with the Eastern Tukanoan groups. Ramirez (1997:12-14) also presents a brief history of this contact and summarizes it in five phases:

 Slavery (1739-1760). During this period, the Portuguese invaded the Upper Rio Negro region to capture Indians and sell them as slaves.
 Approximately 20,000 indigenous people were taken from their villages and sold as slaves during that time.

⁴ Instituto Socioambiental is a nonprofit organization which directs several projects in indigenous communities of the region; CIMI is a nonprofit organization vinculated to the Catholic Church which supports (educational) projects in indigenous communities.

- 2. Creation of new indigenous villages (1761-1829). During this period, the Portuguese administration created villages in 'strategic locations' and started to place the indigenous people there, taking them from their traditional communities to these new villages. According to Ramirez, this process explains why some of the groups are spread out in the region.
- 3. *Commerce and other governmental programs* (1830-1920). It is during this period that the first missionaries arrived in the region, motivated by the governmental program 'civilização e catequese' (civilization and conversion). It was also during this period that the exploitation of natural rubber in the Amazon reached its peak. Many indigenous people were taken from their villages and forced to work as collectors of rubber. It was also during this time that the first ethnographic reports were made by scientists who traveled to the region (for example, Wallace 1858; and Koch-Grünberg 1906).
- 4. The Salesian Missions (starting from 1916). The Salesians arrived in the Upper Rio Negro and slowly freed the indigenous people from oppression and slavery. However, the missionaries forced the indigenous children to go into boarding schools, where they were forbidden to speak their native languages. They were allowed to speak only in Portuguese. Many groups (like the Desano) did not resist this process, which resulted in the end of the traditional long houses, and in abandonment of traditional rituals and other cultural practices.

5. *Modern society* (starting from 1960). Nowadays, many of the indigenous people in the region have a different lifestyle. Many of them have moved to villages (missions), where they have access to electricity and live in individual houses, and the children go to school and receive an education in Portuguese. In the case of the Desano in Brazil, for example, they are abandoning their traditional villages and moving to other places, like Iauaretê and São Gabriel da Cachoeira; as a consequence, they abandon their native language.

1.6 The Vaupés Region of Northwestern Amazonia

The Vaupés Region is named after the river with the same name (Vaupés in Colombia, Uaupés in Brazil). The Vaupés River is one of the largest tributaries of the Rio Negro. It starts in the Colombian territory and part of it serves as the border between Brazil and Colombia. The Vaupés Region, as defined here, is an area that includes the main tributaries to the Vaupés (the Papurí and Tiquié Rivers) and also the Apaporis, Mirití-Paraná and Pirá-Paraná Rivers, as shown in the marked area in the map in Figure 1.1. The first records (travel letters/journals/reports) about the indigenous people of the Vaupés date from 1759.

These early reports refer to the people of the Vaupés as "Vaupés Nations" [Nações Uaupés] due to the homogeneity of the culture and similarities among the languages spoken in the region (cf. Brüzzi A. da Silva 1977:26).

The Vaupés region of the Brazilian-Colombian border is characterized by numerous cultural traits shared by indigenous groups that speak the languages of three different language families (Arawakan, Nadahup [or Makuan] and Tukanoan). This region is famous for extensive multilingualism and linguistic exogamy (a spouse must marry someone who speaks a different language), where people know at least two languages and often several. Consequently, the language-contact situation has had great influence in language change among the languages of the region, and the linguistic exogamy has been a topic of much interest in the literature. For example, discussion of the multilingualism and the exogamous marriage practices in the Vaupés region are found in Aikhenvald (2002), Ardila (1989), Chernela (1982, 1989, 1993), Correa Rubio (1984, 1997), Gomez-Imbert (1991, 1993, 1999), González Ñáñez (2004), S. Hugh-Jones (1993), Jackson (1974, 1976, 1983, 1984, 1988), Sorensen (1967, 1985) and Stenzel (2005, 2006), among others.

The region is also known for other cultural traits shared among the different groups – specific forms of hunting, fishing, ceremonial greeting, shamanism, myths, discourse forms, etc. General descriptions and discussions about the cultural practices of the Tukanoan people (and other groups) in the Vaupes region are found in Århem (1981, 1989, 1996), Brüzzi (1977), Buchillet (1983, 1985a, 1985b), FOIRN and ISA (2000), Giacone (1949), Goldman (1948), C. Hugh-Jones (1977, 1979, 1988), S. Hugh-Jones (1994a, 1994b, 1995, 2002), Jackson (1991, 1995), Koch-Grünberg (2005 [1909]), Meira (1991, 1993), Reichel-Dolmatoff (1971, 1996, 1997), Santos-Granero (2002), and Wright (2002).



Figure 1.1 The Vaupes Region in the Brazil-Colombia border⁵

⁵ Source: www.internationalrivers.org

1.7 The Languages of the Vaupés and the Language Contact Situation

The languages of the three language families that are usually recognized in the literature as belonging to the Vaupés Region are presented in Table 1.1.⁶ One of the first studies describing a contact situation that resulted in diffusion of traits from one language to another was reported by Gomez-Imbert (1996). She showed that children from Baniwa [Arawakan]–Kubeo [Tukanoan] marriages, with Kubeo as their mother tongue and Baniwa as their father's language carry over aspects of animal classification and change the conceptual scheme from one language to that of the other. Their system now reflects the Baniwa distinctions of animacy and gender for animals and the Tukanoan categorization of inanimate entities in terms of shape, carried over to animate entities.

Table 1.1 Languages of the Vaupés Linguistic Area

Language Families

EASTERN TUKANOAN

Bará, Barasana, Desano, Karapana, Kotiria (Wanano), Kubeo, Makuna, Pisamira,
Siriano, Retuarã, Taiwano, Tatuyo, Tuyuca, Tucano, Wa'ikhana (Piratapuyo) and Yuriti.
ARAWAKAN
Baniwa (Kurripako), Kawiyari, Tariana and Yukuna.
NADAHUP (or Makuan)
Hup, Kakua, Nukak and Yuhup.

⁶ See Stenzel (2004:20) for Eastern Tukanoan; Stenzel and Gomez-Imbert (2008) for Eastern Tukanoan and Arawakan; and Epps (2007) for Nadahup (Makuan) languages.

One of the best known cases of contact-induced structural change in the region is the Tariana (Arawakan)–Tukano (Eastern Tukanoan) contact, which was first reported by Aikhenvald (1996) and later extended and discussed in other studies (Aikhenvald 1999, 2001, 2002, 2003, etc.). Dixon and Aikhenvald (1999) presented a list of traits that characterize the languages (mostly Eastern Tukanoan) of the region as belonging to a sub-area within Amazonian Linguistic Area.⁷ Aikhenvald (1999) compares traits of the three unrelated language families in the 'Içana-Vaupés basin' (Arawakan, Eastern Tukanoan, and Nadahup [Makuan]) and concludes that there might be a direction of diffusion from 'East Tu[k]ano[an] to Tariana and to Makú' (p. 411). Some generalizations were presented by Aikhenvald (2002), who characterizes the area as being a case of multilateral diffusion in which languages have influenced one another. This generalization has been debated (see Meira and Gomez-Imbert 2005; Stenzel 2005; Stenzel and Gomez-Imbert 2008). Aikhenvald also points out that indirect diffusion (development of a new category or a new term for a category) is more common than direct diffusion (lexical borrowing or borrowing).

Epps (2005) describes the development of the evidentiality system in Hup as due to the influence of Eastern Tukanoan languages (primarily Tukano) in the area. Other shared traits that have been reported to be the results of diffusion from Tukano to Hup – and to some extent also to Yuhup and Dâw – are presented in Epps (2007). She lists phonological and morphosyntactic shared traits that give strong support for the Vaupés as a linguistic area.

⁷ See Franchetto and Gomez-Imbert (2003) for a review of Dixon and Aikhenvald (1999).

Stenzel and Gomez-Imbert (2008) investigate the contact situation between Arawakan (Baniwa [Kurripako] and Tariana) and Eastern Tukanoan languages, with special attention to the impact of Baniwa on Kotiria (Wanano). They identify four subareas in the region that have been the focus of investigation on language contact and diffusion: (a) on the Upper Cananarí and Upper Papurí Rivers, where there is contact among the Eastern Tukanoan languages Tatuyo, Barasana and Taiwano (Eduuria) and the Arawakan language Kawiyari; (b) on the Apaporis, with contact between Retuarā/Tanimuca (ET) and Yukuna (Arawakan); (c) in the Vaupés area between Mitú (Colombia) and Querarí (Brazil), where there is contact with diffusion of traits from Baniwa (Arawakan) to Kubeo (ET) as reported by Gomez-Imbert (1996); and, (d) on the Vaupés, Papurí and Aiarí Rivers, where there has been contact and diffusion from Tukano (Eastern Tukanoan) to Tariana (Arawakan) as reported extensively by Aikhenvald (2000).

1.8 The Tukanoan Languages

The Tukanoan language family has twenty languages still spoken today. Many Tukanoan languages are extinct. For example: Western Tukanoan: Macaguaje* (Kakawahe, Piojé), Teteté* (Eteteguaje) Ecuador, Colombia (possibly a dialect of Siona), Tama* (sometimes said to be perhaps a Koreguaje dialect); Eastern Tukanoan: Miriti* (Miriti-Tapuyo, Neenoá), Kueretú* (Cueretú, Coretú, Curetú), Arapaso* (Arapaço, Arapasso, Koneá) (cf. Campbell 2012a).

There have been a few attempts to provide a classification for the Tukanoan family (Sorensen 1969, Waltz and Wheeler 1972, Ardila 1993, Ramirez 1997, Barnes

1999, Barnes 2006).⁸ Stenzel (2004:21) compares the classifications available and points out that a conclusive classification of the Tukanoan languages is still needed. I follow Gomez-Imbert (2011), who states that the Tukanoan languages are divided in two main branches: Western and Eastern. ⁹ The Western branch consists of four languages: Koreguaje, Secoya, Siona and Orejón, spoken in areas of Colombia, Equador and Peru. The Eastern branch consists of sixteen languages: Bará, Barasana, Desano, Karapana, Kotiria (Wanano), Kubeo, Makuna, Pisamira, Siriano, Retuarã, Taiwano, Tatuyo, Tuyuca, Tucano, Wa'ikhana (Piratapuyo), and Yuriti.

The estimated number of people of all these Eastern Tukanoan languages combined is only 28,000. Tukano, the largest, has c.10,000 people (Stenzel 2004:20), and Pisamira is the smallest, with fewer than fifty people. As aforementioned, Eastern Tukanoan peoples are well known for their linguistic exogamy and 'obligatory' multilingualism.¹⁰

1.9 The Desano People

The Desano people call themselves **wĩrã** 'wind' or **ũmũrĩmãhsã** 'people of the day'. According to the Desano origin myth, the first Desano existed in spirit and lived in the 'sun's house' (*abewi'i*); they say they are the 'offspring of the sun'. The spirit traveled to the 'house of thunder' (*buhpuwi'i*), the spirit's grandfather; there it started to

⁸ Barnes (1999) follows the classification presented in Waltz and Wheeler (1972). See Franchetto and Gomez-Imbert (2003) for some problems with Barnes's classification.

⁹ See also Stenzel (2004:20) for the same classification following Gomez-Imbert and Kenstowicz (2000).

¹⁰ See Aikhenvald (2002); Chernela (1982, 2004); Correa Rubio (1997); Gomez-Imbert (1991); González Ñáñez (2004); S. Hugh-Jones (1993); Jackson (1974); Sorensen (1967); and Stenzel (2005) for a detailed description of the exogamous system and the multilingualism in the region.

blow smoke and created the first humans from this smoke. The spirit's grandfather (the thunder) gave him more tobacco to continue traveling and creating the Desano people. Then, the first Desano people traveled inside a giant snake, known as the *Transformation Canoe* (Canoa da Transformação), where all the other Tukanoans were also traveling. The 'transformation canoe' stopped at Ipanoré Falls (located in the Vaupés River, between the missions of Taracuá and Iauaretê), where they all disembarked and celebrated with dances. The order in which the groups disembarked from the canoe established the hierarchy within the Tukanoan peoples. The first ones to disembark were the Tukano, followed by the Desano, then the Piratapuyo (Wa'ikhana) people, etc. After the celebrations, the Desano traveled upriver. They settled on the right bank of the Vaupés River between Mitú (the capital of the *Departamento del* Vaupés, Colombia) and Querarí (a small mission village in Brazil) on the *Abiú, Timbó, Murutinga* creeks.¹¹ Some Desano people still live in these traditional locations and are considered to be the 'head' of the group by the Desanos who live in other areas.

Due to fights between the two main Desano clans, *Boreká* and *Dihputiro*, the Desanos spread out southward to other areas, to the Papurí and Tiquié rivers.¹² The traditional Desano communities are located at the headwaters of small streams or creeks in the area of these rivers (cf. Béksta 1988, Reichel-Doumatoff 1971). The Vaupés and Papurí Rivers form a natural border between Brazil and Colombia, and this is the main area where the traditional Desano communities are located (see Figure 1.2). The Boreká

¹¹ This myth was recounted by five Desano elders (Aparício Caiseiro, Marcelino Belles, Quintino Dias, and Francisco Malha) from the community *São Luiz do Virarí*, in the Virarí stream (Papurí River), Colombia. The text was transcribed and translated with the assistance of Frank Matos (my main Desano consultant, also from Colombia).

¹² Boreká and Dihputiro are also considered the two main dialects of Desano.

went to *Virarí Igarapé* (a stream that runs into the Papurí River, in Colombia); the Dihputiro went to *Igarapé Ingá* and *Igarapé Turi* (some of the streams that run into the Papurí River, in Brazil). Eventually, the Boreká started to migrate to other lands south of the Papurí, and ended up in the Tiquié River, in Brazil.¹³

Although the origin myths suggest that Desano is a member of the Tukanoan linguistic family, there are some alternative hypotheses in the literature. For example, it has been suggested that Desano was originally a member of the Arawakan linguistic family, but became 'tukanized' due to the exogamous marriage with Tukanos (cf. Béksta 1988, Koch-Grünberg 1909). These hypotheses are based solely on oral narratives.

Nowadays, the Desano people live in eighteen communities on the Papurí, Tiquié, and Vaupés rivers and their small tributaries in Brazil and Colombia. The Brazilian communities, starting in the Tiquié River area (and its tributary streams), are São João Batista, Santo Antônio, Cucura Manaus, Floresta, Sítio São Pedro, São Lourenço, Santa Rosa, São Sebastião, Urubu Lago, Tucandira, and Piracema. In most of these communities, speakers of Tukano, Yuhup, and Hup are also found. On the Papurí River, the communities are found on both the Brazilian and Colombian sides of the river, which is a natural border between the two countries.

The communities on the Papurí River, starting from downstream, are Turi, São João, Santa Cruz and Santa Marta in Brazil and Olinda, Piracuara, Montfort, São Luiz do Virari, and San José del Viña in Colombia. On the Vaupés River, the two Desano communities are found on the *Igarapé Abiu*, in Colombia. The Desano communities

¹³ This part of the Desano migration was told by Guilherme Fonseca, in January 2009, in the community Monfort, in the Papurí River, Colombia. The text was transcribed and translated with the assistance of Frank Matos, my Desano consultant.

range in size from those with a couple of houses and a total of ten to twenty inhabitants found on the Papurí River in Brazil, to those with ten Desano families, with some thirty inhabitants (São Luis do Virari), also on the Papurí River on the Colombian side. In Brazil, there are concentrations of Desano in a few nineteenth-century mission communities such as Iauaretê, located on the Vaupés River, near the confluence with the Papurí River, and Taracuá, near the confluence with the Tiquié River. These communities are located within the Alto Rio Negro Indigenous Area, which belongs to the municipality of São Gabriel da Cachoeira, a town of approximately 18,000 people. Approximately ninety percent of the population of this town is indigenous from different ethnolinguistic groups (Tukanoan, Arawakan, Nadahup, Yanomaman, and Tupían [speakers of Nheengatu]). A few Desano speakers can be found in São Gabriel da Cachoeira itself (although there are many ethnic Desanos who do not speak the language). Outside the Alto Rio Negro Indigenous Area, in the Médio Rio Negro Indigenous Area, there are two communities, Balaio and Sargento Botelho, where some Desano speakers are also found. These communities are located by the BR-307 road, which links São Gabriel da Cachoeira and Cucuí, a military district on the border with Venezuela.

1.10 The Desano Language: The Sociolinguistic Situation

Desano, like most of the languages of the region, is highly endangered. In order to determine the degree of endangerment of Desano, I consider two factors. The first is that many of the Desano people who still live in the traditional communities are abandoning their traditional language and switching to Tukano, the dominant indigenous language in the region (see also Stenzel 2005:507). The second is a question of migration from their

traditional communities to other towns or bigger villages where they abandon their traditional lifestyle and replace their language with Tukano and/or Portuguese (or with Spanish, in Colombia).

During my fieldwork trips, I found that the number of fluent speakers of Desano is relatively small (an estimate of 150 people in Brazil and about 300 in Colombia) when compared to the figures listed by other sources.¹⁴ An accurate figure of the number of Desano speakers is difficult to determine. The *Ethnologue* 's statistics for this area are outdated (Stenzel 2006). The Instituto Socioambiental¹⁵ lists 1,531 ethnic Desanos in Brazil, though FOIRN (Federação das Organizações Indígenas do Rio Negro) – the local indigenous organization – estimates that there are only 800 Desano people in Brazil – not all of whom identify with this ethnic group are speakers of Desano. COAMA (Consolidation of the Colombian Amazon) gives an estimate of 2,457 ethnic Desanos in Colombia.¹⁶

Table 1.2 shows that number of ethnic Desanos compared with the number of actual speakers of the language in the communities I visited in Brazil. It is important to mention that many Desanos live in urban centers (São Gabriel da Cachoeira and Manaus, Brazil). The numbers presented here refer to the people still living in the traditional communities, or, as in the case of Iauaretê, a mission villages near their traditional communities to which the Desanos who live on the Brazilian side of the Papurí River have migrated. The Desano people are considered one of the most 'disperse' groups in the region.

¹⁴ I visited four Desano communities in the Colombian side of the Papurí River: Olinda, Montfort, Sao Luiz do Virari and Piracuara.

¹⁵ www.socioambiental.org/pib/epi/uaupes/print_source.html

¹⁶ www.coama.org.co
River	Community	Ethnic Desanos	Desano speakers
	Cucura Manaus	3	3
	Piracema	12	12
Tiquié	Santa Rosa	9	9
River	Santo Antônio	8	2
	São João Batista	7	7
	São Luís	4	2
	São Sebastião	29	24
	Tucandira	3	3
	Urubu Lago	8	8
Vaupés	Iauaretê/Vaupés	62	35
River			

Table 1.2 Desano speakers in communities in Brazil.¹⁷

1.11 Previous Works on Desano

There are a few linguistic works available on Desano (many of them done by members of the Summer Institute of Linguistics); they are useful, but in general are not extensive and do not provide sufficient understanding of the most salient properties of the language. There is also a considerable number of anthropological works on different aspects of the Desano culture. In this section, I present a survey of the available linguistic work, followed by a summary of the anthropological works. For the linguistic

¹⁷ See Appendix A for a list of other languages spoken in the communities in the Tiquié River.

publications, I follow a chronological order, i.e., the year of publication. However, the anthropological works are presented following an alphabetic ordering by author.

1.11.1 Previous Linguistic Publications

The first published linguistic notes about Desano are the wordlists collected by Koch-Grünberg, which were published in *Anthropos* (between 1913 and 1916). Other wordlists were collected by the Salesian missionaries (cf. Stradelli 1910, Gianconne 1949, Da Silva 1961), and scholars such as Rivet et al. (1925). All these Desano wordlists were collected together with wordlists of other languages of the region (comparative lists). However, Koch-Grünberg's (1913, 1914, 1915) wordlists seem to be the most accurate regarding the (phonetic) transcription. He was consistent in transcribing nasalization and aspiration, two important phonological traits of the language.

The wordlist produced by Stradelli (1910) was transcribed using Portuguese orthography with some 'adaptations;' nasalization and aspiration are not marked consistently. The wordlist in Rivet (1925) also has some accurate information (although sometimes the same symbol is used to describe different segments). Giaconne (1949) is the least accurate wordlist (several inconsistencies can be found, for example different symbols are used for the same phoneme). Finally, the wordlist in Da Silva (1949) presents 139 items, and consistent transcription. This work was originally accompanied by an audio recording (on vinyl).¹⁸

The first linguistic works treating specific aspects of Desano were published in the late 1960s and early 1970s. Jonathan Kaye is responsible for the first academic works

¹⁸ Although I have been able to find a copy of the original book, I have not been able to find the accompanying audio material.

on the language that became available. In his master's thesis, Kaye (1965) presents a preliminary description of the phonology of Desano using two of the main theories in vogue at that time: the 'neo-Bloomfieldian' theory (American structuralism) and the newly developed 'generative theory' (Chomsky 1965). In his work, Kaye provides arguments in favor of generative theory as the best way to explain some phonological aspects of Desano. Kaye, however, acknowledges the limitation of the theory for explaining the data he has available. One of the main issues in his analysis was to treat as segmental properties what would later be considered as suprasegmental (e.g., nasalization). Kaye does not mention the tone system in the language, although he does mention stress. In the appendix of Kaye's thesis, we find a sample of Giacone's, Koch-Grünberg's and Rivet's wordlists parallel to his own wordlist. Kaye's (1968) article 'Nominalized Relative Clauses' offers an analysis to explain the nominalization process of verbal roots in light of the generative theory of the time. Kaye acknowledges that although that theory explains the nominalization of a limited set of verbs, it fails to provide an explanation for the process as a whole.

In his doctoral dissertation, 'The Desano Verb: Problems in Semantics, Syntax, and Phonology,' Kaye (1970) used the transformational grammar of Chomsky (1965) to explain some syntactic 'problems' in Desano, and uses Chomsky and Halle (1968) to explain some phonological 'problems' in the language. His dissertation has four parts (chapters). Parts I and IV are dedicated to phonology and parts II and III are dedicated to syntax. For the phonology, Part I is a summary of what he presented in his master's thesis, with revisions and a few additional topics. For example, he adds a topic on 'accent' (instead of stress) and a discussion of "glottalization" (i.e., the treatment of glottal stop). In Part IV, still dealing with phonology, he introduces new topics, such as the groundbreaking topic of 'nasal harmony' – the first treatment of nasalization as a suprasegmental feature in Tukanoan languages – vowel harmony, and other phonological processes found in the language. As for syntax, Part II presents an analysis of the verbal system with special attention to evidentiality, modality, aspect and tense. In Part III, Kaye discusses some other syntactic 'problems', giving emphasis to agreement and nominalization (Part III for the most part, repeats what was earlier published in Kaye 1968). Kaye's (1971) article 'Nasal Harmony in Desano' is his last work on Desano. Although the same topic is treated (identically) in his dissertation, his article became a standard reference in considerations of the nasal systems in other Tukanoan languages.

From the early 1970s to late 1990s, the works on Desano were produced mainly by members of the Summer Institute of Linguistics. Miller (1973) presented a basic description of Desano phonology. It provides a brief description of syllable structure, and it was the first work to mention a 'tone system' in Desano (composed of a High tone and a Low tone). It also suggests an orthography for the language, based primarily on Spanish orthography. Mountain (1978) presented a list of 267 words in twelve Tukanoan languages, including Desano. Mountain also presented a brief introduction to the languages in his book (including geographical information, and some of their phonological characteristics).

Miller (1999) is a sketch grammar of Desano and is the most extensive work published on the language. However it does not provide information about several important aspects of the language. For example, in the phonology chapter, Miller does not include tone, a topic mentioned in her ealier article (cf. Miller 1976), where she claims tone exists. In her grammar she mentions instead the existence of a pitch-accent system, basing her analysis on the work of Barnes (1996) for Tuyuca. In a review of Miller's work, Wichmann (2002:804) says, "there is a large set of noun classifiers (but too little said about how they work)." He indicates that the book seems to give equal weight to morphology and syntax, but "actually reads as morphology-in-disguise" (p.804).¹⁹ Huber and Reed's (1992) '*Vocabulário Comparativo: Palabras Selectas de Lenguas Indigenas de Colombia*' is a comparative work with a list of 375 selected words in languages of different families, including Tukanoan, and thus, also includes the Desano language. It offers material for comparative studies, and includes a Spanish-English index of the vocabulary.

Ardila (2000) presents a summary of the typological characteristics of Desano based on Kaye (1970) and Miller (1999). Another recent work is the Desano-Spanish bilingual dictionary compiled by Alemán M., López H., and Miller (2000). This dictionary has only 896 words. Each word is followed by an example sentence. It includes illustrations for animals, birds, and fish. It is a very informative material, and it provides notes on different dialects, grammatical notes, and offers an orthography system based on Spanish.

More recently, my own contribution to the published linguistic work on Desano includes the article 'Acoustic Analysis of Voiceless Obstruents and Nasal Harmony in Desano' (Silva 2008), which presents an investigation of the acoustic properties of the voiceless stops in nasal versus oral contexts in order determine whether these segments are affected by nasal harmony. Although the study considers only a small set of data, I

¹⁹ Proulx (2003) offers another review of Miller (1999).

suggest that voiceless segments are phonetically affected by nasalization, although the articulation does not seem to be affected. Another work is Silva (*to appear*) 'The status of the glottals '?' and 'h' in Desano'. In this article, I offer an alternative analysis for the segments h and ? in Desano. I suggest that these segments are a prosodic realization of the suprasegmental laryngeal feature that occurs after the first vowel within the root. This analysis differs significantly from previous accounts that treated h and ? as full consonant segments (Miller 1999) or as a property of the (first) syllable (Kaye 1970).

1.11.2 Anthropological Publications

The number of anthropological studies on Desano is considerably large when compared to the available published linguistic studies. The first substantial anthropological investigation of aspects of the Desano culture is Reichel-Dolmatoff (1968) '*Desana, simbolismo de los indios Tukano del Vaupés*' (also published in French, cf. Reichel-Dolmatoff 1973). In this work, the author analyzes the cosmology and symbolism present in the Desano narratives, focusing on the creation myths.²⁰

Besides being a detailed ethnographic study, Reichel-Dolmatoff's (1971) *Amazon Cosmos: the sexual and religious symbolism of the Tukanoan indians* offers an analysis of Desano ideology, religious beliefs, and social values.²¹ Reichel-Dolmatoff also published several articles about Desano shamanism and symbolism (cf. Reichel-Dolmatoff 1976b, 1979, 1981, 1987); about the importance of ecology for the Desano people (cf. Reichel-Dolmatoff 1975, 1976a, 1990); about other cultural aspects, such as animal categorization and color categorization (cf. Reichel-Dolmatoff 1978); and general

²⁰ Morey (1971) offers a review of Reichel-Dolmatoff (1968).

²¹ Furst and Furst (1981) provide a review of Reichel-Dolmatoff (1971).

ethnographic studies about the group and its culture (cf. Reichel-Dolmatoff 1989, 1996).²²

Another anthropologist who has published extensively on Desano is Dominique Buchillet. She wrote many studies about the Desano material culture, traditions, symbolism, religious beliefs, medicine knowledge, etc. Buchillet (1983) is a doctoral dissertation about the traditional knowledge of medicine used by the Desanos to treat contagious diseases. Buchillet also published several other studies about the Desanos' knowledge of the ecosystem and its relationship to the treatment of diseases (cf. Buchillet 1987, 1988, 1990a, 1992, 1995a, 1995b).

James Miller and Marion Miller, members of the Summer Institute of Linguistics, have provided a brief description of the Desano people and their lifestyle, during the time they lived with the group (Miller and Miller 1973). Beksta (1988) *A Maloca Tukano-Dessano e seu Simbolismo* provides an analysis of the symbolism of the Desano longhouse within the context of Eastern Tukanoan culture.

Beginning in the 1980s, the Desano people started to write their own history.²³ For example, the book *Antes o mundo não existia* is a narrative by Firminiano Lana, which was transcribed by his son, Luís Lana (with assistance of Berta Ribeiro). In this work, Lana and Lana (1982) describe the origin of the universe and of the mankind, according to the Desanos' traditions. Another published work based on traditional narratives is Fernandes and Fernandes (1996) *Mitologia Sagrada dos Desana-Wari Dihputiro Pora*, with a narrative by Americo Fernandes and his son, Durvalino Fernandes

²² Several of these articles can be found in Reichel-Dolmatoff (1997), which is a collection of his most popular essays.

²³ These were written in Portuguese.

(with assistance of Dominique Buchillet). Lana (2002) *A origem da noite, ou como as mulheres roubaram as flautas sagradas* provides a narrative with illustrations by the author about the Desanos' sacred flute, called *yurupari*. The last work published by the Desano is Toramu and Guahari (2004) *Livro dos Antigos Desana – Guahari Dihputiro Pora*, a version of the origin myths narrated by Tomaru (Wenceslau Galvao) and transcribed by his son Guahari (Raimundo Galvao), with assistance from Dominique Buchillet. To conclude the list of anthropological and ethnographic works on Desano, there are the works produced by Menendez (2005, 2006, 2009) which present analyses of the paintings/drawings produced by the Desano.

Finally, there are several Desano pamphlets ("vernacular publications") prepared (mostly by indigenous teachers in educational projects). There is a small booklet (Castilho et al. 2002) with vocabulary words, and other language material that have been produced for use in the Desano communities.²⁴

1.12 Language Maintenance and Educational Projects

The Desano people are aware that their language is highly endangered. One of the goals of my research is to assist the Desano communities with the creation of educational projects, which aim at language maintenance. In July 2008, I conducted an orthography workshop together with Desano leaders and local teachers. With support from FOIRN, we decided to work towards a standard orthography for the language. Historically the language has not been written and several community members have expressed their

²⁴ Some of these materials were produced with support from the Endangered Language Fund (ELF) and the Center for American Indian Languages (CAIL), at the University of Utah.

frustration with past attempts at writing the language without knowing how. Since the beginning of this research, we have promoted a series of community meetings and workshops with the goal of resolving orthography issues and establishing a standard orthography.

The Desano people are aware of the endangered status of their language and culture. In fact, Desano leaders report that the group is known in the region as 'the ones who speak a borrowed language,' an allusion to the fact that most of the Desano people now speak only Tukano. The elders show their concern for the new generations who do not speak the language, and Desano leaders have been looking for a linguist to help them since the 1990s. The Desano language faces a big challenge. The number of fluent speakers is relatively small, and they are spread out in different villages, days away from each other. Nevertheless, the Desano people are interested in finding ways to keep their language alive.

During 1998-2000 a series of meetings to train indigenous teachers to teach their language promoted by the Secretary of Education took place in São Gabriel da Cachoeira. Thirteen Desano teachers participated in this training; however, they were trained to teach in Tukano. It was during this period that these Desano instructors decided to look for ways to preserve their language. Their first step was to look for a linguist to help them to develop an orthography for their language. In 2000, they had some local educators to help them and attempted to develop an orthography, but the project did not succeed. Since then, the Desano people have been using the writing systems they know -- some based on the Tukano orthography, and others on Portuguese and Spanish. During my first fieldwork with the Desano people in June 2007, I met with some of the Desano leaders and teachers in São Gabriel da Cachoeira. They showed their interest in working together to revitalize their language and culture. During this first meeting they decided to write a project proposal aiming at revitalization of their language and culture and to plan a linguistic workshop to discuss the issues related to their orthography.

In December 2007, they started the 'Projeto Bayawi', supported by FUNAI, the Indigenous Foundation in Brazil (an organ of the Brazilian Federal Government). This project aims to revitalize Desano traditions (and the language) through an immersion program in which the target is the youth. Through this project, the Desano people had several meetings in different communities where the elders would pass their traditional knowledge to the younger generations. Other activities during these meetings involved rituals, songs, traditional stories, and arts and crafts. The Desano are willing to collaborate on projects that can aid in their efforts to preserve their language and culture.

CHAPTER 2

PHONETICS AND PHONOLOGY

2.1 Introduction

This chapter presents an overview of the sounds and their organization in the Desano phonological system. The data in this chapter are based on field transcriptions and instrumental analyses of speech recordings using Praat ©. The speech was recorded as digitalized .wav files, sampled at 44.1kHz, sixteen bit, using a lapel microphone ECM-CS10 and a Marantz 660 digital tape recorder. Speech varieties described here include samples of both the *Boreká* and *Dihputiro* dialects as spoken in communities of the Tiquié and Papurí rivers respectively. Thus, the sound system of Desano as described here is based on a variety of data from approximately fifteen speakers. In general, all varieties share the same phonemic inventory; differences between speech varieties involve only allophonic variation.

Section 2.2 describes the phonemic inventory of Desano. In Section 2.3, I present the suprasegmentals of Desano: the stress/tone system and nasalization. In Section 2.4, I describe two morphophonemic processes: vowel deletion and fusion, and syllable reduction. Finally, in Section 2.5, I provide some notes on the Desano orthography.

2.2 Segmental Phonology

I now provide an overview of the Desano segment inventory based on the analysis of my own data. Desano has a total of fifteen phonological segments, nine consonants, and six vowels. These segments are described here in terms of their phonological features (cf. Kenstowicz 1994), the articulations involved in their realization, and their acoustic properties. All vowels and voiced consonants have nasal counterparts. Nasalization is a suprasegmental feature of the morpheme.²⁵ Thus, nasality can be realized either on the word-level (with words containing only nasal morphemes) or, minimally, on the syllable-level (with words containing both nasal and oral morphemes), as discussed in detail in Section 2.3.3.

Vowels and the features that define them are given in Table 2.1 and discussed in Section 2.2.1. Consonants and their defining features in terms of manner and place of articulation are given in Table 2.2 and discussed in Section 2.1.2.

2.2.1 Vowels

As shown in Table 2.1, Desano has six underlying contrastive vowel segments. Table 2.3 illustrates the vowel distinctions according to the features [back], [front], [high] and [round]. These vowels are representative of the proto-vowel system for the Tukanoan languages (Barnes 1999: 210).²⁶

²⁵ Nasalization as a suprasegmental feature has been identified in all Eastern Tukanoan languages (Barnes, 1999:211).

²⁶ See also Stenzel's (2004:65) analysis for Wanano. See also Gomez-Imbert's analysis for Barasano (in Gomez-Imbert and Kenstowicz, 2001:421).

	FRONT	CENTRAL	BACK
HIGH	i	u	u
MID	e		0
LOW		a	

Table 2.1 Desano vowel inventory

Table 2.2 Desano consonant inventory²⁷

	LABIAL	DENTAL	VELAR	GLOTTAL
PLOSIVE				
[-voiced]	р	t	k	(?)
[+voiced]	b	d	g	
FRICATIVE		S		(h)
APPROXIMANT	W	у		

²⁷ The status of the the glottals P and h is unclear as discussed in Section 2.2.2.2.

	/i/	/e/	/ʉ/	/u/	/0/	/a/
[BACK]	-	-	-	+	+	+
[FRONT]	-	-	-	-	-	-
[HIGH]	+	- 1	+	+	-	-
[ROUND]	-	-	-	+	+	-

Table 2.3 Desano Vowel Phonemes

The features presented in Table 2.3 allow an interpretation of the vowel / μ / as a high central unrounded vowel, as in Miller (1976:108; 1999:9).²⁸ Kaye (1968:6; 1970:12) characterizes the vowel / μ / as a high back unrounded vowel.

In order to provide a quantitative description of the Desano vowels, I used an instrumental technique for characterizing vowels as suggested in Ladefoged (2003: 104-37), which plots the vowels in a vowel space defined by the first two formants (see Figure 2.1).

The words illustrating the six vowels (from two male speakers), and the measurement of the first three formants for each target vowel are shown in Table 2.4.²⁹ A plot of F1 vs. F2 provides a description of the vowel qualities in Desano. Figure 2.1 shows that the vowel **u** in Desano should be characterized as a high central (unrounded) vowel. This account provides a symmetrical vowel system (cf. Lindblom 1986).

 $^{^{28}}$ I use the barred-u (instead of the the IPA barred-i (high central unrounded vowel) for two reasons: (i) it is easier to read and recognize and; (ii) my Desano consultants and collaborators prefer to use 'u' for easiness of representation (and also to differ from the Tukano's transcriptions/orthography).

²⁹ Individual phonemic segments or set of phonemic segments are presented between slashes / /. The phonetic realizations of the phonemes are presented between brackets [].

Vowels	Words ³¹	Formant 1	Formant 2
/i/	[wi [?] i] 'house'	280	2023
/ʉ/	[du ² a] 'to keep'	328	1614
/e/	[sero] 'bench'	501	1663
/a/	[baa] 'food'	612	1303
/0/	[kore] 'to wait'	473	999
/u/	[du ² a] 'to yank	328	1046

Table 2.4 Illustration of Desano vowels and formant measurements³⁰



Figure 2.1 Formant plots of the six vowels of Desano

 ³⁰ Formant values were measured using Praat © version 4.3.
 ³¹ For now, I suppress the indication of tone (see Section 2.3.2 for a description of stress and tone).

Minimal pairs and/or near-minimal pair words illustrating the six contrasting vowels in oral contexts are shown in Table 2.5.³²

All vowels have nasalized vowel counterparts $[\tilde{i}, \tilde{u}, \tilde{e}, \tilde{a}, \tilde{o}, \tilde{u}]$. That is not to say that these vowel segments are marked as either oral or nasal. Nasalization is a suprasegmental feature of the morpheme. Words illustrating the six contrasting vowels in nasal contexts are illustrated in Table 2.6.

i	u	e	a	0	u
	/-k u /	/-k e /	/-k a /		/-k u /
	[-ki]	[-ke]	[-ka]		[-ku]
	'CLS:tree'	'IMP'	'EVID:REASON'		VBLZ
/b i /			/b a /	/b o /	
[bíí]			[báá]	[boo]	
'rat'			'food'	'to squeeze'	
/si/		/s e /	/s a /	/so/	
[sii]		[sɛɛ]	[saa]	[soo]	
'there'		river bank	'to over flow'	'to rest'	
	/t u /		/t a /		/t u /
	[t uu]		[taa]		[tuu]
	'to put'		'to cut'		'to learn'
/-di/		/-d e /	/-d a /	/-d o /	/-du/
[in-]		[-re]	[-ra]	[- ro]	[-ru]
INTER		REF	PL:AN	'NOM:ABSTR'	CLS:concave

Table 2.5 Desano vowel contrasts in oral contexts

 $^{^{32}}$ The symbol y in the phonetic spelling indicates a laryngealized vowel.

ĩ	ũ	ē	ã	õ	ũ
/~bidi/	/~budu/	/~bede/	/~bada/		/~budu/
[៣រិព]	[mʉ̃rʉ̃]	[mẽrẽ]	[mãrã]		[mũrũ]
'to advance'	'deceased'	'to fall'	'have not'		'tobacco'
		/~pe/	/~pa/	/~po/	
		[pẽẽ]	[pãã]	[põõ]	
		'two'	'to open'	'progenee'	
/~si/	/~su/		/~sa/	/~so/	
[sĩĩ]	[sũũ]		[sãã]	[sõõ]	
'to raise (hands)'	'to cover'		'to place'	'there'	
	/~abʉ/	/~abe/	/~aba/	/~abo/	/~abu/
	[ãmʉ̃]	[ãmẽ]	[ãmã]	[ãmõ]	[ãmũ]
	'to prepare'	'to be small'	'to arrange'	'to deliver'	'to fix'

Table 2.6 Desano vowel contrasts in nasal contexts³³

A survey of around 1200 lexical items in my corpus shows that there are sequences of vowels that are not allowed in the language.

Table 2.7 shows constraints on vowel-vowel sequences. The empty slots indicate vowel combinations not found in the data. It is notable that all vowels can be followed by a vowel identical to it. This seems to be a vowel lengthening process in order to maintain the bimoraic structure of the root.

 $^{^{33}}$ Nasalization is indicated by a tilde (~) preceding the nasal morpheme; this is a common convention in Tukanoan linguistics.

	i	u	e	a	0	u
	/di/	/tiʉ/		/dia/	/~bidio/	/diu/
i	[dii]	[tiʉ]		[dia]	[ຫາົກາິອ]	[diu]
	'meat	'to squeeze'		'river'	'Miriõ'	'egg'
		/~bu/		/dʉa/		
u		[m ũũ]		[dʉa]		
		'to lift'		'to keep'		
	/~yei/		/~be/	/~kea/	/~beo/	
e	[nēî]		[mẽẽ]	[kẽã]	[mē ^h ēõ]	
	'to be ugly'		'to hold'	'to attack'	'mom'	
	/~wai/	/aʉ/		/ta/		
a	[wãĩ]	[aʉ]		[taa]		
	'uncle'	'dad'		'grass'		
	/~boa/		/goe/	/doa/	/go/	/~bou/
0	[mõã]		[goe]	[doa]	[goo]	[mõũ]
	'salt'		'to return'	'to sit'	'latex'	'not have'
	/bui/		/bue/	/bua/		/bu/
u	[bui]		[bue]	[bua]		[buu]
	'agouti'		'to study'	'to descend'		'to be hard'

2.2.1.1 Vowel Alternations

In Desano, the main vowel alternations involve the lowering of phonemic /e/ and /o/ to $[\varepsilon]$ and $[\mathfrak{d}]$ respectively, oral contexs; /o/ is also realized either as the phonemes $[\tilde{\mathfrak{d}}]$ or $[\tilde{\mathfrak{u}}]$ in nasal contexts.

2.2.1.1.1 /e/ /o/ lowering into [ε] [ɔ]

In oral morphemes (i.e., morphemes not affected by nasalization, see Section 2.3.3 on nasality), the vowels /e/ and /o/ are usually lowered and then realized as [ε] and [ɔ] when they are in the final syllable of a disyllabic word, as illustrated in (1).

(1)	a. /go.be/	[go'bɛ]	'hole'
	b. /a.be/	[aˈbɛ]	'sun/moon'
	c. /de.ko/	[deh'kɔ]	'water'
	d. /i.go/	[igɔ]	'38G.F'

$2.2.1.1.2 / o \rightarrow [o] \sim [u]$

The vowel /o/ when in final position of a nasal word is realized either as [o] or [u] as illustrated in (2).

(2)	a. /~pigo/	[pĩŋõ] ~ [pĩŋũ]	'tail'
	b. /~bedo/	[bērõ] ~ [bērũ]	'CLS:circle'

2.2.1.2 Vowel Lengthening and Devoicing due to Preaspiration

In Desano, voiceless plosives /p, t, k/ and the voiceless fricative /s/ are preaspirated when they occur morpheme-internally. However, it is also common that, in the same environment, the vowel preceding the voiceless segment is lengthened, and the second part of the vowel becomes voiceless - these are two different realizations the same process of preaspiration. Examples in (3a-c) illustrate the vowel lengthening and partial devoicing.³⁴

³⁴ This has also been observed in Tukano (cf. Ramirez 1997:28).

(3)	a. /goso/	$[gooso] \sim [gohso]$	'thigh'
	b. /bupu/	$[buupu] \sim [bu^h pu]$	'thunder'
	c. /dika/	[diika] ~ [diʰka]	'arm'

I treat this 'pre-aspiration' as a prosodic feature of the root (see Section 2.1.2.2 below). One aspect of this topic that deserves future investigation is the fact that vowel lengthening and devoicing occur even if the voiceless segment is not part of the root morpheme. However, that has been noticed to happen only when the voiceless segment is the voiceless velar /k/, as in the suffix *-ke* 'IMP'. Note that in this case, preaspiration is not triggered, as showed by the forms marked with *. This is seen in (4).

(4)	a. /~sa-ke/	[sãą̃ke] ~ *[sãʰke]	'put.in-IMP'
	b. /~pa-ke/	[pãą̃ke] ~ *[pãʰke]	'open-IMP'
	c. /wa-ke/	$[waake] \sim *[wahke]$	'go-IMP'
	d. /oo-ke/	$[\acute{o}\acute{o}ke] \sim *[\acute{o}^{h}ke]$	'give-IMP'

There are cases in which this vowel lengthening and devoicing do not happen, as illustrated in (5).

(5)	a. /bia-k	ce/	[biake] ~	*[biaake]	'close-IMP'
	b. /~oba	ke/	[õmãke]	~ *[õmãą̃ke]	'wrap-IMP'

The examples in (4) and (5) suggest that vowel lengthening and devoicing is triggered only when the suffix *-ke* is attached to a monosyllabic root. A hyphotesis to be investigated is that maybe the lengthening of the monosyllabic roots has to do with a rule that the root needs to be bimoraic (the preferred root structure, see Section 2.3.1.2), then once it is, the pre-aspiration effect takes place, as a two step process. Further data are

necessary in order to verify whether this process occurs systematically in these environments.

2.2.1.3 Vowel Harmony

Both Kaye (1970) and Miller (1999) suggest that Desano has a vowel harmony system that has the harmonizing feature of height (high to low, low to high) and is restricted to five verbal suffixes: -di 'nom' becomes -de; -bi 'NEG' becomes -be; -bu 'POT' becomes -bo; -ku 'EVID:NVIS' becomes -ko; and -yu 'EVID:HSAY' becomes -yo (Miller 1999:18). Some examples of vowel harmony given by Miller (1999:19) are given in (6).³⁵

(6)	a.	/wa-bu-gi/	'the one (male) about to go'
	b.	/wa-bo-go/	'the one (female) about to go'
	c.	/wa-bu-ri/	'being about to go'

It is evident in examples (6a) and (6c) that the root does not undergo harmony, and that harmony has a right to left directionality, originating in the last suffix. If we consider harmony as a process that affects all the vowels within a domain, i.e., the word (cf. Kenstowicz 1994: 347), the process found in Desano may not be considered canonical vowel harmony. Unless the domain is the suffix domain defined to exclude the root. A possibility would be to treat this process as an alternation between adjacent vowels (for example, like umlaut) which are restricted to a few suffixes. This is a topic for future investigation.

³⁵ Miller's original phonemic representations and translated are preserved.

2.2.2 Consonants

The consonants /b, p, w, d, t, s, j, g, k/ occur word-initially and are contrastive, as we see in the examples in (7). Overall, the consonants are pronounced as indicated by their placement in the phonemic inventory given in the Table 2.2, except for the allophonic variation described in the next subsections.

(7)	a. /p/	pa	[paa]	'hit'
	b. /b/	ba	[baa]	'a type of basket'
	c. /w/	wa	[waa]	ʻgo'
	d. /t/	tudi	[iui]	'scold'
	e. /d/	du	[duu]	'stay'
	f. /s/	suri	[suri]	'nest'
	g. /j/	yudi	[juri]	'fall'
	h. /g/	gubu-du	[guburu]	'foot-CLS:concave'
	i. /k/	ku-du	[kuru]	'knot-CLS:concave'

The consonants /b d g/ are slightly pre-nasalized word-internally after a nasal vowel as illustrated in (8).

(8)	a.	~ya-bʉ	[ɲã ^m bʉ]	see-NON3.PERF	'I saw'
	b.	~yapi-bu	[nã ^h pĩ ^m bu]	potato-CLS:basket	'potato basket'
	C.	~boa-dʉa-gʉ	[mõã ⁿ dʉagʉ]	build-des-3sg:m	'he wants to build'
	d.	~diku-ge	[nī ^h kʉ̃ŋgɛ]	land-LOC	'in the land'
	e.	~uta-gʉ	[ũ ^h tãŋgʉ]	stone-CLS:mountain	'mountain'

It should be noticed that in Table 2.2 I have put the glottal segments 2 and h in parenthesis. This is because instead of treating them as full consonant segments, I analyze

them as being the phonetic realization of what I propose to be a suprasegmental laryngealization that occurs in root morphemes. Thus, for now, let's assume that their status as full consonant segments is unclear. (See Section 2.2.2.2 below on the status P and h).

I turn now to the discussion of the consonant segments. First, I discuss the voiced segments and their allophonic variations, the relation of /d/ to [r] and their allophones, in Section 2.2.2.1. The status of the glottal stop is discussed in Section 2.2.2.2.

2.2.2.1 Allophonic Variation of Consonants

The allophonic variations discussed in this subsection are: the relation of the segments /d/ and [r]; the dialectal variation of [r] and [r] as the phonetic realization of the intervocalic /d/; and the allophonic variation of the approximants /w/ and /y/.

2.2.2.1.1 The Relation of /d/ and /r/

The segment /d/ is a voiced dental alveolar plosive [d]. It is in complementary distribution with [r]. The data in (9) illustrates one of the distributional patterns of these segments: [d] occurs word-initially, and [r] occurs root internally.

- (9) a. doa [doa] 'sit'
 - b. deko [dehko]'water'
 - c. dia [dia] 'river'
 - d. dade [dare] 'ipixuna fish'
 - e. kede [kere] 'jacu bird'
 - f. pedu [peru] 'caxiri drink'

Both [d] and [r] occur word-internally in suffix-initial position, as illustrated in (10) for [d] and (11) for [r]. However, this [d] ~ [r] distribution can be explained as allophony if we consider the morphological criteria. Kaye (1965) claims that [d] occurs as the result of compounding and [r] as the result of suffixation. He noted that many cases of suffixation are subject to nasal assimilation (i.e., if the root is nasal, the suffix will be realized as nasal). This is because suffixes are generally unmarked for nasality (see Section 2.3.3).

(10)	a.	wea-di-g u	[weadigu]	clay-CLS:meat-3SG:M	'statue'
	b.	~boa-dʉa-gʉ	[mõãdʉagʉ]	build-des-3sg:m	'(he) wants to build'
	C.	~gʉya-doa-~b	o [ŋʉ̃ɲãdoamõ]	think-sit-3SG:F.IMPER	F '(she) was thinking'
(11)	a.	~duga-ku-di	[nũŋãkuri]	begin-PERM-INTER	'(can I) begin?'
	b.	g u a-de	[guare]	1pl:incl-ref	'to/for us'
	c.	gasi-du	[gahsiru]	bark-CLS:concave	'canoe'

The word-internal instances of /d/ in (10) occur in a noun classifier (10a) and in verbal roots (10b, 10c).³⁶ The word-internal instances of [r] in (11) occur with grammatical suffixes. The fact that /d/ is realized as [d] in (11a) but [r] in (11c) can be explained by the grammaticalization of these classifiers. The former can occur as a root while the later cannot.

2.2.2.1.2 The $[r \sim r]$ variation

The $[r \sim t]$ variation is dialectal rather than a variation conditioned by other segments in a single dialect. The speakers of the *Dihputiro* dialect use the alveolar flap [r]

³⁶ Classifiers can be full or reduced noun roots, depending on its level of grammaticalization (see Section 4.3.4 in Chapter 4).

to represent the letter < r > in their orthography. The speakers of the *Borekå* dialect of Desano spoken in the Tiquié River use [t] to represent with the letter < l > in their orthography (see Section 2.5 on orthography).

2.2.2.1.3 Allophonic Variation of Approximants

The phoneme /w/ is voiced bilabial approximant [w]. It is frequently realized as a voiced labiodental approximant [v] in word-initial (12a, 12b) and in morpheme-initial positions in intervocalic, oral environments (12c). In nasal environments, it is realized as $[\tilde{w}]$ as in (12d).

(12)	a.	wai	[wai] ~ [vai]	'fish'
	b.	wede	[were] ~ [vere]	'tell'
	c.	baya-wi	[bajawi] ~ [bajavi]	'dance-house (house of dances)'
	d.	~wai	$[ilde{ ext{w}} ilde{ ext{a}} ilde{ ext{l}}]~\sim [ilde{ ext{v}} ilde{ ext{a}} ilde{ ext{l}}]$	'name'

The phoneme /y/ is a voiced palatal approximant (high front glide) [j]. It occurs in word-initial and in word-internal (morpheme-initial) positions. Word-initially, /y/ is frequently realized as a voiced palatal-alveolar affricate [dʒ], as in (13a-c). In nasal environments, /y/ is realized as [n], as in (13d and 13h). In morpheme internal position, /y/ is usually realized as [j], as in (13e, 13f).

(13)	a.	yu'u	[dʒʉ [᠈] ʉ]	Ι
	b.	ya'i	[dʒa²i]	'heron'
	C.	yuu	[dʒuʰu]	'one'
	d.	~yabi	[ɲãmī]	'night'
	e.	kaya	[kaja]	ʻgirau'

f.	baya	[baja]	'sing'
g.	-ya	[dʒaa]	'POSS'
h.	-~ya	[ɲãã]	'CLS:creek'

2.2.2.2 The Status of the Glottals [h] and [?]

In this section, I describe the status of the glottal segments [h] and [?] (these segments occur in root and bound morphemes in Desano). Roots can be monosyllabic, i.e., (C)V or disyllabic, i.e., (C)V.(C)V, as illustrated in Table 2.8 (A) plain roots and (B) laryngealized roots. Syllable structure in Desano is (C)V, except in environments where the laryngeals 2 and h may appear in coda in the first syllable of roots. Vowel sequences are bisyllabic (V.V) (see Section 2.3.1 on syllable structure).

In previous descriptions of the phonology of the language, Kaye (1965, 1970) and Miller (1976, 1999) treated the glottal segments [h] and [?] as full consonant segments. In the present analysis, I consider the distribution patterns of and the phonetic realization of these segments and argue from distribution facts that these segments are better treated as a suprasegmental laryngeal feature that is assigned to root morphemes in Desano.

In the Eastern Tukanoan literature, the glottal stop [?] is referred to as as 'glottalization', and the glottal fricative [h], as 'aspiration'. I will retain this terminology to refer to these laryngeal features.³⁷

³⁷ These, however, are confusing to linguists not familiar with Tukanoan, since aspiration is a secondary manner of articulation of consonants (stops and sometimes affricates, very rarely fricatives) and glottalization is for ejective consonants primarily.

(A) Plain roots		(B) Laryngealized roots		
a. V	<i>i</i> 'do',	a. CV.?V	$du^{2}u$ 'to release', $du^{2}a$ 'to yank'	
b. CV	<i>ba</i> 'food', <i>ga</i> 'hawk'	b. V.?V	o^2o 'to give', a^2u 'dad'	
c. CV.CV	kãrĩ 'to sleep', gobe 'hole'	c. CV?.CV	gu'a 'stool', nõ'mẽ 'maiden'	
d. CV.V	wea 'clay', bua 'to go down'	d. V?.CV	$\tilde{a}^2 r \tilde{i}$ 'to say', $\tilde{o}^2 m \tilde{a}$ 'to run'	
e. V.CV	<i>ārī</i> 'to be', <i>abe</i> 'sun'	e. V.hV	$o^h o$ 'banana', $e^h a$ 'to arrive'	
f. V.V	ãĩ 'to take', oa 'skunk'	f. CV.hV	du^hu 'to leave', du^ha 'to stay'	
		g. CVh.CV	mū ^h tā 'to break', de ^h ko 'water'	
		h. Vh.CV	ahsi 'be hot', ohte 'to plant'	

Table 2.8 Plain and laryngealized roots

2.2.2.2.1 The Glottal Fricative [h]

This segment presents an interesting issue in Desano phonology. Kaye (1965) does not include [h] in the consonant inventory. He notes that the distribution of [h] contradicts the generalization of the (C)V syllable structure. According to Kaye, the occurrence of [h] can be explained by a generative rule that applies to morphemes. Thus, this "rule provides for each sequence VV to be rewritten VhV (where VV is a sequence of any two identical vowels and ' indicates [+accent])" (p. 47). Kaye (1970) includes [h] in the phonemic inventory. In both works Kaye treats the occurrence of [h] etween two vowels, and [h] before voiceless consonants as two different phenomena. The former is a result of an accent placement rule and the later a result of pre-aspiration rule. According to Miller (1999), the glottal fricative [h] is a full consonant, which only occurs intervocalically and is followed by "the echo vowel that precedes [it]" (p. 12).

I start by introducing the occurrence of the glottal fricative [h] that occurs before voiceless segments within the root. In the Eastern Tukanoan literature this has been called 'preaspiration'. In root morphemes in Desano (as well as other ET languages in the Vaupés), there is a (pre-) aspiration that occurs systematically before voiceless segments /p, t, k, s/, and is analyzed as the phonetic coda of the preceding syllable in (C)VCV roots, as illustrated in (14a) and (14b).³⁸ This preaspiration does not occur before voiced consonants; thus, aspiration is disallowed in a structure like (14c).

(14)	a.	$V^{h}C_{[-voice]}V$	[a ^h si-ri]	be.hot-NOM 'hot'
	b.	$CV^{h}C_{[-voice]}V$	[wehko]	'parrot'
	C.	$CV^{h}C_{[+voice]}V$		

Figure 2.2 shows the sound wave and spectrograms for the words [a^hsiri] 'hot' and [we^hko] 'parrot'. The vowel and the aspiration are highlighted together as my claim is that laryngeal feature 'aspiration' is realized after the leftmost (first) vowel of the root morpheme. This is not to say that aspiration is a property of the vowel, but rather, that preaspiration and the partial devoicing of the preceding vowel are ways of describing the same phenomena.

Note that the vowels in the initial V^h and CV^h syllables of V^hCV and CV^hCV roots (Figure 2.2) have about the same timing as the vowels in initial V and CV syllables of VCV and CVCV roots, illustrated in Figure 2.3 for the words [gobɛ] 'hole' and [yeba] 'ground'.

³⁸ Although this preaspiration is a characteristic of Eastern Tukanoan languages, it has been reported to occur in at least one Western Tukanoan language: Siona (see Wheeler 1987:89; and Wheeler and Wheeler 1962: 101).



Figure 2.2 Spectrograms for the words [ahsiri] 'hot' and [wehko] 'parrot'

Note that the timing of the leftmost vowels of CV.CV roots (Figure 2.3) is about the same as the leftmost vowels of V^hCV and CV^hC roots (Figure 2.2). For some speakers, this aspiration that occurs before voiceless segments is realized as a voiceless vowel. It should be kept in mind that [h] is in effect the same thing as a voiceless vowel and it is its role in the language that determines if we write it as a voiceless vowel or as [h] - that is, [h] can be given any vowel quality by just placing the articulators in position for the particular vowel while saying "h"; [h] is slightly broader transcription; voiceless vowel is narrower phonetic transcription.

Examples in (14a) and (14b) can also be realized as [aasiri] and [weeko] respectively. Furthermore, preaspiration does not occur across morpheme boundaries. Thus, in that context, aspiration has a predictable realization: it occurs before voiceless consonants within the root morpheme.



Figure 2.3 Spectrograms for the words [gobɛ] 'hole' and [yeba] 'ground'

Besides its predictable occurrence before voiceless consonants within the root morpheme, the glottal fricative h also occurs between vowels, as illustrated by the forms V.hV (Be) and CV.hV (Bf) above, and repeated below in (15A) and (15B) respectively.

(15)	A. VhV	roots	B. CVhV roots		
	a. oho	'banana'	a.	weha	'paddle'
	b. uhu	'pacu fish'	b.	buha	'to cross'
	c. eha	'to arrive'	c.	behe	'to classify'

The forms in (15B), with the CVhV shape also have an alternative realization in which the glottal fricative is not realized; instead, a voiceless counterpart of the proceeding vowel is realized, as illustrated in (16). Note that this alternative realization of h as a voiceless vowel also occurs in CVCV roots as we saw in (3) and (4).

(16)	a.	[ɲãʰãrīke] ~ [ɲãą̃rike]	'enter-IMP'
	b.	[yuhugu] ~ [yuugu]	'one man'
	c.	$[m ilde{e}^{h} ilde{e} ilde{o}] \sim [m ilde{e}^{i} ilde{o}]$	'mommy'

Note that in (16) the glottal fricative *h* occurs between two identical vowels. I argue that the second vowel is best understood as an echo of the first one (i.e., CV_1hV_1 , where V_1 represents identical vowels). One convincing case is found in instances in which the glottal fricative occurs between two different vowels (i.e., CV_1hV_2 , where V_1 and V_2 are different vowels), as illustrated in (17). In these cases, the glottal fricative may alternate with a voiceless vowel (as a reflex of the same phenomenon), making the voiceless vowel identical to the vowel it precedes (like examples in [16]). As illustrated in (17) below the three patterns are alternates of each other. The alternation pattern of voicelessness and aspiration are phonetically natural patterns (Ladefoged and Maddieson, 1996).

		CV ₁ hV ₂	$CV_1V_1V_2$	$CV_1hV_1V_2$	
(17)	a.	[yuʰí]	[yuui]	[yuʰui]	'support used to hold baskets'
	b.	[mãʰí]	[mãą̃ī]	[mãʰãī]	'to turn around'
	C.	[mʉ̃ʰí́]	[m ũų ĩ]	[mũ ^h ũĩ]	'caranã leaf'
	d.	[gaʰí]	[gaại]	[gaʰai]	'another'

It should be kept in mind that there are other ways to analyze the occurrence of [h]. An alternative analysis is to treat [h] as a full consonant segment (as it has been suggested by Miller 1999) and, thus, avoid the issue of supplying a rule that predicts its presence. A fuller treatment of status of the [h] in Desano is a topic of current investigation (Silva, *to appear*).

2.2.2.2.2 The Glottal Stop [?]

The glottal stop that occurs in Desano has been described a property of the vowel and as a full consonant segment. Kaye (1965) describes the occurrence of the glottal stop within the root as a property of the preceding vowel in the root morpheme. According to Kaye, the glottal stop only occurs between two similar vowels, i.e. in (C)V₁?V₁ roots, where the second vowel is an echo of the first one. Kaye illustrated this with the examples reproduced here in (18).

echo vowel				echo vowel
				- + -
(18)	a.	du'u [du'u] 'chipmunk'	b.	~ku'di [kũ'ũnĩ] 'to bite'

Thus, each plain vowel would have a glottalized counterpart (i², u^2 , e^2 , a^2 , o^2 , u^2) in the inventory. Kaye (1970) extends his previous analysis and characterizes glottalization as a property of the syllable; i.e., the vowel of the first syllable of a word is glottalized. He refers to 'glottalized syllables' which 'may not be phonemic.' Because of its restrictive distribution – only the first syllable of a word (or root morpheme) may be glottalized. Kaye treats glottalization as a contrastive feature of the vowel in sequences of two identical vowels $V_1_V_1$ or a vowel before a consonant V_C . However, in both studies, Kaye does not account for root morphemes that have two different vowels $V_1_V_2$ as is illustrated in (19); in fact, he does not present any data of this type in either of his works.

c. gu'i [gu[?]i] 'turtle'

The glottal ? has also been treated as a consonant segment in Desano as well as in most of the Eastern Tukanoan languages (cf. González de Pérez 2000). Miller (1999), on the basis of comparing minimal pairs of contrasting the glottal stop [?] with the glottal fricative [h], as illustrated in (20), treats the two glottal sounds [?] and [h] segments as full consonant phonemes.

(20) a. $[0^{2}0]$ 'to give'

b. [o^ho]

Miller (1999) also mentions that the glottal stop can only occur in intervocalic position. Thus, when the glottal stop is followed by another consonant, an echo vowel

occurs after the glottal stop, as illustrated in (20) with data from Miller (1999:12).

'banana'

(20) a. /pudu/ [pu'uru] 'after'
b. /oake/ [o'oake] 'sweep' (IMP)

Miller's description of the glottal stop as a consonant segment does not account for the environmental restrictions on this segment. The glottal stop (or glottalization) only occurs in root-medial position, after the first vowel of the root. Furthermore, although segmental restrictions on coda position (coda constraints) are common (cf. Itô 1986; Blevins 1995). The glottal stop, in Desano, is the only segment that can close a syllable; there are no other syllable-final consonants that are restricted to the root-initial position syllable. Thus the glottal stop is both restricted to a single position, coda, and represents the only closed syllable type (CV?) in Desano.

Phonetically, the glottal stop and the glottal fricative (as seen in Figure 2.2) are unusual when compared to the other consonants in the language because they are the only segments that cause the shortening of the preceding vowel or create echo (split) vowels.



Figure 2.4 Spectrograms of [kore] 'before' and [ko're] 'to wait'

Figure 2.4 shows the sound waves and spectrograms of the root morphemes [kore] 'before' and [ko²re] 'to wait', respectively. Note that the timing of the first vowel in the CV sequence in (a) is as long as the sequence CV? in (b).

From this full set of phenomena, it may be argued that glottalization is better understood as a prosodic feature of the root morpheme, occurring on the first vowel, rather than a full consonant segment.

This analysis allows us to observe a relation that would otherwise be missed across two distinct processes. There are striking similarities regarding the occurrence of the glottals sounds [?] and [h] in Desano. The laryngeal sounds [?] and [h] show distributive patterns that are distinct from other consonants. First, they are the only segments that occur in coda position, and the only consonants-like sounds that appear in consonantal clusters. Importantly, note that these segments always appear after the first root vowel. Second, both segments may create a short echo vowel, so that words like *du'a* 'to yank' and *duha* 'to stay' can appear phonetically as [du²ua] and [du^hua],

respectively. Third, these segments (usually [h]) may occur in suffixes as allophonic variation of the segments /s/, /d/, and /k/ in rapid speech, so -ka 'dubitative' and $-\sim da$ 'plural animate' appears phonetically as [-ha] and [-hā] respectively. Last, the occurrence of [h] and [?] causes the shortening of the proceeding vowel.

Phonetic evidence regarding the occurrence of these laryngeal sounds show that the phonetic realization of [h] and [?] is better treated as single prosodic phenomena that occur after the first vowel within root morphemes. An alternative hypothesis is that laryngealization might be a suprasegmental feature that applies to root morphemes in the language.

There have been a few analyses proposing glottalization as a suprasegmental feature in Eastern Tukanoan languages. For example, Malone (1987) claims that glottalization in these languages developed from a suprasegmental in the proto-language. Ramirez (1997) analyzes the glottal stops in Tukano as the realization of a 'laryngealized' tone. These previous analyses have led Stenzel (2007) to treat glottalization as a suprasegmental feature in Wanano. Her analysis is similar to Ramirez's in that she postulates a third suprasegmental feature (besides nasalization and tone), called 'glottalization'.

For Desano, as mentioned above, Kaye (1970) notes that glottalization is a feature of the first syllable in the root and in (Kaye 1965) he groups the sounds [?] and [h] together due to their occurrence. Although Kaye does not mention a 'suprasegmental feature,' his observations regarding these segments provide evidence for that hypothesis.³⁹

An alternative hypothesis is that, roots are specified as [\pm laryngeal]. Thus, as illustrated in Table 2.8, plain roots in (A) are [-laryngeal] and laryngeal roots in (B) are [+laryngeal]. Then laryngealization is further marked by the features [+constricted glottis] and [+spread glottis], which are realized either as [?] or [h] respectively. Thus, [+laryngeal] roots can surface with either ? or *h* after the first vowel of the root morpheme, as illustrated in (22). According to this analysis, [+laryngeal] roots have to be additionally specified in order to explain the two types of surface realization.

(22) a. /dua/ \rightarrow [du^ha] 'to stay' [+SPREAD GLOTTIS] b. /dua/ \rightarrow [du²a] 'to yank'

[+CONSTRICTED GLOTTIS]

In this approach, the target for the laryngeal features is the mora. Thus, [+constricted glottis] or [+spread glottis] associates to the right edge of the first mora. This way, the feature will always be displayed in intermoraic position, as a glottal stop or a glottal fricative after the first vowel of the root morpheme.

2.2.3. Nasal Allophones of Consonants and Vowels

The nasalization process in Desano was first described by Kaye (1970, 1971). Voiceless segments /p/, t, k/ do not have nasal allophones. Silva (2008) shows that the differences found between the voiceless segments /p/, t, k/ in nasal versus oral

³⁹ Cross-linguistically, laryngealization as a prosodic feature is also found in Mixtec. Macaulay and Salmons (1995) state that laryngealization in Mixtec is a feature of roots.
environments warrant treating these segments as though they undergo nasality. The fricative /s/ cannot (at least by definition) undergo nasality, because you have to close the velum to produce an [s]; thus it is treated here as having no nasal counterpart.

Table 2.9 shows the consonants and vowels and their nasal allophones. Minimal pair examples for oral/nasal consonants in word-initial position are given in (22) below. A description of nasalization process in Desano is presented in Section 2.2.2.5.

(23)	a.	[b] / [~b]	ba	[baa]	'to eat' cf.	~ba	[mãã]	'path'
	b.	[d] / [~d]	doa	[doa]	'to sit' cf.	~doa	[nõã]	'who'
	C.	[g] / [~g]	ga	[gaa]	'hawk' cf.	~ga	[ŋãã]	'to continue'
	d.	[s] / [~s]	so	[soo]	'to rest' cf.	~so	[sõõ]	'be red'
	e.	$[w] / [\sim \! w]$	wa	[waa]	'to go' cf.	~wa	[w̃ãã]	'be.good'
	f.	[y] / [~y]	ya	[yaa]	'poss' cf.	~ya	[ɲãã]	'to see'

Table 2.9 Nasal allophones of consonants and vowels

Consonants						Vowels						
	LAF	BIAL	CORO	DNAL	VEI	LAR	FRO	ONT		BA	.CK	
									[- RO	UND]	[+RC	UND]
PLOSIVE	b	m	d	n	g	ŋ	i	ĩ	ŧ	ũ	u	ũ
FLAP			[1]	[ĩ]	I		e	ẽ			0	õ
APPROXIMANT	W	ŵ			y	ŋ					а	ã

2.3 Suprasegmental Phonology

In this section I describe syllable types and structure, and the two major suprasegmental features of Desano: the mixed system of stress and tone, and nasalization and its mechanisms of spreading. The description of tone presented here is a tentative analysis that analyzes tone and stress as a mixed system.

2.3.1 Syllable Structure

Desano syllable structure is predominantly CV and V. The syllables in the languages are generally open, except that there are also occurrences of the syllable shapes $(C)V^{h}$ and $(C)V^{2}$, as shown in (24e, f, h, i).⁴⁰ The glottal stop and the glottal fricative are the only segments that can occur in coda position (as discussed in Section 2.2.2.2). It should be noticed that a vowel-initial syllable can only occur word-initially or after another vowel, but there are no VC.V syllables. This phenomenon follows mostly from the fact that there are no (C)VC syllables, except where the final C is glottal stop or a glottal fricative.

(24)	a.	bui	[bu.i]	'agouti'
			CV. V	
	b.	~ u bu	[ũ.mũ]	'boy'
			V.CV	
	C.	~yega	[nẽ.ŋã]	'knee'
			CV.CV	

⁴⁰ These syllable shapes are similar to the ones presented by Miller (1999:15). Kaye (1968:20) postulates the structure of Desano syllable as (C)V([?] or [h]), where C is an *"optional consonant followed by the syllabic nucleus plus an optional choice of either glottalization or aspiration"*.

d.	do'e	[do. [?] e]	'traira fish'
		CV.CV	
e.	~ne'ka	a [ɲē².kā]	'to be tired'
		CV?.CV	
f.	~a'di	[in. ^ç ã]	'to say'
		V?.CV	
g.	eha	[e. ^h a]	'to arrive'
		V.hV	
h.	y u ku	[yʉʰkʉ]	'tree'
		CVh.CV	
i.	uti	[u ^h ti]	'wasp'
		Vh.CV	

Any consonant (both oral ones and their nasal allophones) can occupy the onset position of word-initial syllables, with the exception of the glottal stop 2 and glottal fricative *h*. Word-internally both the glottal stop 2 and the glottal fricative *h* can occupy the onset slot, like the other consonants, as in (24d, 24g). All syllables must have a vowel nucleus, and sequences (C)V₁V₂ (where V₁ and V₂ are different vowels), as in (24a), are analyzed as two separate syllables (CV₁.V₂). Thus, these vowel sequences in Desano are generally considered to belong to separate syllables and are not diphthongs. The reason for this is because each vowel in a (C)V₁V₂ sequence has a separate energy burst; that is, there are two separate intensity pulses, as shown in the spectrograms for the words [ŋãī] 'parakeet' and [nõã] 'bone' in Figure 2.5.



Figure 2.5 Spectrograms for $(C)V_1V_2$ sequences

Furthermore, when asked to say words with $(C)V_1V_2$ sequences slowly, speakers tend say these two vowels separately. Kaye (1965) also reports that "when speaking slowly, a speaker of Desano will cut sequences of the type CVVCV into sequences of the type CV V CV (where a space represents a pause, i.e., a syllable boundary)" (pp. 36-37).

While the vowels in (C)V₁V₂ sequences can be considered to belong to different syllables, the like vowels sequences, i.e. (C)V₁V₁ (where V1 and V1 are the same vowels), are considered as belonging to the same syllable. The examples in Figure 2.6 show the spectrograms for the words ga [gaa] 'hawk' and ye [yee] 'jaguar'. There is only one energy burst with one intensity pulse.



Figure 2.6 Spectrograms for $(C)V_1V_1$ sequences

2.3.1.1 Association Rules

The association rules used to establish the syllable structure of Desano are based on the procedures outlined by Dobrovolsky and Czaykowska-Higgins (2001), which reflect universal constraints on syllabification according to the following steps:

a) Nucleus formation: identify the vowel as the nucleus (N) of the syllable;

b) Onset formation: following the onset principle, associate any C with the nucleus to its right, forming the onset (O);

c) Coda (Cd) and Rhyme (R) formation: associate any remaining unassociated C (in Desano, it can only be either the glottal fricative h or the glottal stop 2 with the nucleus to its left.

(25) ASSOCIATION RULES

a. ~boa [mõã] 'salt' σ ~b o 'a CVV CV.V



2.3.1.2. Syllable Quantity and Weight

In the previous section, we saw that syllables in Desano tend to be 'codaless'. The exception being the (C)V.CV laryngealized roots, in which 2 and h can fulfill the coda position in the first syllable. Because syllables in Desano have no coda, the rhyme is

formed by a single vowel and the notions of mora and foot can be used to describe quantity and weight.

Each vowel (rhyme) is assigned to one single unit of quantity (mora). Onsets are extra-moraic, and, thus, are directly linked to the syllable (cf. Hayes 1995:53). The laryngeal sounds 2 and h have no effect on weight, as these exceptional closed syllables do not bear stress. Therefore, they share the same mora with the preceding vowel. In (26) the various syllable shapes found in Desano are each associated with a mora.

(26)	Syllable shapes and moraic association									
	σ	σ	σ	σ	σ	σ				
	1	1		- 1	Λ	Λ				
	μ	/μ	μ	μ	/μ	/μ				
	1	/								
	V	C V	V ?	V h	C V ?	C Vh				

Thus, root morphemes (plain roots and laryngealized roots as illustrated in Table 2.8) are bimoraic. ⁴¹ As mentioned in Section 2.2.1, plain root morphemes of the type $(C)V_1V_2$ are considered disyllabic, and thus have a bimoraic structure, as illustrated in (27).

(27)	a. bui [bui] 'agouti'	b. oa [oa] 'skunk'
	σ σ /μ μ	σσ μμ
	C V. V	V.V
	bui	o a

Many monosyllabic plain roots of (C)V shape are realized with a long vowel;

however, in Desano, vowel duration is not phonemically distinctive. These monosyllabic plain roots are realized phonetically with long vowels and interpreted as having bimoraic

⁴¹ Bimoraic structure has been analyzed for other Eastern Tukanoan languages, such as Tukano (Ramirez 1997:53-56), Barasana (Gomez-Imbert and Kenstowicz 2000:421), Tatuyo (Gomez-Imbert 2004), and Wanano (Stenzel 2004).

structure, as illustrated in (28). The bimoraic structure accounts for the assignment of stress and tone. Generally stress and tone falls on the second mora of a bimoraic morpheme (stress and tone are discussed in section 2.3.2).



Although the great majority of roots in Desano are bimoraic, there are a small number of monomoraic roots in the language. A small sample is shown in (29).⁴²

- (29) a. ~bo [mõ] 'piraiba fish'b. ~de [nẽ] 'buriti fruit'
 - c. ~bu [mũ] 'umari fruit'

2.3.2 Stress and Tone in Desano: A 'Mixed Stress-Tone' System

In this section, I provide a preliminary analysis to account for a mixed stress-tone system (so-called 'pitch accent' systems) in Desano. This analysis follows Michael's (2011) approach to analyzing mixed stress-tone (MST) in two unrelated Amazonian languages, Iquito (Zaparoan) and Kakataibo (Panoan) spoken in Peru (cf. Michael 2010). I show that the Desano stress-tone system has a striking resemblance to these two other languages, which indicates that MST might be an areal characteristic.

Although there is a general consensus in the linguistic literature regarding the properties of tone systems and stress systems, the properties of an MST system as an

⁴² These roots tend to be lengthened when they occur with no additional morphology, for example, [ga] may be realized as [gaa].

independent system are still a matter of debate. According to Hulst and Smith (1988), 'pitch-accent systems' are in a continuum between 'true tone languages' (e.g., Chinese) and 'stress accent languages' (e.g., English). In pitch accent languages, prominence is marked by pitch shape; a given syllable in a word carries a tonal shape such as LH.⁴³ In a stress accent language prominence is marked by factors other than pitch, such as duration and vowel quality difference in the marked syllable. In tone language usually every syllable has a tonal shape, and the tones are contrastive.

The typology for 'tone languages' and for 'stress languages' can be stated in this way: there is a continuum between a pure tone system and a stress system. Pitch accent systems fall in the middle. In a mixed system prominence is marked by both stress-like properties and pitch shape. Hyman (2007, 2009) suggests that instead of trying to classify languages with labels ("tone", "pitch-accent", "stress") we should characterize languages by their individual properties ('stress-like properties' vs. 'tone-like properties'). Thus, the analysis presented here takes into account the individual properties of the word-prosodic system of Desano, which is a language with stress-like and tone-like properties.

2.3.2.1 Evidence for a 'Mixed Stress-Tone' System

According to Michael (2010), a good indication that a language may have an MST system is if the prosodic systems of the language combine contrastive tone with regular stress. There is clear evidence of contrastive tone in Desano, as illustrated with the minimal pairs in (30)-(32).⁴⁴ The examples in (31) and (32) are multimophemic

 $^{^{43}}$ L = Low tone, H - High tone.

⁴⁴ The acute accent on the vowel ' \dot{v} ' indicates high tone; '' indicates primary stress is assigned to the syllable that follows.

words. In (31) the final syllable $-y\tilde{u}$ (/-~yu/) is the specific class marker 'CLS:palm', showing a contrast between between LH-L in (31a) and LL-H in (31b). In (32) the final syllable $-m\tilde{t}$ (/-~bi/) is the agreement marker '3SG:M', showing a contrast between LL-H in (32a) and LH-L in (32b).

(30)	a.	(wấ. 'ĩ)	'name'
	b.	(wã. 'í)	'uncle'
(31)	a.	(mĩ.hĩ. 'yũ)	'buriti palm tree'
	b.	(mĩ.hĩ. 'yấ)	'wild palm tree'
(32)	a.	(ã.rī. ˈmī́)	'he is'
	b.	(ã.㎡. ˈmĩ)	'he says'

There is also evidence that pitch prominence is assigned to the mora that bears stress.⁴⁵ In Desano, stress falls on the last mora in a bimoraic root, as illustrated in (33).

- (33) a. [yu^hkú] 'bark'
 b. [yu⁹ú] 'I'
 - c. [wi²i] 'house'

The link between pitch and stress can be seen in words in which the prominent pitch moves in order to maintain the same metrical position in a word when extra morphology is added to the word, as illustrated in (34).

- (34) a. yu^hku-gú 'tree'
 - b. yuhku-gu-ré 'to/for/at the tree'
 - c. yu^hku-du^hpú 'tree branch'
 - d. yu²-ré 'mine'

⁴⁵ It should be noticed that this is true even in nontonal languages such as English, where part of the traits that bundle with energy prominence of stress is also pitch (and length).



Figure 2.7 Prominent pitch movements

A sample illustration of prominent pitch movement in order to maintain metrical position is show in Figure 2.7 for the words *yuhkú* 'bark', and *yuhkúduhpú* 'tree branch'.

2.3.2.2 Acoustic Correlates of Stress in Desano

In the previous sections we saw that the great majority of root morphemes are bimoraic. In this section, I describe the assignment of stress in Desano considering acoustic evidence.

Generally, it can be difficult to characterize the acoustic correlate of stress (Lehiste 1970; Laddv1996; Beckman 1986; Hayes 1995). There are a number of phonetic measures that can be used to detect the differences between stressed and unstressed syllables (Gussenhoven 2004:14). In order to establish stress assignment, I consider the following based on Ladefoged (2001:231): (a) acoustic evidence, in the form of vowel lengthening in the stressed syllable compared with the shorter length of vowels in nonstressed syllables; (b) increase of loudness of the sound produced in the stressed syllable, and (c) the rise of (intonational) pitch.

The acoustic evidence considered here is lengthening of the vowel in the stressed syllable. ⁴⁶ A systematic measurement of vowel length in a variety of words of different lengths (root morphemes and their inflected forms), showed a variation in rhythmic lengthening of the stressed vowel. A sample of the words measured is illustrated in Table 2.10, where vowel length is measured in stressed and non-stressed syllables. Stress is marked with ''' preceding the stressed syllable. Kaye (1970:20) claims that verb roots are accented on their final syllable. Figure 2.8 shows the spectrograms for the words *kuye* 'eye' and *kuyeri* 'eyes' and the points of measurement for vowel length in each word. Note that although there is no variation in intensity (shown by the wavy line), vowel length is considerably different.

WORD	GLOSS	VL1	VL2	VL3
ku'ye	eye	0.066s	0.097s	
ku'yeri	eyes	0.103s	0.140s	0.060s
wa ^h 'ka	skewer	0.046s	0.065s	
wa ^h 'kari	skewers	0.041s	0.089s	0.040s

Table 2.10 Vowel length in Desano words in seconds

⁴⁶ I used PRAAT © for creating spectrogram and measuring the vowels.



Figure 2.8 Vowel length in Desano words.

Throughout the Desano data, a regular pattern was observed: the vowel of the right-most syllable in the root is the longest vowel of the root morpheme, suggesting that primary stress falls on the second (rightmost) syllable in a disyllabic root.

Another evidence used to establish stress assignment in Desano was 'speaker's judgment'. Speaker judgment was elicited through a task with one of my main Desano consultants. ⁴⁷ For this task, we first made a list of identical recorded words said by different speakers. These words were then transcribed and played back; then, the speaker was asked to underline the vowel in the 'strongest' part for each word. The speaker was first trained with disyllabic roots, and then was asked to underline the strongest part in inflected roots. The speaker's judgment was consistent with the duration facts, indicating

⁴⁷ Frank Matos is a fluent Desano speaker from San Jose del Viña, in Colombia. I have given him basic linguistic training and he has become a research assistant in the Desano Language Documentation Project.

that one of the main acoustic correlate of stress in Desano is the lengthening of the vowel in the stressed syllable.

2.3.2.3 Stress in Desano

Stress placement in Desano generally follows a straightforward rule: stress falls on the syllable containing the last mora of a noun root morphemes. Since verb roots cannot occur in isolation (i.e., without additional morphology); they do not receive stress. Stress in assigned to verbs only after additional morphology is added to it creating a phonological word. Thus, stress consists of left-to-right exhaustive parsing, and although degenerate feet may be allowed, it is not very common. There is no closed syllable in the language. Even though there are some exceptional syllable structures, which have either [?] or [h] in the coda position forming (C)V? and (C)Vh shapes, these syllables are not stressed. This can be seen as evidence that stress is assigned before any suprasegmental feature (as mentioned in Section 2.2.2.2, glottalization as treated as a possible suprasegmental feature in the language). Thus, stress in Desano is assigned primarily on the basis of moraic position.

The words in (35-39) illustrate the basic iambic pattern. The left-most disyllabic forms correspond to the root morpheme, followed by additional morphology to their right.

(35)	a.	(o.'a)	'skunk'
	b.	(bi.'i)	'rat'
(36)	a.	(ku.'ye)(_ri)	'eyes'
	b.	(yu. 'ku)(gu)	'tree'

(37)	a.	(di.'ta)(ru. ge)	'to/from/at the lake'
	b.	(yʉ.ˈkʉ)(dʉ.ˌpʉ)	'tree branch'
(39)	a.	(yʉ.ˈkʉ)(dʉ. pʉ)ri	'tree branches'
	b.	(õ. 'mã)(bi. ra)ri	'trick, game'

(40) a. (wa. i)($\eta \tilde{o} \tilde{a}$)(me r \tilde{a}) 'with the fishbone'

b. (ga. 'ki)(po. a)(da. ri) 'strands of monkey hair'

An indication that stress is assigned to the mora and not the syllable can be seen in words such as the ones in (41a-b), which demonstrate that these bimoraic roots form their own foot. When additional morphology is added to form a new phonological word, the language maintains its preferred bimoraic structure, with stress on the second mora, as illustrated in (42a-b).

It should be noticed that if we were to treat Desano feet as disyllabic, rather than bimoraic, the language would allow the unattested forms in (42c-d).

(41)	a.	(wi.'i)	'house'
	b.	(mã.ˈã)	'path'
(42)	a.	(wi.'re)	'to/at the house'
	b.	(mã. ˈre)	'to/at the path'
	c.	*(wi.'i)(_re)	
	d.	*(bã. 'ã)(_re)	

The stress assignment in Desano shows that a prosodic word has to contain at least one well-formed (i.e., bimoraic) foot. Having assigned the foot structure as rightheaded, being assigned from left-to-right, we can assign prominence in the structure of the phonological word. In Desano, this prominence is set to the left, i.e., left-headed. The illustration in (43a-c) shows that these parameters allow us to generate the correct surface forms for the words yu^h 'kugu 'tree' (43a), di^h 'taruge 'to/from/at the lake' (43b), and ga^h 'kipoadari 'strands of monkey hair' (43c).

(ga.'ki)(po._a)(da._ri) 'strands of monkey hair'

Up to this point, I have used only nouns to illustrate stress assignment in Desano. This is because the great majority of noun roots can also function as independent words in this language. On the other hand, verb roots generally need additional morphology in order to be used as phonological words. Despite these differences, phonological words formed from verb roots are stressed following the same metrical parameters assigned for nouns.⁴⁹ This is illustrated in the examples in (44).

(44)	a.	o ² o-ke	\rightarrow	$(o^2$. 'ke)	'give-IMP'
	b.	õ²mã-mã	\rightarrow	(õ². 'mã)(ˌmã)	'they ran'
	c.	sua-mõ	\rightarrow	(su. 'a)(mõ)	'she sieves (it)'

⁴⁸ The asteristik '*' marks all and only the segments that qualify to bear stress. The dot '.' marks the segments that do not bear stress.

⁴⁹ Miller (1999:15) assigns stress to the level of the syllable and claims that stressed syllables have high pitch. She does not distinguish 'stress' from 'tone'.

With the facts stated above, we arrive at an analysis of stress in Desano, summarized as follow: (a) primary stress is in the final mora within the root; (b) stress in the phonological word falls on the left-most foot.

2.3.2.4 Desano Tone

In this section I present a preliminary sketch of the tone system in Desano.

Desano has both stress accent and tone. There are two tones, H and L. In general, every word carries an H tone, though there are exceptions.

The tonal minimal pairs in (45)-(48) evidence the existence of lexical tone.⁵⁰ These words result in four surface tone melodies: high (H), high-low (HL), low-high (LH) and low (L) in words with bimoraic structure.

(45)	a . H-	L	[wãĩ]	'name'
	b. L-	H	[wấi]	'uncle'
(46)	a. L-	L	[sʉa]	'to like'
	b. L-	Η	[sʉá]	'to sieve'
(47)	а . Н-	H	[táá]	'grass'
	b. L-	Η	[taá]	'to cut'
(48)	a. L-	Η	[duʰá]	'to stay'
	b. L-	L	[duʰa]	'to return'

This is a tentative analysis and much needs to be said about the tone system in this language. A future investigation might show how lexical tone and metrical tone interact with each other in the language. Michael (2009) shows that in Iquito (Zaporan) lexical

 $^{^{50}}$ High (H) tone is marked with an acute accent over the vowel ' \acute{v} '; Low (L) tone is unmarked.

tone is insensitive to metrical structure (i.e., stress). Metrical tone is, then, assigned to the syllable that bears stress. This might be the case in Desano. Because details of tone in Desano are still under investigation and native speakers prefer orthographic representation which does not mark tone overtly, I do not mark tone in the phonetic transcription here (see notes on orthography in Section 2.5). Where no tone marking is indicated in the phonetic transcription, it is because the tone pattern isn't consistent, or its realization is not clear at the moment.

2.3.3 Suprasegmental Nasalization

Table 2.11 shows a summary of the underlying segments and their respective nasal realization. It should be noted that voiceless stops do not have a nasal realization. The description of nasalization presented here accounts for the basic characteristics of the phenomena; however, it does not pretend to be exhaustive. Nasal harmony in Desano was described in Kaye (1970; 1971).

Nasal harmony in Desano is a suprasegmental feature of the morpheme. Morphemes are inherently marked as oral [-NASAL], nasal [+NASAL] or are unmarked for nasality [ØNASAL]. As noted in Section 2.1.3, all voiced segments have a nasal realization. Kaye (1971:37) states that "it is unclear whether the voiceless segments... have distinct oral and nasal realization." In an acoustic study comparing voiceless segments in nasal versus oral contexts, Silva (2008) argues that at least phonetically voiceless segments are affected by nasality, although it might not be realized in the surface form.

Underlying phonemes														
р	t	k	S	b	d	g	W	j	i	ŧ	u	e	а	С
co	Voice nsona	eless nts - r	10	m	n / ř	ŋ	ŵ	ր	ĩ	ũ	ũ	ē	ã	Õ
nasal realization						Seg	ment	s witł	n nasa	al rea	lizati	on		

Table 2.11 Desano phonemes and nasal realization

Some examples of suprasegmental nasalization occurring in root morphemes are given in (49).

		UNDERLINED FORM	SURFACE FORM	
		[+NASAL]	[+NASAL]	
(49)	a.	~dobe	nõmẽ	'female'
	b.	~duku	nũ ^h kũ	'forest'
	C.	~sea	sẽ²ã	ʻpiaba fish'
	d.	~pidu	pĩrũ	'snake'
	e.	~gabe	ŋãmẽ	'to want'

2.3.3.1 Nasal Spreading

As mentioned earlier, nasalization is a property of the morpheme; however, the scope of spreading is the phonological word. Roots are specified as being either [+NASAL] or [-NASAL]. Suffixes are generally unspecified (unmarked) for nasality, although there are a number of suffixes that are specified as [+NASAL]. Unmarked [ØNASAL] morphemes can be realized either as [+NASAL] or [-NASAL], depending on the

specification of the preceding morpheme. The examples in (50)-(54) show some of the morphemes that are unmarked for nasality and how they are realized depending on the $[\pm NASAL]$ feature of the preceding morpheme: [-NASAL] (examples a) and [+NASAL] (examples b).

[ØNASAL] MORPHEMES

		[-N	NASAL]			NASAL]			
(50)	-go	a.	wʉa-go	[wʉago]	b.	~kari-go	[kãrĩŋõ]		
	3sg:f		be.tall-3sc	be.tall-3sg:F		sleep-3sg:	:F		
			'she is tall	,		'she is sle	eping'		
(51)	-gu	a.	w u a-g u	[wʉagʉ]	b.	~kari-g u	[kãrīŋ ũ]		
	3SG:M		be.tall-3sc	3:M		sleep-3sG:	Μ		
			'he is tall'			'he is sleeping'			
(52)	-de	a.	igo-de	[igore]	b.	~ig u -de	[ĩŋ ũ ĩẽ]		
			3sg:f-ref			3SG:M-REF			
			'to/for/at he	er'		'to/for/at ł	nim'		
(53)	-do	a.	disi-do	[di ^h siro]	b.	~pigo-do	[pĩŋõrõ]		
	PART		mouth-CLS	s:body.part		tail-CLS:bo	ody.part		
			'a mouth'			ʻa tail'			
(54)	-di	a.	wi'i-di	[ini ⁹ iw]	b.	~japi-di	[nã ^h pĩrĩ]		
	PL:IN		house-PL:I	N	sweet.potato-PL:IN				
			'houses'			'sweet pot	atoes'		

In the examples above, we see that the $[\pm NASAL]$ feature of the root morpheme spreads to the unmarked morpheme. However, marked morphemes block nasal

spreading. In examples (55)-(58) words are formed by sequences of [+NASAL] [-NASAL] $[\emptyset$ NASAL] morphemes. Oral morphemes block the nasal feature from spreading to the unmarked morpheme to the right.

- (55) ~uba -be -go [ũmãbego] 'she isn't tall'be.tall-NEG-3SG:F
- (56) ~diku -ge -de [nĩhkũgere] 'in/on/at (the) earth' earth-LOC-REF
- (57) ~pabu -goda-gu [pāmugoragu] 'he (is) transforming (creating)'
 ferment-reach-3sG:M
- (58) ~buku -bidi -go [mũhkũbirigo] 'she is not happy'
 be.happy-NEG-3SG:F

Generally oral morphemes are not affected by nasalization even when they are surrounded by two [+NASAL] morphemes, as illustrated in (59)-(60).

+N -N +N

- (59) ~ai -de -~da [ãīderã] 'they took (it)' take-PERF-PL:AN
- (60) ~bobe -bidi -~bi [mõmēbirimī] '(he) doesn't work'

work-NEG-3SG:F.IMPERF

Nasalization generally spreads from left to right. Kaye (1970, 1971), Bivin (1986), and Miller (1999) claim that certain suffixes assimilate nasalization from the morpheme to their right. Both Kaye (1970, 1971) and Bivin (1986) suggest a set of rules to explain nasal assimilation from right to left. Miller (1999) claims that right-to-left

nasal assimilation is 'totally lexicalized', since it is limited to only a few morphemes. Most of the examples used by these authors to illustrate right-to-left assimilation are words with a sequence [+N] [-N] [+N], as illustrated here in (61)-(62) with data from Miller (1999:14).⁵¹

(61)	a.	wēhēbīrā	'the ones who don't kill'
	b.	wēhēbigi	'the one who doesn't kill'
(62)	a.	wēhēyõrā	'they killed (reported)'
	b.	wẽhẽyoro	'you killed (reported)'

It is important to notice that the morphemes assimilating the nasal feature from the rightmost morpheme are between two [+NASAL] morphemes in the examples in (61a) and (62a). Furthermore, inherently oral morphemes that are realized as nasal in these environments have a monosyllabic shape. In (60) above, the 'negative' morpheme *-biri* is not affected by nasality even though it is between two [+NASAL] morphemes. Thus, considering that right-to-left nasal assimilation is restricted to a few suffixes, it is better treated as an idiosyncratic process affecting only a small number of [-nasal] suffixes (in [59], for example, *-de* 'perfect aspect' remains oral even though it is between two nasal morphemes).

In my corpus, this idiosyncratic left-to-right assimilation seems to be common to the evidential suffixes -yu 'quotative/folklore' and -yo '2nd hand/hearsay', as illustrated in the examples (63) and (64). In (63a), the oral status of -yu is evident from the fact that it is not affected by nasality from the preceding morpheme, and it blocks nasality to spread to the unmarked morpheme. In (63b) -yu follows an oral morpheme, but it is

⁵¹ I maintain the author's original transcription and glosses.

realized as nasal; in this case, it assimilates nasality from the nasal morpheme in the right. However, in (63c) -*yu* remains oral, even though it has a similar environment as in (63b).

		$+\mathbf{N}$	-N	ØN		
(63)	a.	~digi	-yu	-do	[nĩŋĩyuɾo]	'(they) stood up'
		stand-	EVID:QU	JOT/FOLK-NON3	.IMPERF	
		-N	-N	+N		
	b.	adi	-yu	-~bi	[aɾiŋũmĩ]	'(he) said'
		say-EV	/ID:QUC	ot/folk -3 sg:m.	IMPERF	
		-N	-N	+N		
	c.	boka	-yu	-~da	[boʰkayurā]	'they met'
		meet-I	EVID:QU	JOT/FOLK -3 PL:A	N.PERF	

The same idiosyncrasy can be seen with the suffix *-yo* in (64) below. It is an oral morpheme as shown in (64b), but assimilates the nasal feature from the nasal morpheme to its right (64c).

		-N	-N			
(64)	a.	duya	-yo		[duyayo]	'(it) lacked (something)'
		lack-E	VID:HS	AY		
		$+\mathbf{N}$	-N			
	b.	~adi	-yo		[ãñyo]	ʻ(it) was'
		be-EV	ID:HSA`	Y		
		-N	-N	$+\mathbf{N}$		
	C.	wa	-yo	-~da	[waanõrã]	'(they) traveled'
		go-ev	ID:HSA	Y - PL:AN.PERF		

2.4 Morphophonemic Processes

There are two major morphophonemic processes that occur in Desano: vowel deletion and fusion, treated by Kaye (1970) as 'coalescence' and syllable/segment reduction. I present here a summary of these processes as discussed in Kaye (1970) and Miller (1999), presenting my own data.

2.4.1 Vowel Deletion and Fusion

In Desano, two types of phenomenon are found to occur in vowel-vowel sequences: deletion and fusion. Examples of deletion are seen with the following verb roots: $\sim ari$ 'be', $\sim a'ri$ 'say' and *ari* 'come'. When those verbs roots are followed by the 'perfective' aspect, -a, the high front vowel /i/ is deleted, and the verb roots have the surface forms [$\tilde{a}r\tilde{a}$] 'be', [$\tilde{a}^{2}r\tilde{a}$] 'say' and [ara] 'come', as illustrated in (65)-(67).

(65) $\tilde{a}r\tilde{a}bu$ wekuruge⁵²

~adi-a-bʉ	weku-du-ge
be-perf-non3.perf	balaio.basket-CLS:concave-LOC
(It) is in the balaio baske	et.'

(66) yāmāpu ā'rāpu īgūre

~yaba-pʉ	~adi-a-pʉ	~igu-de
deer-CONTR	say-PERF-3SG.M.PERF	3sg:m-ref
'The deer said to h	nim.'	

⁵² The first line of the four-line examples corresponds to the orthographic representations adopted by the Desanos.

(67) ara yu'u

adi-a yuu come-PERF 1sG 'I came.'

Other verb roots that end with the high front vowel and are followed by the 'perfective' -a do not undergo this process, as shown in the examples (68)-(69).

(68) nõ'õ mũ'ũ koadukaru apiari

~doo	~b uu	koa-duka-du	api-a-di						
where	2sg	gourd-fruit-CLS:concave	leave-perf-inter						
'Where did you leave the gourd?									

(69) kariamĩ

kadi-a-~bi

sleep-perf-3sg:m.IMperf

'(He) slept.'

Fusion occurs in like vowel-vowel sequences across morphological boundaries. For example, when the 'perfective' suffix -a is preceded by a morpheme that also ends in /a/ these two vowels are fused (there is no lengthening of the fused vowel), as shown in (70). When the preceding vowel is other than /a/, the vowels are not fused, as shown in (71).

(70) waaduarã

wa-dʉa-a-~da

go-des-perf-3pl.an.perf

'(They) wanted to go.'

(71) yu'u mũ'ũrẽ dorebea ã'rīyũmĩ

yuu ~bu'u-de dode-be-a ~a'di-~yu-~bi 1SG 2SG-REF command-NEG-PERF say-EVID:QUOT/FOLK-3SG:M.IMPERF 'I don't command you, he said.'

Vowel deletion and fusion present an issue that deserves more attention. Kaye (1970:180-186) treats these two processes as coalescence and suggests a set of derivational rules in order to explain the deletion and fusion of vowels.

2.4.2 Syllable Reduction

Syllable reduction in Desano is very common.⁵³ It generally involves morphemes that have a (C)VCV shape and are reduced to CV, where the first syllable is deleted. For example, the verb *api* [a^h pi] 'to leave' reduces to *-pi* when it is in a noninitial position. The process seems to This is illustrated in (72); note that in (72a), the verb has its full shape; whereas in (72b), its first syllable has been deleted.

(72) a. mũ'ũãrẽ nĩkũ ahpirã

	~b uu -~a-de	~dikʉ	api-~da		
	2sg-pl:an-ref	ground	leave-AN:PI		
	'(They) are leaving l				
b.	mãrĩ põrãrẽ mãs u pi				

~badi ~po-~da-de ~basu-api

1PL progenee- PL:AN-REF raise-leave

⁵³ According to Stenzel (p.c.), syllable reduction in serial verb constructions is quite common. In these contexts, the phonological reduction of the second vowel indicates that the verb function is also different in the morphological structure of the word.

'(That) we leave our children.'

Miller (1999:18) shows examples in which the reduction has to do with the loss of the intervocalic /g/. This is illustrated in (73).

(73) a. waauta

wa-gʉ-ta

go-3sg:m-emph

b. yãũ waa

~ya-gʉ-ka

see-3 SG:M-EVID: reason

'He (will) see.'

In (74), besides the reduction of /g/, the initial consonant is also deleted. This reduced form is used as a term of endearment. In general, the form pagu [pagu] 'father' is ore commonly used. The form can be also reduced when bounded to a preceding morpheme, as shown in (74b), in which pagu is reduced to -pu.

pa-gʉ

genitor-3SG:M

'dad'

b. yu'pu

yuu-pa-gu

1sg-genitor-3sg:M

'my father'

2.5 Notes on Orthography

The Desano orthography is currently in the process of being standardized. In this dissertation, I adopt the orthography that has been developed by the Desanos in the 'Desano Language Workshops'. A summary of the Desano phonemes and their respective orthographic representation is shown in Table 2.12.

For the intervocalic /d/ which is realized as flap [r], the Desanos have decided to use the actual 'flap' symbol 'r'. The Desano have also decided that the high mid unrounded vowel /i/ be represented as 'u' in order to 'distinguish' their orthography from that of Tukano.

The Desanos are currently in the processes of discussing and deciding on the best ways to represent nasalization, aspiration, glottalization, and tone. However, based on the most recent workshop, they have decided on the following:

- Nasalization is marked with a '~' in all vowels of nasal morphemes (and words).
 For example *igū* 'he' (not *igu/igū*); *wãī* 'uncle' (not *wãi/waî*).
- Aspiration is marked with the grapheme h before the consonants p, t, k, s. For example gahpi 'monkey', dehko 'water', ahsiri 'hot'. The consonant h will also be used in words such as oho 'banana' and uhu 'pacu fish'.
- Glottalization is marked with a '. For example, wa'i 'fish', wi'i 'house'.
- Tone is not marked for now.

In the next chapters, I adopt the Desano orthography as listed here instead of the phonetic transcription.

CONSONANTS								VOWELS							
phoneme	/p/	/t/	/k/	/b/	/d/	/g/	/s/	/w/	/j/	/i/	/i/	/e/	/a/	/0/	/u/
oral grapheme	р	t	k	b	d	g	s	W	у	i	u	e	a	0	u
nasal grapheme				т	n				ñ	ĩ	\tilde{u}	ẽ	ã	ð	ũ

Table 2.12 Desano phonemes and graphemes

2.6 Summary

This chapter presented an overview of Desano phonology. I described the phonemic inventory of the language, which is comprised of 11 consonants, and 6 vowels, and their variants. I showed that the status of the glottal segments 2 and h as full consonant segments is debatable.

In Section 2.3, I presented the suprasegmentals of Desano, starting with a description of the syllable structure. Stress and tone were treated as a mixed system creating an interaction of *lexical tone* and *metrical tone*. The other suprasegmental trait presented was nasalization. I provided a description of the system and its basic process of nasal spreading.

In Section 2.4 I described three morphophonemic processes: vowel deletion, vowel fusion, and syllable reduction. Finally, in Section 2.5, I provided some information on the Desano orthography which I adopt in the following chapters.

CHAPTER 3

PARTS-OF-SPEECH

3.1 Introduction

This chapter introduces the parts-of-speech in Desano. It presents analyses of the patterns related to word formation and its internal structure. It is intended as an introduction to the topics to be discussed in the chapters on nominal morphology (Chapter 4) and verbal morphology (Chapter 5). In this chapter, an account for the 'word' in Desano is given considering the distinction between phonological and grammatical word. However, the 'word' is defined according to the grammatical criteria.

In Desano, members of the parts-of-speech are divided into *open classes* versus *closed classes*. Members of the open classes are *nouns* and *verbs*. Like many Amazonian languages, Desano does not have a morphosyntactically distinct adjective class. In Desano, adjectival expressions (semantic adjectives) are derived from nouns and verbs. Adverbial notions are expressed in several ways in Desano. Temporal adverbial expressions are expressed by nouns referring to time. Manner and quality adverbial expressions are derived from nominalized verbal roots. Quality adverbial expressions are coded is through serial verb constructions. The closed classes consist of *personal pronouns* (and other pro-forms), *demonstratives* and *interrogatives*, *negators*, *discourse markers*, *adverbial conjunctions*, and a small set of *interjections*.

3.2 The 'Word' in Desano

The criteria used for characterizing the phonological word in Desano are based on the cross-linguistic framework presented in Dixon and Aikhenvald (2002). According to this typological framework, the 'word' can be defined according to phonological and grammatical criteria. These two kinds of criteria are discussed below with respect to the specific characteristics of the 'word' in Desano.

3.2.1 The Phonological Word

According to these criteria, a phonological word is defined as a unit not smaller than a syllable (in Desano, the word can be formed minimally of one syllable) that is characterized by (A) its segmental features; (B) its prosodic features; and (C) its phonological rules, to which I now turn.

(A) Segmental features. Every phonological word must have at least two moras (or morae), that is, they are bimoraic. Thus, if a word has the minimally monosyllabic structure (C)V, the vowel must be a long vowel, giving the shape (C)VV (vowel length is not contrastive in Desano), as illustrated in (1).

- (1) a. ga [gáa] 'hawk'
 - b. ta [táá] 'grass'
 - c. oa [oá] 'this'

With respect to phonotatics and segment restrictions in Desano, allows [r] and [h] only in word-medial position. Words may not begin with these segments.⁵⁴

⁵⁴ Although in the practical orthography speakers tend to write *ho* 'banana', *hu* 'pacú fish', these words are pronounced as [oho] and [uhu] respectively.

(B) Prosodic features. An important characteristic of the phonological word is that it contains one primary stress (see Section 2.3.2.3). Another prosodic feature that characterizes the phonological word in the Desano is tone assignment – it has at least one H tone per phonological word (see Section 2.3.2.4).

(C) Phonological rules. The scope of nasalization spreading (nasal harmony, see Section 2.3.3) is the phonological word. This process does not extend beyond word boundaries. Besides the characteristics listed in (A, B, C), Desano speakers recognize a 'word' as the utterance that can be separated by a space in writing.

3.2.3 The Grammatical Word

According to the framework used for the identification of 'word' (cf. Dixon and Aikhenvald 2002:19), a grammatical word has as its core element one or more lexical roots to which morphological processes have applied, co-occur in a fixed order, and have a conventionalized coherence and meaning. Thus, grammatical word here means how words are defined in terms of their grammatical behavior.

The grammatical word in Desano has as its base element an independent root, although the language has many instances of root compounding that constitute a complex stem.⁵⁵ These complex stems, in turn, can take one or more suffixes. The morphemes that are suffixed to the root and/or stem occur in a fixed order.⁵⁶

⁵⁵ In Eastern Tukanoan languages, only noun roots can stand alone as words; verb roots always require some kind of accompanying inflectional morphology (Gomez-Imbert and Kenstowicz 2000:421).

⁵⁶ See Chapter 4 and Chapter 5 for the morphology of nominal and verbal structures, respectively.

3.3 Parts of Words: Roots and Formatives

Words in Desano may consist of more than one morpheme. These morphemes, in turn, can be classified in several ways. The most important distinction of morphemes in Desano is that between roots, clitics, and formatives. In Desano, roots are the base of both the phonological and grammatical words. In this sense, a root is "a base that cannot be analyzed any further into constituent morphemes" (Haspelmath, 2002:17). Formatives are considered here as 'the markers of inflectional information' (cf. Bickel and Nichols 2007:172).

Clitics are morphemes that do not meet the criteria for phonological word (cf. Aikhenvald 2002b). The most common clitic is the possessive morpheme ya which is phonologically dependent on other morphemes, as shown in (2). The most common occurrence of ya is as an enclitic (i.e., following the noun root), as in (2a), although it can also behave like an independent root and receive nominal morphology, as in (2b).

- (2) a. nähseä ya nihkü
 - ~dase-a=ya ~diku Tukano-PL:AN=POSS land 'Tukano's land' b. ĩgũ yago ~igu ya=go

3SG:M POSS=3SG:F 'his wife'

3.4 Parts of Speech: An Overview

Miller (1999:21) indentifies seven 'major parts of speech' for Desano: nouns, verbs, modifiers in noun phrases, adverbs, pronouns, interjections, and particles. Miller's characterization seems to rely mostly on semantic properties, rather than on grammatical ones, and she makes no distinction between open versus closed word classes. In the following discussion, I identify the parts of speech of Desano in a quite different way.

The parts of speech of Desano as presented here are based on grammatical criteria (i.e. morphological and syntactic criteria), not semantic criteria, although semantic evidence might be used to support the classification of a particular word in a given class. These grammatical criteria include the properties of a word's distribution, its syntactic functions, and its morphological and syntactic specifications.

I follow the theoretical assumption that the parts of speech in Desano consist of the classes of words (*open* or *closed*) found in the language.⁵⁷ The term 'word' used here, and in the remainder of the chapter, refers to the grammatical (not phonological) word. Thus *open* classes (also called 'major classes') include words that are members of large categories; they are unrestricted, as new words can be added to these classes. For example, nouns and verbs are generally characterized as open classes. On the other hand, *closed* classes (also called 'minor classes') include words that are members of small categories; they are restricted, as new words are generally not added to these classes. Personal pronouns, numbers, and determiners are characterized as closed classes.⁵⁸ To illustrate this in Desano, we find words from the open class of nouns that are loanwords

⁵⁷ This analysis is based on Schachter and Shopen's (2007) cross-linguistic characterization of part-of-speech systems.

⁵⁸ For lengthy discussions about the characterization of parts of speech, see Chapter 2 of Givón (2001); Chapter 7 of Lyons (1968), and Schachter and Shopen (2007).

productively introduced into the language. For example, *kārīnā* 'chicken', *tabua-mīhī* (plank-CLS:flat) 'bed', *pīsānā* 'cat' are borrowed into the language from the Portuguese words *galinha* 'chicken', *tábua* 'plank' (note the classifier *-mīhī* is Desano), and *bichano* 'kitty', respectively. While these examples are clear instances of loanwords being added into the open class of nouns in Desano, there are no clear instances of loanwords added to the closed classes.

3.4.1 Open Classes

Two major lexical classes are identified in Desano: noun and verb. These are characterized in relation to each other. Some nouns are derived from verb roots, and some verbs are derived from nouns. Adjectives and adverbs, which are open classes in many languages, are identified as closed lexical classes in Desano; 'adjectival' and 'adverbial' notions are expressed through nominal and verbal roots and bound morphology (and clitics).

Members of the closed classes are: adjectives, adverbs, personal pronouns, interrogatives, deictics, numerals, quantifiers, and interjections. In the following sections I characterize the members of the open classes: nouns and verbs. As mentioned, the notion of 'word' used in the discussion here is the grammatical word, not the phonological word.

3.4.1.1 Nouns

The prototypical members of this class are roots (or stems) that are heads of noun phrases, as in (3a), and are the arguments of a predicate – the subject and object of the

clause as in (3b), or a locative argument, as in (3c). A noun in Desano may be recognized based on a number of formatives that are associated with nouns. Nouns might occur in its bare form (uninflected), as in (1) above.

- (3) a. yaa wi'i
 - yaa wii POSS house 'my house'
 - b. Guaho peru iriamõ

Guaho pedu idi-a-~bo Guaho caxirí drink-PERF-3SG:F.IMPERF 'Guaho drank the caxirí.'

- c. ĩgũ nũhkũge ãrĩmĩ
 - ~igu ~duku-ge ~adi-~bi 3sg:m forest-loc be-3sg:m.imperf

'He was in the forest.'

Nouns in Desano also inflected for number (4a), and noun classification,

including classifier (4b) and gender (4c).⁵⁹ Nouns may also undergo regular

compounding, i.e., a noun formed by two independent roots as illustrated in (4d).

(4) a. mãhã poari

~baha poa-di

macaw hair-PL:IN

'macaw feathers'

⁵⁹ See Chapter 4 for further discussion on nominal morphology.
b. yuhkugu ta'abe ĩgũrẽ paabu

yuku-gu	taa-be	~ig u -de	pa-a-bʉ
tree-CLS:trunk	cut-CEL	he-REF	hit-perf-non3.perf
	0 1	111.11 (.1	1

'(I) broke a piece of wood and hit him (the monkey).'

c. mãhĩgo bago imõ

~bahi-go	ba-a-go	i-~bo
child-3sg:F	eat-PERF-3SG.F	do-3sg:f.imperf
'The girl ate.'		

d. yuhkubuhka

yuku-buka

tree-fruit

'fruit (of a tree)'

Nouns derived from verbs. Many nouns are derived from verb roots by applying nominal morphology, as illustrated in (5), where the nominalizer suffix *-di* is attached to the verb root to form a noun. Nominalization is discussed in Section 4.3.5, Chapter 4.

(5) guare bari deyobea

g u a-de	ba- di	deyo-be-a
1pl:incl-ref	eat-NOM	appear-NEG-PERF

"... there is not enough food for us."

3.4.1.2 Verbs

The prototypical verb functions as the head of the predicate. Examples of Desano verbs as predicates of an intransitive and a transitive clause are given in (6) and (7), respectively.

(6) ĩgũ yuriamĩ

~igu yudi-a-~bi 3sg:M fall-PERF-3sg:M.IMPERF 'He fell.'

(7) Karu pĩrũrẽ weheapʉ

karu~pidu-dewee-a-puCarlossnake-REF:OBJ kill-PERF-3SG.M.PERF'Carlos killed the snake.'

Verbs are generally cross-referenced, with respect to the subject, for person, number, and gender. The grammatical categories associated with verbs are tense/aspect/modality (TAM), mood, and evidentiality. Verbs can be formed by a single root (simple), as illustrated in (6) and (7) above, or by verb root compounding as illustrated in examples (8) and (9).

(8) mãrĩ werenĩrã ia pare

~badi wede-~adi-~da i-a pare 1PL:EXCL tell-say-3PL:AN.PERF do-PERF then 'Then, we are narrating...'

(9) taãidihari eheomāhsū inyūmā

ta-~ai-dii-a-di eo-~basu i-~yu-~ba

cut-take-descend-PERF-PERF feed-tame do-EVID:QUOT/FOLK-3PL:AN.IMPERF '(They) collected (the nests), and bred (the birds).'

Verbs derived from noun roots. Some verbs are derived from noun roots (by applying verbal morphology or functional derivation with no morphological marking), as shown in (10).

(10) imīsõkāyāge ĩgũ dui wãĩku gohanũgãdi
i-~biso-ka-ya-ge ~igu dui
DEM:PROX-squirrel-CLS:triangular-CLS:creek-LOC 3SG:M Luis
~wai-ku goa-~duga-di
name-VBLZ write-begin-PERF

'From squirrel creek, he, named Luis, began to write.'

3.4.1.3 Derived Categories: 'Adjectival' and 'Adverbial'

Desano does not have adjectives or adverbs as separate categories. Words coding adjectival expressions or adverbial expressions are derived from verbs and/or nouns, as described in the following subsections.

3.4.1.3.1 'Adjectival'

In the traditional typology of word classes, it has been claimed that while nouns and verbs are found in every language, not all languages have an adjective class (Dixon 1977, 1982). However, Dixon (2006) claims that a class of adjectives can be defined for all languages. In the current typological literature, it is often claimed that in South American languages, a class of adjectives is either lacking or it is a (very) small lexical class (cf. Campbell 2012b). Gildea (2012), for example, argues that languages of the Cariban family, in the Amazon, do not have a morphosyntactically distinct adjective class. In these languages, adjectival expressions (semantic adjectives) are derived from nouns and verbs.

To date, the prevailing view in the descriptions of Eastern Tukanoan languages (e.g., Gomez-Imbert 1982 for Tatuyo; Gomez-Imbert 1997 for Barasano; Ramirez 1997 for Tukano; Stenzel 2004 for Wanano) has been that there is no category of 'adjective', rather that these are qualities expressed by stative verbs and some nouns.

Most of the 'adjectival' notions in Desano are expressed by stative verbs and some nouns.⁶⁰ The present analysis is based on Schachter and Shopen's (2007) treatment of the topic. According to these authors, languages that lack an adjective class can be divided in two groups:⁶¹

- (i) adjectival-noun languages, in which adjectival meanings are expressed primarily by nouns; and
- (ii) adjectival-verb languages, in which adjectival meanings are expressed primarily by verbs.

Desano fits into group (ii), an adjective-verb language. In general, in Desano, adjectival expressions are derived from stative verbs (which I call descriptive verbs). As stated by Stenzel (2004:105), the evidence that adjectival expressions, in Wanano, are derived from verb roots is evidenced by the fact that these roots require a nominalizing morphology in order to derive nominal words (adjectivals). In Desano, this statement

⁶⁰ That adjectives share properties with noun and verbs have been suggested as a characteristic of the Tukanoan language family (Barnes 1999:221; 2006:138).

⁶¹ Schachter and Shopen (2007:16).

applies only if the noun being modified by the 'adjectival expression' is an inanimate noun, as in (11) and (12). When the noun being modified refers to an animate entity, the nominalizer suffix -di is not used. Instead, the noun class morphology for animates, is added directly to the descriptive verb root, as illustrated in (13) and (14).

(11) yuhkugu wuarigu ara

y u ku-gu	wua-di-gu	~adi-a
tree-CLS:trunk	be.big-NOM:IN-CLS:trunk	be-perf
'The tree is big.'		

(12) yuhkugu ameriguga ara

yʉkʉ-gʉ	~abe-di-gʉ-~ga	~adi-a
tree-CLS:trunk	be.short-NOM:IN-CLS:trunk-DIM	be-perf
'The tree is (very) small.'		

(13) ěrā dehko ărīmī ěrā pagu wuagu

~eda	deko	~adi-~bi	~eda	pa-g u	wʉa-gʉ
3AN:PL	half	be-3sg:m.IMPERF	3PL:A	N genitor-38G:M	t be.big-3sg:m
'Among them there was their big father.'					

(14) ĩgũ yo'gu wãikudi ãrīmĩ ĩgũ yo'gu bugu

~igu yogu wãī-ku-di ~adi-~bi ~igu yogu bugu
3SG:M stutter name-VBLZ-PERF be-3SG:M.IMPERF 3SG:M stutter elder
'He, the stutter, was called old stutter.'

Descriptive (stative) verbs do not need to be nominalized (through nominal morphology) in order to express function as an adjectival expression. These verbs roots

can also receive verbal inflection and still have an adjectival meaning, as shown in

Examples (15) - (17).

(15)

(16)

mũ'ũ	poberika
~b uu	pobe-bidi-ka
28G	be.fast-NEG-EVID:REAS
'You are not fast.'	
wãbu yãbu yu'u ãrīyũmĩ	

~wa-bu~ya-buyuube.good-NON3.PERFsee-NON3.PERF1SG~adi-~yu-~bisay-EVID:QUOT/FOLK-3SG:M.IMPERF

'It was good, I saw it. He said.'

(17) mũ'ã mãhsĩrõ werebeoke

~buu-a ~basi-do wede-beo-ke

2SG-PL:AN know-NOM:ABSTR tell-be.quick-IMP

'Tell (us) quickly something known by you.'

3.4.1.3.2 'Adverbial'

Adverbial notions are expressed in several ways in Desano. Temporal adverbial expressions are expressed by nouns referring to time, for example $\sim kadu$ 'yesterday', *dopa* 'today'. Temporal adverbial expressions derived from these time words can contain the nominalizer suffix *-do* and the suffix *-de*, which codes temporal adjuncts, as illustrated in (18). It can also take the diminutive suffix *-~ga*, to express

intensity/proximity, as in (19) and (20). The suffix *-de* codes the derived adverbial as a temporal expression argument.

(18) ã doparõrẽ mũ'ũ pãdero pare

~a dopa-do-de~bu²u~pa-de-dopareyes today-ADV-REF:TMP2SGopen-PERF-ADVthen'Yes, now you have opened one.'

(19) mārī yēhkūsērā pererā imā doparogārē

~badi	~yeku-~se-~d	a	pede-~da	i-~ba
1pl	grandfather-11	PL:EXCL-PL:AN	end-PL:AN	do-3pl:an
dopa- do-~	-ga-de			
today-ADV-DIM-TMP				
'Our grandparents are vanishing right now.'				
igo yāmīg	ã wagokũmõ			
igo	~gabi-~ga	wa-go	kudi-~bo	

-		-	
3sg:f	night-DIM	go-38G:F	walk-3sg:F.IMPERF

'She leaves tomorrow.'

(20)

Manner and quality adverbial expressions are generally derived from verb roots by the adverbializer suffix *-do*; however, they generally do not take nominal morphology, as shown in (21) and (22). The diminutive suffix *-~ga*, can occur in these constructions, and it expresses intensity, as illustrated in (23). Manner adverbial expressions do not take case markers.

(21) wãrõ mõmẽūkā ãrīgū

~wa-do ~bobe-gu-ka ~adi-gu

be.good-ADV work-3SG:M-EVID:reason say-3SG:M 'Saying he works well.'

(22) õ daharo payogāripu dehkoyārī eheagū yā

~0	daha-do	payo-~gadi-p u	deko-~yadi
DEIC:PROX	be.slow-ADV	fly-advance-3SG.M.PERF	half-almost
eha-~gʉ	~ya		
arrive-3sG	:M see		

'Here, flying slowly, seeing (he) was arriving almost half of the way.'

(23) dia ãmērõgā yura nēmõādero ārābu

dia	~abe-do-~ga	yuda-~debo-a-de-do
river	be.small-ADV-DIM	raise-increase-PERF-PERF-NON3.IMPERF
~ari-a-b u		

be-PERF-NON3.PERF

'The water (level) went up a little bit.'

Another way that quality adverbial expressions are coded is through serial verb constructions, as illustrated in (24) and (25).

(24) mũ'ũreta buriyāduarimī

~b uu -de-ta	buri-~ya-dua-di-~bi	
2pl-ref-emph	be.hard-see-DES-PERF-3SG:M.IMPERF	
'He wanted very much to see you.'		

(25) yu'u yẽãrīka

- yuu ~ye-~adi-ka
- 1sg be.bad-be-EVID:reason

'I am feeling unwell. (lit. I'm bad.)'

Other adverbial expressions are derived from nominal roots followed by the nominalizer suffix *-do*, and the 'manner' suffix *-pa*, as illustrated in (26). This example also shows a common adverbial expression formed by the particle $n\tilde{e}$ 'nothing' followed by the noun *gora* 'pinpoint' and the temporal suffix *-de*, which occurs frequently in narratives in temporal expressions meaning 'in the beginning'.

(26) õpã në gorare

~o-pa	~de	goda-de
DEIC:PROX-MANN	nothing	pinpoint-REF:TMP
'Thus, in the begin	nning (lit. Li	ke this, in the beginning)'

3.4.2 Closed Classes

In Desano, the closed classes that can be identified are (a) personal pronouns and other pro-forms (reflexives and reciprocals), (b) demonstratives, (c) interrogatives, discussed in Chapter 4. Other members of the closed classes are: the negator $\sim de$, the discourse marker *baa*, the 'adverbial' conjuctions *pare* 'then' and *daha* 'again', and a small set of interjections.

The negator $\sim de$ occurs as an unmarked object as in (27), or more commonly it can occur in adverbial expressions, as in (28).

(27) në duariyura igurë

~de	dʉa-bidi-~yu-~da	~igu-de		
nothing	keep-NEG-EVID:QUOT/FOLK-3PL.AN.PERF	3sg:m-ref:obj		
'They didn't keep anything for him.'				

(28) (...) nẽ õẽãripu

NEG be.tired-PERF-NEG-3SG.M.PERF

'(The tuiuiu bird flew) without tiring.'

The word *baa* occurs in discourse, at the end of the utterance, indicating that the statement said by the speaker implies mutual (or general) knowledge, as illustrated in

(29).

(29) irisibure yē'ēmārāsā doutosā mārīrāmā baa

~yeeb u -~ra-~sa	douto-~sa
so.and.so-PL:AN-ADD	doctor-ADD
baa	
mutual.knowledge	
	~yeebu-~ra-~sa so.and.so-PL:AN-ADD baa mutual.knowledge

'At that time, doctor people didn't exist (here).'

There are two adverbial words that occur quite frequent in discourse. The words *pare* 'then' and *daha* 'again' are used in narrative as conjunctions to connect independent clauses. The adverbial conjunction *pare* can appear at the beginning or at the end of a sentence or between two clauses, as illustrated in (30).

(30) a. ārõge ārīkā ārīta ia yu'u

~ado-ge		~adi-ka	adi-ta	
DEM:PROX-LO	С	be-EVID:reason	be-EMPH	
i-a	y uu			
do-perf	1sg			
'I say we live here.'				

b. guaya mãhkãpu erota duaka pare

gʉa=ya ⁶²	~baka-p u	edo-ta		
1PL:EXCL=POSS	settlement-CONTR	DEIC:DIST-EMPH		
dua-ka	pade			
stay-EVID:REASON	CONJ			
'Then, our community is there.'				

The adverbial conjunction *daha* is used in narratives functioning as marker of important events in a narrative. In (31), the speakers uses *daha* to indicate that the event that took place at the 'tapir's creek' is relevant to the story.

(31) a. wehkuyage daha?

weku-ya-ge daa
tapir-CLS:creek-LOC again?
'In the tapir's creek again?'

b. wehkuyagere daha

weku-ya-ge-de daa tapir-CLS:creek-LOC-REF again

'Again, in the tapir's creek.'

Finally, a small class of interjections and ideophones can be identified. The words in this class are distinct from the other classes in that they do not take any type of nominal morphology. Some examples are: *paa* "oh" (surprise), ahau "okay", bee "oh" (admiration), agua "ouch", ũhũ (cursing), ũmã "I don't know". Examples with interjections are illustrated in (32) and (33).

⁶² The iqual symbol '=' indicates that the following morpheme is a clitic.

(32) pa yeta ārīkūmī

pa ye-ta ~adi-kʉ-~bi INTERJ jaguar-EMPH be-DUB-3SG:M.IMPERF 'Ah, it must be a jaguar.'

(33) ahau ãrīpu

aau ~adi-pʉ

INTERJ say-3SG.M.PERF

'Okay, he said.'

3.5 Summary

In this chapter I have defined the notion of 'word' in Desano according to phonological and grammatical criteria. We saw that only two open classes of words can be identified in the language: nouns and verbs. Desano does not have adjectives nor adverbs. Adjectival and adverbial notions are derived from nominal and verbal roots. Other word classes derived from nominal and verbal roots are discussed in Chapter 4.

CHAPTER 4

NOMINAL MORPHOLOGY

4.1 Introduction

In this chapter, I discuss the structure of the noun and noun phrase in Desano. First, a summary of the structural characteristics of nouns is presented in section 4.2. The major types of nouns found in Desano (animate and inanimate) are described in section 4.3. Section 4.3 also describes other subcategories of these major two types and some processes of derivation. Then, the pronouns and other pro-forms are described in section 4.4, followed by a description of other lexical and grammatical morphemes in section 4.5. Finally, section 4.6 presents a description of the structure of noun phrases and types of modifiers. The chapter ends with a summary in section 4.7.

4.2 Structure of Noun Roots

The noun roots in Desano have the following characteristics:

- i. they are bimoraic: /bii/ [bi[?]i] 'rat', /bui/ [bu[?]i] 'agouti';
- ii. they are specified in the lexicon as nasal or oral: /~gapi/ [ŋã^hpĩ] 'sweet potato', /gapi/ [ga^hpi] 'padú powder'; and
- iii. they are lexically (underlyingly) specified for tone (each root requires at least one high tone): [mõå] 'salt', [mmå] 'piraíba fish'.

Items in (i) and (ii) are also true for verbs. (iii) is true only for some verbs, since there are some verbal roots that are not specified for tone (see Chapter 2, Section 2.3.2.4). The great majority of the roots in Desano have a CVCV shape.

4.3 Types of Nouns

Nouns in Desano make a distinction between 'animate' and 'inanimate' entities.⁶³ Animate nouns are subdivided as human and non-human entities. Nouns with nonhuman referents are subdivided as 'individual', i.e., nouns that refer to one single entity (such as fish, monkey, snake); and 'collective' nouns, i.e., nouns that refer to a set of the same type of entity (such as stars, bees, mosquitoes). Singular animate nouns are either feminine or masculine. Plural animate nouns do not have a gender distinction. Inanimate nouns can be noncountable (mass nouns) or countable. Inanimate nouns take classifiers; these are suffixes that denote salient characteristics of the noun they are associated with. Miller listed more than 100 classifiers; most of them designate shape and physical attributes (Miller 1999:35-44). Figure 4.1 shows the hierarchy of nouns in Desano with its meajor features. It should be noted that the features in square brackets have positive and negative values to account for the classes of nouns found in the language. Under the feature [animate], the negative value refers to 'inanimate' nouns. The label 'inanimate' is kept in the chart to identify the node.

⁶³ The 'animate/inanimate' distinction is found in other Easterm Tukanoan languages, for example Tatuyo (Gomez-Imbert 1982, 2007), Tukano (Ramirez, 1997:199) and Wanano (Stenzel 2004:119). This is the pattern in all Easterm Tukanoan languages (cf. Gomez-Imbert 1982, 1996, 2007; Barnes 1999, 2006).



Figure 4.1 General class markers

4.3.1 Animate Nouns

Animate nouns have two subcategories: animate nouns with human referents, and nouns with nonhuman referents. These subcategories differ in their coding for gender and number. Nouns that refer to nonhumans have two subcategories: 'individual' and 'collective' nouns. Table 4.1 shows these subcategories of animate nouns.

	CLA	SS	NU	MBER
HUMAN	MASC -U	FEM -0	SG - <i>gu</i> 'M'	PL
			<i>-go</i> 'F'	-a/-~da
NON-HUMAN	Individua	ıl	Ø	
	Collectiv	e	-~b u	

Table 4.1 Subcategories of animate nouns

4.3.1.1 Nouns with human referents

Nouns with human referents are obligatorily marked for gender. The gender coding suffixes are -u/-gu '3 SINGULAR MASCULINE' and -o/-go '3 SINGULAR FEMININE', which occur with nouns with human referents. These ending vowels can be analyzed as the result of fusion of the gender coding suffixes on the noun root. Thus, the nouns in (1) and (2) can be analyzed as being grammaticalized as inherently feminine (1) or inherently masculine (2). ⁶⁴ These forms may have natural (inherent) gender, but it still looks very much like they bear the masculine or the feminine suffix, the final vowel.

(1) INHERENTLY FEMININE

a.	yẽhk õ	'grandmother			

- b. bur**o** 'old woman'
- c. mẽ**õ** 'mom'

(2) INHERENTLY MASCULINE

a.	yẽhk ũ	'grandfather'
b.	bur u	'old man'
C.	kũm ũ	'faith healer'
d.	ũ m ũ	'man'

However, there are some nouns that are masculine but end in vowels other than *u*,

as illustrated in (3).

(3) a. kãmĩ 'younger brother'
b. wãĩ 'uncle'
c. yee 'shaman'

⁶⁴ This analysis of morpheme reduction and fusion follows that of Gomez-Imbert (1982:113) for Tatuyo and Stenzel (2004:129) for Wanano.

Some noun roots do not occur without the gender suffixes -o/-go, '3 SINGULAR FEMININE'; -u/-gu, '3 SINGULAR MASCULINE', as in (4); or without number, -a/-~da'PLURAL ANIMATE' as the examples in (5). It should be noticed that the gender markers in (4) also function to signal the singular of these nouns. An alternative analysis would be to consider '-g' the singular marker, and '-o/-u' the feminine/masculine gender markers.

(4)	ROOT		FEMININE		MASCULINE		
	a. p	aa-	'progenitor'	pa go	'mother'	pa gu	'father'
	b. m	nãã-	'progeny	mã gõ	'daughter'	mã gũ	'son'
	c. n	nãhĩ-	'child'	mãhĩgo	'girl'	mãhĩgu	ı 'boy'
	d. b	uu-	'elder'	bu go	'old woman'	bu gu	'old man'
(5)	ROOT	Г		PLURA	L		
	a. n	nãhĩ-	'child'	mãhĩ rã	'children'		
	b. b	uu-	'elder'	mũ rã	'elders'		

As shown in the examples above, animate nouns also bear marking for plural. It should be noted that the phoneme /d/ is represented by < r > in the orthography. Plural nouns are marked with the plural suffix -a or $-r\tilde{a}$ (/-~da/) as illustrated in (6). The plural morpheme /-a/ is a reduced form of /-~da/.⁶⁵

(6)	a.	yee	ʻshaman'	yea	'shamans'
	b.	ũmũ	'man'	ũ m ã	'men'
	c.	kãmĩ	'young brother'	kãmĩ ã	'young brothers'

⁶⁵ According to Gomez-Imbert (1982:113) the plural suffixes /-a/ and /-~da/ are the same suffix. Stenzel (2004: 131) also treats them as the same in her analysis of Wanano. Thus, /-a/ is reduced from /-~da/.

Animate masculine nouns are marked for plural in this way. Most plural feminine nouns have this form of the plural followed by $n\tilde{o}m\tilde{e}$ – the lexical root for 'female'. This is illustrated in (7).

(7) mũrã 'older men' vs mũrã nõmẽ 'older women'

Nouns referring to some kinship terms are also marked with $s\tilde{a}m\tilde{a}$ 'vagina' followed by the plural suffix $-r\tilde{a}$ /-~da/, as shown in (8). In (8b) the word $s\tilde{a}m\tilde{a}$ has been reduced to $s\tilde{a}$.

(8)	a. yēhkū 'grandfather' vs	yēhkūsāmārā 'grandparents'
	b. yẽhkõ 'grandmother' vs	yẽhkõsã nõmẽ 'grandmothers'

4.3.1.2 Nouns with nonhuman referents

As mentioned above, nouns with nonhuman referents can be subclassified as 'individual' and 'collective' nouns. Individual animate nouns refer to a single entity (for example 'monkey', 'fish', 'deer'); whereas 'collective' nouns refer to a group of the same entity, usually beings that live and move together – these are usually insects, some animals, and other entities such as 'stars'. Individual nouns with nonhuman referents can be further subclassified to distinguish nouns that refer to 'higher level' animates from those that refer to 'lower level' animates.⁶⁶ I describe each of these subsets in the following order: higher level individual animates; lower level individual animates; and collective animate.

⁶⁶ The analysis distinguishing between 'higher level individual' animates and 'lower level individual' animates follows Stenzel's (2004) analysis of Wanano.

4.3.1.2.1 Higher level individual animates

These nouns are similar to nouns with human referents in that they can be marked for gender in their singular form, where female is the marked feature, as shown in the examples in (9). Alternatively, these nouns could just be treated as exceptional members of the 'human' class. It may well be that cognitively, some of these are thought to have human attributes, be human: 'evil spirit' certainly so, 'jaguar' probably; 'dog' by metaphoric extension. Futhermore, because these nouns take the same endings as the human nouns, they can be considered to belong to that class.

		SINGULAR	FEMININE SINGULAR	PLURAL
(9)	a.	ye	ye-go	ye- a
		ʻjaguar'	jaguar-3SG:F	jaguar-PL.AN
			'female jaguar'	'jaguars'
	b.	diaye	diaye-go	diaye- a
		'dog'	dog-3sg:f	dog-PL.AN
			'female dog'	'dogs'
	c.	wãhtĩ	wãhtĩ- gõ	wāhtī-a
		'evil spirit'	evil.spirit-3sg:F	evil.spirit-PL.AN
			'female evil spirit'	'evil spirits'

It should be noticed that the bare unmarked singular form is inherently masculine. In general, Desano speakers talk about these animals and beings without making references to their gender. However, they point out that these nouns can be optionally signaled as female with the female gender marker -go. Another similarity that these nouns have with the nouns in the human animate category is that they also mark plural with *-a*. Like in the human animate nouns, the plural suffix also codes masculine gender. In order to express plural for females, the word *nome* 'female' follows the singular feminine noun, as illustrated in (10).

(10)	a.	yee-go	yee-a nõmẽ
		jaguar-3sg:F	jaguar-PL:AN female
		'female jaguar'	'female jaguars'
	b.	diaye- go	diaye -a nõmẽ
		dog-3sg:f	dog-PL:AN female
		'female dog'	'female dogs'

Not all nouns referring to animals are linguistically marked in a way similar to how human animate nouns are marked. Gomez-Imbert (1996:456) states that "some animals are closer to human beings than others, and this is reflected at the linguistic level." The jaguar in the Desano myths, for example, is a deity that was created by the sun (Reichel-Dolmatoff 1971). For most of the Tukanoan groups, there is also a mythical association between the jaguar and the shaman (Jackson 1983: 197). Dogs are pets and hunting companions, and considered an important animal for the Desano people.⁶⁷

4.3.1.2.2 Lower level individual animates

The members of the lower level individual animates category are most of the nonhuman animates (i.e. animals). Members of this category are not overtly marked for gender, i.e., they do not take gender markers. The plural form is marked with the suffix -a

⁶⁷ Here I follow the analysis of Stenzel (2004) for Wanano. She notes that 'Dogs ... play an extremely important role in Wanano daily life both as pets and as hunting companions, making them the most important animal in Wanano communities and one with which humans develop a closer relationship' (p. 134).

'plural animate', in common with human animates. Examples of the members of this class are given in (11).

		SINGU	LAR	PLURAL	
(11)	a.	pãmũ	'armadillo'	pãmũ ã	'armadillos'
	b.	yahi	'heron'	yahi a	'herons'
	c.	pĩrũ	'snake'	pĩrũ ã	'snakes'
	d.	kero	'firefly'	kero a	'fireflies'
	e.	gahki	'monkey'	gahki a	'monkeys'
	f.	bu'i	'agouti'	bu'i a	'agoutis'

However, when the nouns in (11) occur as Subject of a clause they trigger verb agreement (with 'masculine' being the default value), as shown in (12) and (13) below. In (12a) and (13a), the singular nouns *gahki* 'monkey' and *bu'i* 'agouti', respectively, require singular verbal agreement. In (12b) and (13b), these nouns are plural and require the plural verbal agreement.

(12) a. gahki mühtägügä ärämī

gaki	~buta-~gu-~ga	~adi-a-~bi
monkey	be.small-3sG:M-DIM	be-PERF-3sg:m.IMPERF
'The monkey	was very small.'	

b. gahkia mühtaraga arīra

gaki -a	~buta-~da-~ga	~adi-~ da
monkey-PL:AN	be.small- PL:AN- DIM	be-pl:An.perf

'The monkeys are very small.'

(13) a. bu'i duriamĩ

bui dudi-a-~bi
agouti flee-PERF-3SG:M.IMPERF
'The agouti fled.'
b. bu'ia duriamã

bu'i-a dudi-a-~ba agouti-PL:AN flee-PERF-3PL:AN.IMPERF 'The agoutis fled.'

It should also be noted that although these nouns are overtly unmarked for gender (genderless), they require verbal agreement in the third person masculine. This show that 'masculine' is the default gender for genderless nouns. These examples support Kaye's (1971) claim that "masculine is the unmarked member of the gender category" (p. 92).

4.3.1.2.3 Collective animates

The members of this category refer to animate nouns that are collectives. They refer to animals that are typically encountered in a group (or seen as a group). These nouns are inherently plural, i.e., plural is the unmarked category. To refer to a single member of the collective, the singularizing suffix $-m\tilde{u}$ /-~bu/ is used to individuate and refer to it, as illustrated in (14). It should be noticed that (14e) refers to a celestial entity and is a member of the animate class because its status as an animate being in the Desano mythology.

(14)	a.	mẽgã	'maniuaras'	mēgā mū	'a maniuara' ⁶⁸
				maniuaras-IND)
	b.	ĩã	'worms'	ĩã mũ	'a worm'
				worms-IND	
	c.	mõmẽ	'bees'	nõmẽ mũ	'a bee'
				bees-IND	
	d.	uhti	'wasps'	uhti mũ	ʻa wasp'
				wasps-IND	
	e.	nẽhkã	'stars'	nẽhkã mũ	ʻa star'
				stars-IND	

In her analysis of Wanano, Stenzel (2004:138) suggests that animate nouns can be represented in a hierarchy according to the ways they are coded linguistically for gender and number. The hierarchy of animate nouns in Desano is shown on Table 4.2. Human animate nouns are higher in the hierarchy because they are coded for the largest number of features (gender and number), followed by higher level individual animates, which are optionally marked for feminine when singular (singular and plural forms are inherently masculine). The penultimate category in the hierarchy is the lower-level individual animates, which are not overtly marked for gender; singular is expressed by the bare root. The last (and lowest) category in the hierarchy is the collective animate nouns, which do not mark gender and whose roots are inherently plural.

⁶⁸ 'Maniuara' is a type of edible ant.

	HUMANS >	HIGHER-LEVEL IND >	LOWER-LEVEL IND >	COLLECTIVES
GENDER	obligatory	optionally marked for	not possib	ole
		'feminine'		
SG		-до/-дн	bare root	-~bu
PL		-a/-~da	1	(inherently
				plural)

The present analysis is based on agreement classes as proposed in Corbertt (2007), in which noun classes can be distinguished syntactically according to the agreement they take. However, while Corbett treats noun classes and classifiers as different phenomena, I treat them as part of the same system of nominal classification – this is also the approach adopted by Gomez-Imbert (2007b) for the analysis of Tatuyo, also an Eastern Tukanoan language.⁶⁹

⁶⁹ My analysis follows Seifart's (2005) approach to analyzing the nominal system for Miranã (Witotoan), later adopted by Gomez-Imbert (2007b) for the analysis of Tatuyo. Current studies in Tukanoan languages (and other Amazonian languages as well) also show that it might be the case that there is only one system of nominal classification. Stenzel (2004:125) describes Wanano as having both noun class morphology and noun classifiers within a single system.

4.3.2 Inanimate Nouns

Inanimate nouns are subdivided into countable (count) and non-countable (mass) nouns. Count nouns are further subdivided according to the way they mark number. Table 4.3 below shows the subcategories for inanimate nouns.

4.3.2.1 Mass Nouns

Mass nouns, or noncount nouns, are not marked for gender or number. The bare root of these nouns signals an entity that is not countable -- referring to mass substances such as liquid, powder, collection of units (i.e., vegetal species). Mass nouns differ from count nouns in that they are not morphologically marked for plural, as illustrated in (15).

- (15) a. dehko 'water'
 - b. mõã 'salt'
 - c. yeba 'ground'
 - d. poga 'manioc flour'

One could speculate whether the final vowel -*o* in (15a) and -*a* in (15b-d) are gender marker and plural markers, respectively. The sentences in (16) show that that is not the case. These mass nouns do not require the verb to agree neither for masculine/feminine gender nor for singular/plural number, as shown in (16a) for *dehko* 'water' and (16b) for *poga* 'manioc flour'. They both require nonthird person verbal agreement (see Section 5.5.7).

(16) a. dehko ahsiro ia

deko asi-do i-a water be.hot-NON3.IMPERF do-PERF 'The water is hot.'

b. poga weawadero ārābu

poga	wea-wa-de-do	~adi-a-bʉ
manioc.flour	wet-go-PERF-NON3.IMPERF	be-perf-non3.perf
'The manioc f	lour was wet.'	

There are many mass nouns that are derived from verb roots by adding the nominalizing suffix -ri /-di/.⁷⁰ Some of them are shown in (17).

(17) a. ba-ri 'food'

eat-NOM:IN

- b. ohte-ri 'plantation' plant-NOM:IN
- c. bayi-ri 'blessing'

bless-NOM:IN

4.3.2.2 Count Nouns

Inaninate count nouns can be divided according to how they inflect for number. Although most inanimate count nouns follow a regular pattern of inflecting for number, some are irregular. For example, some unsuffixed noun roots are inherently plural and others are inherently singular (or have general reference). Inanimate count nouns that are inherently singular are pluralized with the suffix *-ri* /-di/. Inherently plural nouns do not take the pluralizer suffix. Furthermore, count nouns that are inherently plural are

⁷⁰ In some Eastern Tukanoan languages, such as Wanano (cf. Stenzel 2004), this suffix generally occurs to derive animate nouns. In Desano, -ri /-di/ seems to function like it does in Tukano (cf. Ramirez 1997), more for deriving inanimates.

singularized (or individualized) by means of specific class marker suffixes (classifiers) which are attached to the root.⁷¹ Table 4.3 summarizes the characteristics of unsuffixed roots and the way they inflect for number.

Table 4.3 Unsuffixed inanimate noun roots

Inanimate	u	nsuffixed ro	ot	take	take
Nouns				plural suffix <i>-ri</i>	classifier (sg form)
	inherently plural	inherently singular	generic reference		
gahsi		1		i	
'canoe'	\checkmark				~
weha					
'paddle'	~				~
yãhpĩ					
'sweet potato'		\checkmark		~	
wehka					
'skewer'		~		~	
oho					
'banana'			\checkmark	1	✓
mîhĩ					
'açaí berry'			~	~	~

⁷¹ Typologically, individualization is one of the main functions of nominal classification (cf. Craig 1992: 295; Croft 1994: 162).

It should be noted that, the inanimate nouns can be subgrouped in three subsets according to the unsuffixed noun roots. I discuss the characteristics of these subsets in the following sections.

4.3.2.2.1 Inherently plural count nouns

I treat the inherently plural forms as 'irregular' plurals. These nouns do not form a large group, when compared to the nouns that are pluralized with the regular plural suffix *-ri* /-di/.

Borrowed inanimate nouns form their plural with the regular plural suffix -*ri* /-di/, as illustrated in (18).

(18)	a.	sabadu	'Saturday'	\rightarrow	sabadu-ri	'Saturdays'
	b.	proyetu	'project'	\rightarrow	proyetu-ri	'projects'

As illustrated in (19a) below, the unsuffixed noun *gahsi* codes plural, whereas (19b), the specific class marker -ru /-du/ 'round/oval/oblong' has a singularizing function, individualizing the noun and making it as a singular noun.

(19) a. gahsi eheayoro

gasi ea-yo-do
canoe arrive-EVID:HSAY-NON3.PERF
'The canoes arrived.'
b. gahsiru eheayoro
gasi-du ea-yo-do
canoe-CLS:round/oval/oblong arrive-EVID:HSAY-NON3.PERF

'The canoe arrived.'

4.3.2.2.2 Inherently singular count nouns

Inherently singular nouns are much more common and they take the regular plural suffix -*ri* /-di/. Some examples are given in (20).

(20)	a.	wi'i	/wii/	'house'	wi'i -ri	/wii-di/	'houses'	
------	----	------	-------	---------	-----------------	----------	----------	--

b. wahka /waka/ skewer wahka-ri /waka-di/ skewer
--

c. yāhpī /~yapi/ 'sweet potato' yāhpī-rī /~yapi-di/ 'sweet potatoes'

Many inanimate noun roots denoting body parts have the suffix *-ro* /-do/ in their singular form, as in the examples in (21). These roots are pluralized with the plural suffix *-ri*, in accordance with the plural for inanimate nouns, as shown in (21). The suffix *-ro*, attaching to nouns for body parts is also found in Tukano (Ramirez 1997) and Wanano (Stenzel 2004) and is analyzed, in those languages, as a 'partitive' suffix, which derives a part from a whole. However, for Desano, I analyze *-ro* as a classifier coding 'body parts', as illustrated in the examples in (21). The same suffix has been described by Miller (1999:42) as a classifier for 'certain body appendages'.

(21)	a.	gāmī -rõ	/~gabi-do/	gãmĩ- rĩ	/~gabi-di/
		'ear-CLS:body	v.part'	'ears'	
	b.	dihsi -ro	/disi-do/	dihsi- ri	/disi-di/
		'mouth- CLS:body.part'		'mouths'	
	C.	kui -ro	/kui-do/	kui -ri	/kui-di/
		'eye-CLS:body.part'		'eyes'	
	d.	ãhpĩ- rõ	/~api-do/	ãhpĩ- rĩ	/~api-di/
		'nipple-CLS:b	ody.part'	'nipples'	

e.	pĩgõ- rõ	/~pigo-do/	pĩgõ- rĩ	/~pigo-di/
	'tail-CLS:bod	y.part'	'tails'	

It should be noted that the suffix *-ro* is attached to inanimate nouns and functions as a 'singularizer', two characteristics of specific class markers (classifiers). Thus, the term 'CLASSIFIER: body part' is used for the morpheme *-ro* that occurs on body parts nouns in Desano. The term PARTITIVE has a more generally accepted meaning for a very different function (it refers to partially but not wholly affected nouns, for example as a case affix for not fully affected objects of transitive verbs).

It is important to notice that not all inanimate nouns denoting body parts are singularized by the specific class marker *-ro*. The body parts listed in (22), for example, do not take *-ro*. Instead, they are singularized by specific class markers coding 'shape', which are affixed to the noun root.

- (22) a. dihpu-**ru** /dipu-du/ 'head' head-CLS:round/oval/oblong
 - b. ĩgĩ-rũ /~igi-du/ 'nose'
 nose-CLS:round/oval/oblong
 - c. puru-pu /pudu-pu/ 'back'

back-CLS:body

d. wũyũ-gu /~wuju-gu/ 'neck'

neck-CLS:cylindrical

Comparing the examples in (21) with the examples in (22), we notice that some nouns referring to body parts can take different classifiers for different circumstances (or contexts). I hypothesize that body part nouns in (22) might just be the ones where

121

specific noun class markers coding 'shape' overshadowed the *-ro* 'body part' one, due to shape saliency. Use of shape classifiers is generally used with nouns which have a 'salient shape'. This is the case with exceptional animate nouns that receive shape classifiers, such as *bugu-bu* 'anteater-CLS:concave'.

Some nouns referring to body parts can also have other specific class marked (usually coding 'shape') suffixed to them. For example, there is a local bird called $p\tilde{r}g\tilde{o}s\tilde{e}$ (from $p\tilde{r}g\tilde{o}$ 'tail' and -*se* 'CLS:Y.shape') whose main characteristic is its 'Y shape' tail (some speakers refers to this bird's tail as 'scissors tail').

4.3.2.2.3 Unsuffixed roots with generic reference

Unsuffixed inanimate roots can also have generic reference, as in (23a). These nouns can be individualized (or singularized) by the specific class marker they may take, as in (23b-e).

(23)	a.	/~bii/	[mĩʰĩ]	<mihi></mihi>	'açaí fruit' (generic)		
	b.	/~bii -du /	[mĩʰĩɾũ]	<mihiru></mihiru>	'an açaí fruit'		
		açaí.fruit-CLS:round/oval/oblong					
	C.	/~bii-~ to /	[mīʰītõ]	<mĩhĩtõ></mĩhĩtõ>	'a bunch of açaí fruit'		
	açaí.fruit-CLS:bunch						
	d.	/~bii-~yu/	[mīʰīɲū]	<mĩhĩñũ></mĩhĩñũ>	'açaí fruit tree'		
		açaí.fruit-CLS:palm.of					
	e.	/~bii -~du /	[mĩʰĩnũ]	<mîhînũ></mîhînũ>	'açaí fruit season'		
		açaí.fruit-CLS:day					

The specific class markers (classifiers) in Desano can be divided into three subsets according to their shape and function (lexical versus grammatical):⁷²

i. CV suffixes that are attached to different types of noun roots.

ii. CV suffixes that can be analyzed as reductions of full noun roots.

iii. CV(C)V noun roots that can generally stand as free morphemes.

The list below gives a sample of inanimate specific class markers. This is not an exhaustive list of all classifiers found in Desano. It shows the most commonly recurring class markers in the corpus.

4.3.2.2.4 A sample of specific noun markers

 $/-gu/ \langle gu \rangle$ cylindrical/trunklike. This specific noun marker is used to classify most trees. It is used to refer to living trees (i.e., trees that are still standing), as in (24ab).⁷³ This classifier is generally used for wooden objects with cylindrical shape. Other nouns that this classifier appears with are those designating things that stand vertically (post, fence), including the noun for statue, as in (24d). This is also the classifier used for 'neck', illustrated in (24e), which could be explained by a metonymic relation as it is in vertical position as a trunk.

(24) a. /yuku-gu/ [yu^hkugu] <yuhkugu> 'a tree' trunk-CLS:cylindrical/trunklike

⁷² This analysis is based on Stenzel's (2004) and Gomez-Imbert's (2007b) analyses of the classifier systems of Wanano and Tatuyo, respectively.

 $^{^{73}}$ According to one of my main consultants, if a tree is fallen on the ground (dead), one refers to it as *yuhku* 'trunk'. Stenzel (p.c.) notes that, in Wanano, the 'vertical' position more than the actual shape (or maybe the combination) is really the salient feature of nouns marked with -gu.

- b. /wua-di-gu/ [wuarigu] <wuarigu> 'a big tree'
 be.big-NOM:IN-CLS:cylindrical/trunklike
- c. /pea-gu/ [peagu] <peagu> 'rifle' log-CLS:cylindrical/trunklike
- d. /wea-di-gu/ [weadigu] <weadigu> 'statue'
 clay-CLS:meat-CLS:cylindrical
- e. /~wʉyʉ-gʉ/ [wʉ̃nʉ̃gʉ] <wʉ̃yʉ̃gʉ> 'neck' neck-CLS:cylindrical/trunklike

/-**du**/ <*ru*> *round/oval/oblong*. This specific class marker is used with concrete nouns designating objects that have round/oval/oblong shapes. There are many different entities that are classified with this class marker: vessels (canoe, baskets, and gourds); rounded/oval fruits (pineapple, buriti fruit [*Mauritia flexuosa*], açaí fruit, etc.) and their seeds; it is also the class marker used to classify other items such as 'eggs', 'lake', 'sand', '(round) fruits'. Examples of nouns classified by *-ru* is shown in (25).

- (25) a. /gasi-du/ [ga^hsiru] <gahsiru> 'a canoe' canoe-CLS:round/oval/oblong
 b. /weku-du/ [we^hkuru] <wehkuru> 'a balaio basket⁷⁴ balaio.basket- CLS:round/oval/oblong
 - c. /~sera-du/ [sērārū] <sērārū> 'a pineapple'
 pineapple- CLS:round/oval/oblong
 d. /~de-du/ [nēērū] <yērū> 'a buriti fruit'

buriti.fruit-CLS:oval/oblong

⁷⁴ A 'balaio' basket is a specific type of basket used to carry manioc. Other general baskets are classified with -bu 'concave/convex'.

- e. /dita-du/ [di^htaru] <dihtaru> 'a lake' lake- CLS:round/oval/oblong
- f. /diu-du/ [diuru] <diuru> 'an egg' egg-CLS:round/oval/oblong

/-**bu**/ \langle *bu* \rangle *concave/convex*. This specific class marker is used with concrete entities denoting concave/convex and wavy shapes. These include some natural entities such as rapids and animals with some shape saliency such as anteater, mutum bird, jacamim bird. Although class markers are generally used for inanimate nouns, animals which have a salient shape usually take a shape classifier.⁷⁵ Another noun in this class is 'basket', except for the 'balaio' basket which is classified with the class marker -*ru* 'oval/oblong'. Examples of nouns classified with -*bu* are shown in (26).

- (26) a. /~uta-bu/ [ũ^htãmũ] <ũhtãmũ> 'a rapid' stone-CLS:concave/convex
 - b. /pui-bu/ [puibu] <puibu> 'a basket'
 basket-CLS:concave/convex
 - c. /kada-bu/ [karabu] <karabu> 'a mutum bird'
 mutum.bird-CLS:concave/convex
 - d. /~mo-di-bu/ [mõ^hõdibu] <mõhõdibu> 'an anteater' hand-meat-CLS:concave/convex

Besides the shape class markers presented above, there are also other types of class markers that describe specific characteristics or configurations of the nouns they are attached to. These are the following.

⁷⁵ In this aspect, Desano is similar to Kubeo (also an Eastern Tukanoan language) and Arawakan languages in the region in which animate nouns receive a classifier.

/-~yu/ <yu> palm tree. This specific class marker is used to designate trees that have palm-like leaves. The class marker is attached to the noun root designating the fruit that grows on the palm tree in order to make the noun corresponding to the tree itself. This is shown in (27).

(27)a. $/\sim de - \sim vu/$ [nẽẽnũ] <nēyū> 'buriti palm tree' buriti.fruit-CLS:palm.tree b. /oo-~yu/ [o^hopũ] <ohoyū> 'banana palm tree' anana-CLS:palm.tree c. i-~vu/ [mīhīpū] <mîhīyū> 'açaí fruit palm tree' çaí.fruit-CLS:palm.tree

/-~to/ <to> palm. This specific class marker is used with all noun roots for kinds of fruits that grow on palm trees. It is also used with the noun for 'hand' (it should be noticed that all the examples in 28 have in common the 'hand-like' shape). Examples in (28) show the use of this class marker.

(28)a. /oo-~to/ [o^hotõ] <ohotõ> 'palm of banana' banana-CLS:palm 'palm of açaí fruit' b. /~bii-~to/ [mī^hītõ] <mīhītõ> açaí.fruit-CLS:palm c. /~boo-~to/ <mõhõtõ> 'a hand' [mõ^hõtõ] hand-CLS:palm

/-da/ <da> *thread*. This specific class marker is used with nouns that refer to rope-like objects (hair, fishing line, or other fibrous material). It is also used with the

noun root 'tail' to derive the noun 'rainbow'. Examples of nouns with this class marker are shown in (29).

- (29) a. /poa-da-di/ [poadari] <poadari> 'hairs' hair-CLS:thread-PL.IN
 - b. /~pigo-da/ [pīŋūda] <pīgõda> 'a rainbow'
 tail-CLS:thread

 $/-\sim d\mathbf{u}/\langle d\mathbf{u} \rangle day/season$. This specific class marker is used with nouns to refer to time (day or season). In (30a), for example, the expression 'a good day' is derived from a nominalized verbal root $\sim waa$ 'be good' and the noun classifier for day $-\sim du$. In (30b), the noun root $\sim uri$ 'pupunha fruit' is used to derive the noun for 'pupunha season'.

- (30) a. /~waa-di-~du/[wārīnū] <wārīnū> `a good day'
 be.good-NOM:IN-CLS:day/season
 - b. /~udi-~du/ [ũrĩnũ] <ũrĩnũ> 'season of pupunha fruit'
 pupunha.fruit-CLS:day/season

/-tode/ <*tore*> *hollow*. This specific class marker is used with nouns that have a hollow configuration. Some examples are given in (31).

(31) a. /pa-tode/ [paatore] <patore> 'belly' belly-CLS:hollow

b. /yuku-tode/ [yuhkutore] <yuhkutore> 'hollow tree'⁷⁶
 trunk-CLS:hollow

 $/-\sim ye/\langle ye \rangle$ tangled. This specific class marker is used with nouns denoting things that are 'tangled', when referring to concrete nouns, as in (32a); it can also refer to

⁷⁶ *yuhkutore* refers to a dead tree, i.e., a tree fallen on the ground. To refer to a hollow tree that is still standing, *yuhkugutore* is used.
abstract nouns, referring to an 'amount of things'. In (32b), for example, the classifier – $\sim ye$ is used with the demonstrative to derive 'this (story)' at the end of a narrative containing several plots.

(32)	a.	/~budu-~ye/	[mʉ̃rʉ̃nẽ]	<mũrũyē></mũrũyē>	'tobacco'
		tobacco-CLS:t	angled		
	b.	/i-~ye/	[inē]	<iyē></iyē>	(all of) this

DEM:PROX-CLS:tangled

The following noun class markers can be analyzed as phonologically reduced forms from full noun roots. They are not grammaticalized classifiers, as they can occur both as root morphemes and as (classifier) suffixes.

/-ya/ <ya> creek. This specific class marker is a reduced form of the noun yaa 'creek'. As noun, ya /ya/ is realized as [yaa], with a two mora structure which is reduced to one mora when used as a class marker, being realized as /-ya/.⁷⁷ The status of yaa as a noun can be seen in example (33a), in which it occurs as a root noun to which other suffixes are attached. As a class marker, -ya is used with noun roots to derive creek names, as illustrated in (33b-c). This class marker can also be attached to nominalized descriptive verbal roots, as shown in (33d).

- (33) a. /ya-ge/ [yage] <yage> 'in the creek' creek-LOC
 - b. /~diba-ya/ [nīmānā] <nīmāyā> 'poison creek'
 poison-CLS:creek

⁷⁷ See Section 2.3.1.2 on syllable shape and quantity, in Chapter 2.

- c. /weku-ya/ [we^hkʉya] <wehkʉya> 'tapir creek' tapir-CLS:creek
- d. /guya-di-ya/ [guyariya] <guyariya> 'dangerous creek'
 be.dangerous-NOM:IN-CLS:creek

/-ku/ < ku > tree. This specific class marker is a reduced form of the full noun root yuhku 'tree', from which the first syllable is deleted, as shown in (34a). This class marker seems to be in the process of being grammaticalized. Instances in which the full noun root is used are also common, as in (34b).

- (34) a. /~budu-ku/ [mūrūku] <mūrūku> 'a tobacco plant' tobacco-CLS:tree
 - b. /~budu-yuku/ [mũrũyuʰku] <mũrũyuhku> 'a tobacco plant' tobacco- CLS:tree

/-wi/<wi> house. This specific class marker is a reduced form of the full noun root wi'i 'house'. Although this reduced form is used with some noun roots, as in (35a), the full noun root also appears frequently attached to other nouns creating a compound, as in (35b).

- (35) a. /baya-wi/ [bayawi] <bayawi> 'a house of dance'dance-CLS:house
 - b. /abe-wii/ [abewi'i] <abewi'i> 'house of the sun' sun-house

4.3.3 Compound Nouns

Compounding is a productive way to form new words. Compounds can be formed from two noun roots as shown in the examples in (36). It should be noticed that the semantic head of the compounds is the leftmost root.

- (36) a. /~saya-di-bedo/ [sãnãribero] <sãyãribero> 'a ring'
 dress-NOM:IN-CLS:circle
 - b. /~baa-poa/ [mãhāpoa] <mãhāpoa> 'monkey's hair' monkey-hair
 - c. /~yapi-pui/ [nã^hpípuí] <yãhpĩpui> 'a sweet potato basket'
 sweet.potato-basket
 - d. /abe-wii/ [abéwi²i] <abewi²i> 'the sun's house' sun-house

4.3.4 Class Markers and the Grammaticalization Continuum

The specific class markers (classifiers) fall in a continuum somewhere between lexical items and grammatical morphemes.⁷⁸ This continuum between lexical and grammatical for classifiers was proposed by Grinevald (2002). This grammaticalization continuum shows that some lexical items (full nouns) can go through a process of becoming full grammatical items (noun-class suffixes). Table 4.4 gives a sample of nouns and noun-class markers according to the grammaticalization continuum.

⁷⁸ This analysis is consistent with Stenzel's (2004) analysis of Wanano and Gomez-Imbert's (2007b) analysis of Tatuyo.

	LEXICAL					GRAMMATICAL
	ROOTS		REDUCED	ROOTS		SUFFIXES
/yʉkʉ/	[yʉʰkʉ]	tree	/-kʉ/ [-kʉ]	tree	/-gʉ/ [-gʉ]	CLS:cylindrical
/wii/	[wi [?] i]	house	/-wi/ [-wi]	creek	/-~to/ [-tõ]	CLS:palm
/~ya/	[yãã]	creek	/-ya/ [-ya]	house	/-du/ [-du]	CLS:oval/oblong
					/-~du/ [-n \tilde{u}]	CLS:day/season
					/-bu/ [-bu]	CLS:concave/convex

Table 4.4 Grammaticalization continuum of class markers

4.3.5 Derived Nouns

Verbal roots can be nominalized when the nominalizing suffix for inanimate nouns, -ri /-di/ for general inanimate nouns and -ro /-do/ for (inanimate) abstract nouns, are attached to them. This seems to be a productive process of derivation. Examples in (37) illustrate nouns derived with the nominalizer -ri /-di/. Nouns derived with the nominalizer -ro /-do/ are shown in (38). Discussions of derived animate and inanimate nouns are presented in the following subsections.

(37)	a.	/ba- di /	[bari]	<bari></bari>	'food'
		eat-NOM:IN			
	b.	/wede- di /	[wereri]	<wereri></wereri>	'language'
		tell-NOM:IN			

	C.	/bue-di/	[bueri]	<bueri></bueri>	'study'
		study-NOM:IN			
	d.	/bayi -di /	[bayiri]	<bayiri></bayiri>	'blessing'
		bless-NOM:IN			
(38)	a.	/~baa-do/	[mãʰãrõ]	<māhārð></māhārð>	'enough'
		many-NOM:AI	BSTR		
	b.	/pede-do-de/	[pererore]	<pereore></pereore>	'at/in/on the end'
		end-NOM:ABS	TR-REF		
	c.	/~wa-do/	[wãrõ]	<wārõ></wārõ>	'good'
		be.good-NOM	ABSTR		
	d.	/pepi-do/	[pepiro]	<pepiro></pepiro>	'thought/feeling'
		think-NOM:AE	BSTR		

4.3.5.1 Derived Animate Nouns

Table 4.5 summarizes the process of nominalization of verb roots resulting in animate nouns.

Table 4.5 Nominalization resulting in animate nouns

	Noun classification morphology	
	General class nouns (for animates)	
Verb Root +	-gu/-go/-~da	
	(Masculine/Feminine/Plural)	

Animate nouns derived from verbal roots do not require the nominalizer morphemes /-di/ 'inanimate concrete' nor /-do/ inanimate abstract' to be attached to the verb root. Verb roots suffix general class morphology for animate nouns directly and these suffixes reveal that the form has been nominalized.

This is illustrated in (39) where the animate nouns are derived from the verb *bue* 'to study'. Example in (39a) gives the masculine form; the feminine form is given in (39b), and the plural form is given in (39c).

- (39) a. /bue-gu/ [buegu] <buegu> 'male student' study-38G:M
 - b. /bue-go/ [buego] <buego> 'female student' study-3sg:F
 - c. /bue-~da/ [buerā] <buerā> 'students' study-PL:AN

Examples in (40) show additional animate nouns derived from verbal roots.

(40)	a.	/guya- gu /	[guyagʉ]	<guyagu></guyag	'dangerous male'
		be.dangerous-	3sg:m		
	b.	/guya -go /	[guyago]	<guyago></guyago>	'dangerous female'
		be.dangerous-	3sg:f		
	c.	/guya-~ da /	[guyarã]	<guyarã></guyarã>	'dangerous ones'
		be.dangerous-	PL:AN		
	d.	/kede wede-	gu/ [kere wereg	gu] <kere th="" wereg<=""><th>su> 'story teller (male)'</th></kere>	su> 'story teller (male)'

story tell-3SG:M

 e. /kede wede-go/ [kere werego] <kere werego> 'story teller (female)' story tell-3sg:F

In the examples above, the noun-class markers, when attached to verb roots, appear to function as suffixes for 'agentive nominalization', where agents are typically animate entities.

4.3.5.2 Derived Inanimate Nouns

As mentioned above, and illustrated in (37) and (38), inanimate nouns derived from verb roots require the nominalizers suffixes /-di/, for inanimate nouns, or /-do/ for inanimate abstract nouns, attached to the verb root. In order to derive concrete inanimate nouns, specific class markers are attached to the stem formed by Root + either /-di/ or /do/. It should be noted that the nominalizer /-do/ is not followed by the specific class markers (classifier) nor by the plural suffix for inanimates /-di/. This process is summarized in Table 4.6.

Table 4.6 Derived inanimate nouns

	nominalizer	noun classification morphology
Verb Root +	1 • .	
	-di +	specific class markers (singular
	'inanimate:concrete'	inanimates) or -di (plural inanimate)
	-do	
	'inanimate:abstract'	

In the examples in (41) below, the derived inanimate nouns have specific class markers (classifiers) that individualize the noun (making it more specific).

- (41) a. /guya-di-ya/ [guyariya] <guyariya> 'dangerous creek'
 be.dangerous-NOM:IN-CLS:creek
 - b. /~saya-di-bedo/ [săpăribero] <sāyāribero> 'ring' wear-NOM:IN-CLS:circle
 - c. /wee-di-du/ [we^heriru] <weheriru> 'fishing hook'
 kill-NOM:IN-CLS:concave

Verb roots that have a descriptive sense can be used to derive animate nouns (42a), inanimate specific nouns (42b), and inanimate generic nouns (42c).

(42) a. ĩgũ guyagu ãrĩmĩ

	~igʉ	guya-g u	~adi-~bi	
	3sg:m	be.dangerous-3sg:M	be-3sg:m.IMP	ERF
	'He is	dangerous.'		
b.	dia guy	variya ãrã		
	dia	guya-di-ya		~adi-a
	river	be.dangerous-NOM:II	N-CLS:creek	be-perf
	'The ri	ver is dangerous.'		
C.	ĩgũ guy	yari ārīmī		
	~igʉ	guya-di	~adi-~	bi
	3sg:m	be.dangerous-NOM:II	n be-3sc	:M.IMPERF
	'He is	in danger.'		

4.3.5.3 Nouns Derived from Particles

(43)

Some animate and inanimate nouns are derived from particles. This is illustrated in (43) below, with the particles coding deixis and anaphora. Although most of these nouns are used with a head they modify, i.e., a noun they refer to, they can also be used independently, that is, having a pronominal function (this is discussed further in section 4.4.1). The particle *iri* 'DEM:PROX' is also given as *i* by different speakers.

8	a. i/idi	idi-~ye	[ĩnẽẽ]	'this topic'
		DEM:PROX-topic		
		i-bu	[ibu]	'this basket'
		DEM:PROX-CLS:baske	et	
		idi-gʉ	[uigu]	'this tree'
		DEM:PROX-CLS:trunk		
		i-ro	[oni]	'this thing' (inanimate)
		DEM:PROX-INAN		
		i-di	[ini]	'these ones' (inanimate)
		DEM:PROX-PL:IN		
		idi-sibu	[irisibu]	'this time'
		DEM:PROX-CLS:time		
		idi-p u	[iripʉ]	'this other'
		DEM:PROX-CONTR		
		idi-ta	[irita]	'this' (emphatic)
		DEM:PROX-EMPH		

b.	si	si-pʉ	[sipʉ]	'that place'
		DEM:DIST-CONTR		
		si-go	[sigo]	'that woman'
		DEM:DIST-3SG:F		
		si-bu	[sibu]	'that basket'
		DEM:DIST-CLS:basket		
C.	~0	~ o- ta	[õta]	'here' (emphatic)
		DEIC:PROX-EMPH		
		~ 0 -ge	[õŋge]	'to/around here'
		DEIC:PROX-LOC		
		~ o -p u -ge	[õpüge]	'to/around here (not there)'
		DEIC:PROX-CONTR-LO	С	
		~ 0- ~b uu	[ðmũʰũ]	'from here'
		DEIC:PROX-CLS:origin		
		~ 0- a	[õã]	'these here' (animate)
		DEIC:PROX-PL:AN		
d.	~\$0	~ so- a	[sõấ́]	'those there' (animate)
		DEIC:DIST-PL:AN		
		~ so -ge	[sõŋgé]	'to/around there'
		DEIC:DIST-LOC		
e.	gai	gai- ~da	[gaʰirā]	'others' (animate)
		other-PL:AN		
		gai-gʉ	[gaʰigʉ]	'(an)other one' (masculine)

other-38G:M		
gai -go	[gaʰigó]	'(an)other one' (feminine)
other-3SG:F		
gai-du	[gaʰidʉ]	'(an)other day'
other-CLS:day		
gai-pu	[gaʰipʉ]	'(an)other'
other-CONTR		
gai- bu	[gaʰibu]	'(an)other basket'
other-CLS:basket		

4.3.6 Negative Nominals

Negative nominals are formed with $\neg de$ [nee] preceding the noun. In this position, it has the meaning 'none of X' or 'no X', as illustrated in (44a-c). The negative particle $\neg de$ can also occur before a verb, in this position, it can mean 'nothing', 'nobody', or 'never', as illustrated in (44 d-e). Verb negation is discussed in Chapter 5.

(44) a. mārīrē nēŏ mārā

~badi-de	~de- ~o	~bada
1pl:incl-ref	NEG-DEIC:PROX	have:exist

'There wasn't anything of this for us.' (lit. none of this existed for us)

b. mãrĩ mõrã nẽ bari

~badi	~bo-~da	~de	ba-di
1pl:incl	not.have-PL:AN.PERF	NEG	eat-NOM:IN
'We don't hav	ve food.'		

c. mãrĩ mõrã nẽ ũrĩ

	~badi	bo-	~d	la	~de	~ u di	
	1pl:in	CL not	ha	ave-PL:AN.PERF	NEG	pupunl	na.fruit
	'We de	on't have pu	pı	ınha fruit'			
d.	nẽ boh	kasõmẽ mã	ĩi	rire			
	~de	boka-~sob	e	~badi	idi-de		
	NEG	find-will.n	ot	1pl:incl	DEM:P	ROX-REI	- -
	'We w	ill never fin	d t	this.'			
e.	nẽ d u a	biriyũrã ĩg ũ	rẽ				
	~de	d u a-bidi-~	/u	-~da			~ig u -de
	NEG	keep-NEG-I	EV	id:quot/folk - f	L:AN.PI	ERF	3SG:M-REF

'(They) didn't leave anything for him.'

4.4 Pronouns and Other Proforms

In Desano it is possible to identify five proforms. They can be distinguished as personal pronouns, reflexive pronouns, reciprocal pronouns, demonstrative pronouns, and interrogative pronouns. They are discussed separately in the following subsections.

4.4.1 Personal Pronouns

In general, pronouns are dependent upon something else for their semantic value. In Desano, the personal pronouns are free morphemes. They distinguish three persons (first, second, and third) and two numbers (singular and plural). In the third person

singular, gender (masculine and feminine) is distinguished. The first person plural has an inclusive/exclusive distinction. The personal pronouns are shown in Table 4.7.

Some sentences illustrating the uses of the first person pronouns are given in (45) and (46). The sentence in (45a) was uttered by a male person; thus, the first person triggers third person singular agreement on the verb, marked by the suffix *-gu*. The same sentence said by a female is shown in (45b); the verb now triggers agreement with 'third person singular feminine, marked by the suffix *-go*. The first person plural contrasts inclusive (including the addressee) versus exclusive (not including the addressee), as illustrated in (46a) and (46b) respectively. In (46a), the speaker describes an event in which both the speaker and the hearers are involved (i.e. they all 'know'). In (46b), the speaker is telling about an event that includes him and others, but not the hearer.

Table 4.7 Desano personal pronouns

	SINGULAR		PLUR	AL
	MASC	FEM		
1st person	yuu <yu'u></yu'u>		~badi <mãrĩ></mãrĩ>	gua <gua></gua>
			inclusive	exclusive
2nd person	~b uu <	m̃ũ'ũ>	~bʉa <n< th=""><th>nũ'ã></th></n<>	nũ'ã>
3rd person	~igʉ	igo	~ida <ĩrā>	or <ērā>
	<īgũ>	<igo></igo>		

(45) a. yu'u wîrãũ ãrã

yuu ~wida-gu ~adi-a

1SG desano-3SG:M be-PERF

'I am Desano (male).'

b. yu'u wîrãgo ãrã

yuu ~wida-go ~adi-a

1SG desano-3SG:F be-PERF

'I am Desano (female).'

(46) a. mārī māhsīka

~badi ~basi-ka

1PL:INCL know-EVID:REAS

'We know.'

- b. gua erore
 - gua edo-de

1PL:EXCL there-REF

'We (are) there (we live there).'

The second person plural $m\tilde{u} \tilde{a} / bua/ can be analyzed as a composite formed by <math>m\tilde{u} \tilde{u} / bua/ 2nd$ person singular' and the marker for 'plural animate' -a.

4.4.1.1 Anaphoric Uses of Pronouns

The third person pronouns can have an anaphoric function. These pronouns can be analyzed as being derived from the 'demonstrative proximal' particle i + the class marker referring to an antecedent noun or to something identified in the context – like

saying 'this' or 'that' when pointing at an object, 'this is mine' (pointing at a knife): -gu(for masculine or general animate singular), -go (feminine singular), -da (animate plural) and -di/-do (inanimate).⁷⁹ The sentences in the following examples illustrate the anaphoric function. Example (47) is a sentence from a narrative in which the speaker is talking about a male character in a traditional tale, after introduced the character, the speaker refers to this character by using the anaphoric pronoun -igu 'he'.

(47) ĩgũ ehoyũmĩ

i-~gu eo-i-~yu-~bi DEM:PROX-3SG:M arrive-EVID:QUOT/FOLK-3SG:M.IMPERF 'He arrived.'

In another narrative, the speaker first introduces his daughter by her name (Guahó), but later in the discourse he uses the pronoun *igo* 'she' anaphorically to refer to this daughter previously mentioned in the beginning of the narrative. This is illustrated in (48).

```
(48) igo mõ'mē pihimõ
```

i-go ~bobe-pii-~bo DEM:PROX-3SG:F work-release-3SG:F

'She left work.'

In (49), the anaphoric pronoun $\tilde{t}r\tilde{a}$ 'they' refer to animate entities – a group of children previously mentioned in a traditional story. Thus, $\tilde{t}r\tilde{a}$ is used to refer to animate entities.

⁷⁹ The anaphoric particle root *i* may be derived historically from the form *ti* 'anaphoric' found in other Eastern Tukanoan language (cf. Ramirez 1997:320 for Tukano, and Stenzel 2004:161 for Wanano). But in these languages, *ti* is not a 'proximate demonstrative'.

(49) ĩrã ba badoakuriyũrã

i-~da ba ba-doa-kudi-~yu-~da
DEM:PROX-PL:AN food eat-sit-walk-EVID:QUOT/FOLK-PL:AN.PERF
'They sit and ate the food.'

Besides being used with noun-class suffixes for animate nouns to form personal pronouns, the anaphoric particle *i* is also used to form anaphoric pronouns that refer to inanimate nouns, as shown in (50).

(50) a. õrē i nīhkūmūhū

~o-de i ~diku-~buu
DEIC:PROX-REF DEM:PROX earth-CLS:origin
'here in this earth'
b. iyare
i-ya-de

DEM:PROX-CLS:creek-REF

'in this creek'

4.4.2 Reflexive and Reciprocal Expressions

Desano does not have reflexive pronouns. Reflexive expressions are coded by the reflexive interpreted noun *basu*, as illustrated in (51).

(51) yu'u bahsu yãã

y uu	basu	~ya-a
1sg	REFLEXIVE	see-PERF
'I see	myself.'	

In many cases, *basu* has an emphatic function, as shown in (52a-b)

(52) a. ērā basu gāmē kēāpanūrūkā

~eda bas	u	~gabe	~kea-pa-~dʉdʉ-ka
3PL:AN REF	LEXIVE	RECIPROCAL	attack-hit-spank-EVID:REAS
'They foug	ht and beat	themselves.'80	
~ 1 ~ 1 1	. ~1 ~~	~ ~	

b. mũ'ũ bahsuta yãke ãrĩyũmĩ pare

~buu basu-ta ~ya-ke ~adi-~yu-~bi

2SG REFLEXIVE-EMPH see-IMP say-EVID:QUOT/FOLK-3SG:M.IMPERF

"See for yourself," he said'.

The nominal $\sim gabe$ [$n\tilde{a}m\tilde{e}$] RECIPROCAL is used in constructions expressing

mutual actions; it codes co-reference with a co-occurring nominal. It has the meaning of 'each other' or 'one another' in English, as shown in the examples in (53).

(53) a. yuhunũ ẽrã gãmẽ bohkayurã gihkuri ye mẽrã

	yuu-~d u	~eda	~gabe	boka-~yu-~da
	one-CLS:day	3PL:A	N RECIPROCAL	meet-EVID:QUOT/FOLK-PL:AN.PERF
	gikudi	ye-~b	eda	
	tortoise	jaguar	-COM/INSTR	
	'One day, the	tortoise	e and the jagua	r met each other.'
b.	ĩrã gãmẽ mĩm	iĩãmã		
	~eda ~gabe	•	~bibi-a-~ba	
	3PL:AN RECIPH	ROCAL	kiss-perf-3pi	.:AN.IMPERF

⁸⁰ In many languages reciprocals are just a kind of reflexive construction; here that seems to be the case, though the reciprocal $\sim gabe$ also occurs.

'They kissed each other.'

c. ĩrã gãmẽ kẽãmã

~eda	~gabe	~kea-~ba
3PL:AN	RECIPROCAL	attack-3PL:AN.IMPERF
'They	fought each oth	ner.'

4.4.3 Demonstrative Pronouns

In Desano, the demostrative pronouns make a distinction between 'proximal' and 'distal', as illustrated by the examples in (54) and (55), respectively. The particle *iri* 'DEM:PROX' is also realized as *i* by different speakers.

(54)	i/idi	a.	idi-~ye	[ĩnẽẽ]	'this topic'
			DEM:PROX-topic		
		b.	i-bu	[ibu]	'this basket'
			DEM:PROX-CLS:basket	t	
		C.	idi-g u	[uigu]	'this tree'
			DEM:PROX-CLS:trunk		
		d.	i-do	[oni]	'this thing (inanimate)'
			DEM:PROX-IN		
		e.	i-ri	[ini]	'these ones (inanimate)'
			DEM:PROX-PL:IN		
		f.	idi-sibu	[irisibu]	'this time'
			DEM:PROX-time		

	g.	idi-pʉ	[iripʉ]	'this other'
		DEM:PROX-CONTR		
	h.	idi-ta	[irita]	'this (emphatic)'
		DEM:PROX-EMPH		
(55) si	a.	si-pʉ	[sipʉ]	'that other'
		DEM:DIST-CONTR		
	b.	si-go	[sigo]	'that woman'
		dem:dist-3sg:f		
	C.	si- bu	[sibu]	'that basket'

DEM:DIST-CLS:basket

In Desano, the demonstrative and the third person personal pronouns are not distinguished from one another. In Section 4.4.1.1, we saw that the third person personal pronouns $\tilde{i}g\tilde{u}$ 'third person singular masculine', igo 'third person singular feminine', and $\tilde{i}r\tilde{a}/\tilde{e}r\tilde{a}$ 'third person plural animate' are derived from the demonstrative proximal suffix i + the class marker referring to the antecedent noun: $-g\tilde{u}/-\sim gu/$ 'third person singular masculine', -go/-go/ 'third person singular feminine' and $-r\tilde{a}/-\sim da/$ 'plural animate'. Demonstrative pronouns and personal pronouns that overlap are found in many languages (cf. Schachter and Shopen 2007:27).

The fact that the forms in (54) can function as pronominal forms is illustrated in lines of examples (56) and (57). The inherent properties of the nouns for 'pigeon' (singular, animate) and 'girls' (plural, animate) in (56) are coded in (57) with the demonstrative *i* 'this', by the suffix *-gu* 'singular, masculine' referring to the pigeon, and by the demonstrative *i* 'this', and by the suffix *-~ra* 'plural, animate' referring to the girls. (56) yuhunũ buha yãpu nõmẽ nẽõ mãhsārā kurira

yuu-~dʉ bua **i**-∼da ~dobe ~ya-pʉ see-3SG.M.PERF DEM:PROX-PL:AN one-CLS:day pigeon female ~de-o ~basa-~da kudi-~da be.young-3sg:f people-pl:AN walk-pl:AN.PERF wa-~yu-~da go-EVID:QUOT/FOLK-PL:AN.PERF 'One day, the pigeon saw young girls going on a walk.' ĩgũpu buhapu yãpeyapu ẽrãrẽ ~igu-pu bua-pu ~ya-peya-pu 3SG:M-CONTR pigeon-CONTR see-be.on.top-3SG:M.PERF

i-~da-de

(57)

DEM:PROX-PL:AN-REF

'He observed them from above.'

The example in (58) illustrates the demonstrative forms for inanimate nouns. The specific class marker (classifier) *-gu* 'trunk', of the noun *yuhkugu* 'tree' in (58a), is coded on the first demonstrative form in line (58b).

(58) a. (...) ãrĩyũrõ yuhkugu

~adi-~yu-doyuku-gube-EVID:QUOT/FOLK-NON3.IMPERFtree-CLS:trunk'(by the shore of a lake) there was a tree.'

b. irigu beară inīră iroge doară werenīyūră

idi-gu bea-~da i-di-~da

DEM:PROX-CLS:trunk land-PL:AN.PERF do-PERF-PL:AN.PERF i-do-ge doa-~da wede-~adi-~yu-~da DEM:PROX-place-LOC sit-PL:AN.PERF tell-say-EVID:QUOT/FOLK-PL:AN.PERF '(They) landed on this tree, sit and talked (there).'

Another demonstrative is formed by the nominal *gahi* 'other' is used to refer to a new entity (or another of the same type), as illustrated in (59)-(61).

(59) gahinū ariyūmī daha te mūrārõta

(60)

(61)

gai-~dʉ	adi-yu-~bi		daa	te
other-CLS:day	come-EVID:QUOT/FOLK-3SG:N	M.IMPERF	again	till
~b u da-do-ta				
place-NOM:ABSTR-	-EMPH			
'Again, he came a	nother day, at the same place.	,		
gahig u arikumĩ da	ha			
gai-gʉ	adi-ku-~bi	daa		
other-3SG:M	come-ADVERS-3SG:M.IMPERF	again		
'Another one will come again.'				
gahibup u re ahpiab	du yu'u			

gai-bu-p u -de	api-a-bʉ	yʉʉ
other-CLS:basket-CONTR-REF	leave-PERF-NON3.PERF	1sg
'I put (it) in the other basket.'		

Pro-form		
gloss		
who	nðã	
what	yẽ 'ẽ	
where	nõ'õ	
which	di'i (also nĩ'ĩ)	

Table 4.8 Interrogative proforms in Desano

4.4.4 Interrogative Pro-Forms

Interrogative pro-forms are a set of nominal words that can occur as independent words. The five interrogative pro-forms that can be identified in Desano are listed in Table 4.8.

The pro-form $n\tilde{o}\tilde{a}$ 'who' can occur as independent root, as in (62), or it can appear with nominal morphology. In questions involving actions that imply direction, for example, the suffix *-pu* 'contrastive' is attached to the proform, as illustrated in (63) and (64).

(62) nõã ērārē siubiragu ārīkuri?

~doa	i-~da-de	siʉ-bida-gʉ	~adi-ku-di
who	DEM:PROX-PL:AN-REF	call-play-3SG:M	be-predic-inter
'Who will be the captain of our team?'			

.

(63) nõãpu sipumāsēge taribuhua mūtārõkuri? ārīyūrā ērā

~doa-pʉ	si-pʉ	~base-ge
who-contr	DEM:DIST-CONTR	shore-LOC

tadi-buha-~buta-do-ku-di ~adi-~yu-~da ~ida pass-across-go.ahead-PERF-PREDIC-INTER say-EVID:QUOT/FOLK-PL:AN 3PL:AN "Who will cross to the shore first?" They said.'

(64) nõãpū buriõmã?

~doa- pʉ	budi-~oba
who-CONTR	be.hard-run
'Who runs faster?	,,

The pro-form $y\tilde{e}'\tilde{e}$, when used as an independent word, codes a meaning similar to the pro-form 'what' in English, as illustrated in (65). However, the pro-form $y\tilde{e}'\tilde{e}$ can also occur with the word $n\tilde{o}h\tilde{o}$ 'uncertainty' suffixed to it, deriving a new proform expressing doubt, as illustrated in (66). In the narrative, the question in (65) is uttered when the speaker infers that nothing else is missing; whereas (66) infers that something is missing, but the speaker is not sure what is missing.

(65) yẽ'ẽ duyariro ã'rĩyũmĩ?

~yee	duya-di-do	~adi-~yu-~bi		
what	lack-NOM:IN-NON3.PERF	say-evid:quot/folk-3sg:m.imperf		
"What is missing?" he said.' (lit. what is missed)				

(66) yẽ'ẽnõhõ duyariro ărīyūmĩ?

~yee-~doo	duya-di-do		
what-uncertainty	lack-inter-non3.perf		
~adi-~yu-~bi			
say-EVID:QUOT/FOLK-3SG:M.IMPERF			
"What is missing?" he said."			

This 'uncertainty' meaning is also expressed in examples (65) and (66).

(67) në'ënôhô mãrĩrê igo o'ari?

~dee-~doo	Maria-~de	igo	oo-a-di	
what-uncertainty	Maria-REF	3sg:f	give-PERF-INTER	
'What did she give to Maria?'				

(68) në'ënôhôrë werenîari?

~dee-~doo-de	wede-~adi-a-di
what-uncertainty-REF	tell-say-PERF-INTER
'What did (they) talk abo	out?'

The pro-form $y\tilde{e}$ \tilde{e} can also derive other pro-forms with the demonstrative function to refer to a person (or people) whom the speaker does not know or cannot

identify.⁸¹ These forms are illustrated in (69)-(71).

(69) ĩ yẽ'ẽmũ ãrĩmĩ padere Peduru Baron

~i	~yeebu	~adi-~bi	pade-de	peduru baron
DEM:PROX:AN	so.and.so	be-3sg:m.imperf	priest-REF	Pedro Baron
'Was this one (so-and-so) Priest Pedro Baron?'				

(70) sĩ yẽ'ẽmũ oagu ĩgũ yãhãmĩ

~si	~yeebʉ	oagʉ	i-∼gʉ	
DEM:DIST	so-and-so	ethnic.group	DEIC:PROX-3SG:M	
~yaa-~bi				
join-3sg:M.IMPERF				

⁸¹ The Desano forms $y\tilde{e}$ ' $\tilde{e}m\tilde{u}$ and $y\tilde{e}$ ' $\tilde{e}m\tilde{a}r\tilde{a}$ have been translated into Portuguese as *fulano* 'so-and-so masculine' and *fulanos* 'so-and-so plural', respectively. For now, I do not know the appropriate gloss for the suffixes - $m\tilde{u}$ and $m\tilde{a}r\tilde{a}$ that is attached to $y\tilde{e}$ ' \tilde{e} .

'That indigenous person (so-and-so) joined (the mission).'

(71) sõã yẽ' ẽmãdã ãdãmã

~so-a	~ye'ebu- ~da	~adi-a-~ba		
DEIC:DIST-PL:AN	so-and-so-3PL:AN.PERF	be-perf-3pl:an.imperf		
'Those ones were (Tukano).'				

The pro-form di'i 'which' can also occur as an independent word as illustrated in (72). It can also take nominal morphology, in this case, usually the same classifier of the head noun, as illustrated in (73) and (74). In most of its occurrences, the pro-form di'i is realized as oral, however, instances where it is realized as nasal is also found as in (75), albeit not so common.

(72) di'i kurumāhārā ārīrī õā nāhsēāpu

	dii	kudu-~baa-~da	~adi-di	~ o-a
	which	knot-location-PL:AN	be-inter	DEIC:PROX-PL:AN
	~dase-a-p	u		
	Tukano-pl:AN-CONTR			
	'Which cl	an are these Tukano fi	rom?'	
(73) di'iru gahsiru eheayori?				
	dii-du	gasi-du	ee-a-y	vo-di
	which-CLS	s:round bark-CLS:rou	nd arrive	-PERF-EVID:HSAY-INTER
	'Which ca	anoe arrived?'		
(74)	di'ibu pui	bu m ũ'ũ ahpiari?		
	dii-bu	pui-bu	~b uu	api-a-di

which-CLS:concave basket-CLS:concave 2sg leave-PERF-INTER

'In which basket did you leave (it)?'

(75) nī'ī diaye mũ'rẽ kũrĩãrĩ

~dii	diaye	~b uu -de	~kudi-a-di
which	dog	2sg-ref	bite-PERF-INTER
'Which dog did bite you?'			

The pro-form $n\tilde{\sigma}$ ' $\tilde{\sigma}$ 'where' can occur independently, as in (76) or, more commonly, with nominal morphology indicating origin (77) or location (78).

(76) nõ'õ mũ'ũ koaduhkaru ahpiari

~doo	~bʉʉ	koa-duka-du	api-a-di
where	2sg	gourd-fruit-CLS:concave	leave-perf-inter
'Where di	d you p	ut the bowl?'	

(77) nõmühü äräri igü

~doo- ~b uu	~adi-a-di	~igʉ
where-CLS:origin	be-perf-inter	3SG:M
'Where is he from	?'	

(78) nõ'õge ãrãri mũ'ũã wãĩ

~doo -ge ∼adi-a-di	~b uu- ~a	~wai
where-LOC be-PERF-INTER	2sg-pl:An	uncle
Whereabouts are you (living), u	uncle?'	

4.5 Other Nominal Morphemes

In the discussion of the nominal morphology so far, I have presented a description of root types and nominalization of verbal roots, together with their core lexical and grammatical morphemes marking noun class (gender and classifiers) and number. In this section, I present other lexical and grammatical morphemes that can occur with a noun.

In the previous sections, we saw that a noun can be formed by an independent noun root (simple noun) or by an independent root plus a dependent noun root (complex noun), other nouns are derived from verbs roots and particles (derived nouns). Dependent noun roots are roots that cannot occur as independent roots; a complex noun is a noun formed by an independent root plus a dependent root (plus the nominal suffixes). Table 4.9 shows the order in which morphemes can be suffixed to the nominal root.

As shown in nominal morphology template in Table 4.9, there are eight morphemes that can occur (suffixed) with nominal roots. It should be noted that verb roots can be nominalized by the nominative suffixes -di <-ri> and <-ro>. Morphemes coding class and number were discussed in previous sections. These morphemes are represented in the first slot in the nominal template. In the following subsections, the nominal morphemes in slots 2-7 are described, followed by the description of the 'referential' -de <-re>, which can occur in any position after root/steam+position 1 morphemes.

154

Table 4.9	Nominal	morpho	logy
			`~ <i></i>

Nominal Root								
		1	2	3	4	5	6	7
Verb Root	NOM:CONCR	CLASS/NUM	AUG	CONTR	LOC	COM/INST	REF	ADD
	-di	-н/-дн М	-do	-p u	-ge	-~beda	-ta	-k u
	NON:ABSTR	-0/-go F	DIM			SOL		-~sa
	-do	-a/-~da pl:an	-~ga			-~pe		
Particle		-di pl:in						
		-~bu IND			OB.	TECT		
$Noun_{indep} + (Noun_{dep})$		CLS			-(de		

4.5.1 The Diminutive

The diminutive is expressed by the suffix -ga. In general it is used with both animate and inanimate nouns, as illustrated in (79).

(79)	a.	~bibi -~ga	<mīmīgā></mīmīgā>	'little hummingbird'
		hummingbird-DIM		
	b.	~kabi -~ga	<kāmīgā></kāmīgā>	'little brothers'
		brother-DIM		
	c.	gaki-~ ga	<gahkigā></gahkigā>	'little monkeys'
		monkey-DIM		
	d.	wii-~ga	<wi'igã></wi'igã>	'little house'
		house-DIM		

The diminutive suffix follows the plural markers (80a-b) and the class markers (80c-d).

(80)	a.	pee-~da-~ga	<perãgã></perãgã>	'little two (animate entities)'
		two-PL:AN-DIM		
	b.	wi'i-di-~ga	<wi'igā></wi'igā>	'little houses'
		house-INPL:AN-DIM		
	C.	~baa-g u -~ ga	<mãgũgã></mãgũgã>	'little son'
		progeny-3sg:M-DIM		
	d.	dip u -du-~ga	<dihpurugã></dihpurugã>	'little head'
		head-CLS:concave-DI	M	

The suffix -ga is also used to indicate even greater precision: 'closer than close', when referring to location, and 'more recent', when referring to time. These are

illustrated in (81a-d). This 'greater precision' meaning is also expressed when the suffix – $\sim ga$ is attached to adjectival expressions, as in (81e).

- (81) a. doka-~ga <dohkagā> '(right/exactly) under' under-DIM
 - b. podo-~ga <porogã> '(very) close'
 close-DIM
 - c. ~edo-~ga <eroga <eroga 'there' (distal)

DEIC:DIST-DIM

- d. ~kadu-~ga <kãrũgã> '(very) recent' recent-DIM
- e. ~abe-do-~ga <ãmērõgã> 'a little' be.little-NOM:ABSTR-DIM

The diminutive suffix is also used with pronouns (82a-b), with demonstratives (82c), and with numerals (82d). When used with pronouns (82a-b) and demonstratives (82c) the diminutive can also function for endearment. In (82d), with the numeral 'one', the diminutive gives the meaning 'by oneself'.

- (82) a. ~igu-~ga <ĩgũŋã> 'little him' 3SG:M-DIM
 - b. yʉu-~ga <yu'ugã> 'little me' 1sg-DIM
 - c. $\sim i c a$ $\langle iga \rangle$ 'little this'

DEM:PROX-DIM

d. yuu-go-~ga <yuhugogã> 'little her (by herself)'
 one-3SG:F-DIM

The diminutive suffix -aga is also used in some words that refer to time as in (83). In these words, the diminutive suffix seems to have lost its transparency as a grammatical morpheme and has become lexicalized.

(83)	a.	~gabi-~ ga	<gãmĩgã></gãmĩgã>	'tomorrow'
		night-DIM		
	b.	dopa-~ ga	<dopagã></dopagã>	'today'
		today-DIM		

4.5.2 The Augmentative

The augmentative is coded by the morpheme *-do*, as shown in the examples in (84), which generally occur to refer to non-feminine nouns.

- (84) a. americano-gu-do <āmērikānūguro> 'big American (male)'
 American-3sg:M-AUG
 - b. pea-~basa-gu-do <peamãhsãgũro> 'big white (male)'
 fire-person-3sg:M-AUG
 - c. ~ye-yu-gu-do <yēyũguro> 'big buriti palm tree'
 buriti-CLS:palm-CLS:trunk-AUG

Although the augmentative suffix is *-do*, most of the expressions indicating size are expressed with the verbal morpheme *wua-* 'be big' with animate and inanimate nouns, as illustrated in (85). It should be noticed that, in the case of inanimate nouns, the verb is nominalized (85c-d).

(85)	a.	∼ u bu v	wua-gi	ł	<ũmũ wuagu>	'big lad'
		lad t	be.big-	3sg:m		
	b.	~dobe-o)	wʉa- go	<nõmẽõ wʉago=""></nõmẽõ>	ʻbig woman'
		female-3	3sg:f	be.big-3SG:F		
	C.	wʉa- di-j	ye		<wearinge></wearinge>	'big fruit kernel
		be.big-N	IOM:IN	-CLS:seed		
	d.	wua-di-	wi		<iwinatiwi></iwinatiwi>	'big house'
		be.big-N	IOM:IN	-CLS:house		

The suffix -do <-ro> can co-occur with the nominalized verb *wua-* 'be big', in this case, the augmentative suffix functions as an intensifier, as illustrated in (86).

(86)	a.	wʉa-di-~yu-gʉ- do	<wʉariyũgʉro></wʉariyũgʉro>	'a very big palm tree'
		be.big-NOM:IN-CLS:p	alm.tree-CLS:trunk-A	UG
	b.	wʉa-di-wi-gʉ- do	<weariwigero></weariwigero>	'a very big house'

b. wua-di-wi-gu-do <wuariwiguro> 'a very big ho

 $be.big\text{-}\textsc{NOM:IN-CLS:}house\text{-}\operatorname{CLS:}trunk\text{-}\operatorname{AUG}$

4.5.3 The Locative: -ge

The locative morpheme *-ge* codes location, as in (87)-(89) and direction, as in (90)-(91). The locative marker *-ge* in Desano does not have cognates in other Eastern Tukanoan language. For example, the locative marker in Wanano is *-pu* (cf. Stenzel 2004:170); Tukano also has a suffix *-pu* which can mark location, although Ramirez

prefers to call it a 'focalizer' (cf. Ramirez 1997:218).⁸² It should be noticed that in these examples, the locative marker occurs with a noun and with nominalized verbs.

(87) ühtämügere wabita guya ãrīka

~uta-bu-ge-dewa-bidi-taguya-~adi-kastone-CLS:concave-LOC-REFgo-NEG-EMPHbe.dangerous-be-EVID:REAS'Do not go by the rapids, it's dangerous.'

(88) igo barisoraroge doamõ

igoba-di-soda-do-gedoa-~bo3SG:Feat-NOM:IN-cook-NOM:ABSTR-LOCsit-3SG:F'She is sitting in the kitchen.'

(89) pagu poroge wãgãyũmĩ

pa-gu

progenitor-3sg:M near-LOC wake.up-EVID:QUOT/FOLK-3sg:M.IMPERF

~waga-~yu-~bi

'(He) woke up near (his) father.'

- (90) aria yu'u abewi'ige
 - adi-a yuu abe-wii-**ge**

come-PERF 1SG sun-house-LOC

podo-ge

'I came from the house of the sun.'

- (91) i yebagere aribuku
 - i yeba-ge-de adi-bu-ku DEIC:PROX ground-LOC-REF come-NON3.PERF-ADVER

⁸² Tariana (Arawakan) has the locative marker *-se* (Aikhenvald 2003:148), with similar function of *-ge* in Desano (Aikhenvald 2002:103).

'(For me) to come to this land.'

The locative *-ge* can also occur on adverbial expressions, as in (92)-(93) and particles, as in (94).

(92) i ũmũge pũrĩ ãrĩyũ nẽgere

i~ubu-ge~pudi-~adi-~yu~de-ge-deDEIC:PROX universe-LOCbe.dirt-be-EVID:QUOT/FOLKnothing-LOC-REF'In the beginning, this universe was impure.'

(93) tātus ciglos purugere ariyūmī $\tilde{1}g\tilde{u}^{83}$

tātus ciglos pudu-ge-de adi-yu-~bi many centuries after-LOC-REF come-EVID:QUOT/FOLK-3SG:M.IMPERF

~igu

3sg:m

'After many centuries, he came.'

(94) erore erotamērā mārī ārību eroge õā paia buera sibuge

edo-de	edo-ta-~beda	~badi	~adi-bu
DEIC:DIST-REF	there-EMPH-COM/INST	1pl:incl	be-NON3.PERF
ero-ge	~ o-a	pai-a	bue-da
DEIC:DIST-LOC	DEIC:PROX-PL:AN	priest-PL:AN	teach-IMPERF
sibu- ge			
while-LOC			

'(We) lived there during the time in which these priests (were) teaching.'

⁸³ The expression 'tatus ciglos' is a borrowing from Spanish.'

4.5.4 The Contrastive: -pu

The contrastive morpheme *-pu* is used to mark 'contrast' between arguments in adjacent sentences. The Desano contrastive morpheme *-pu* yields a meaning similar to that described by Stenzel (2004:175) for the contrastive morpheme *-se'e* in Wanano: *but/however X, the other X,* or *X, though/on the other hand.* In (95) and (96), for example, the morpheme *-pu* marks not only the contrast between the two Subjects, but also a contrast in location, also shown in examples (97) and (98).

(95) mũ'ũã ipu mõ'mẽke

~b uu -a	i-pu	~bobe-ke
2sg-pl:an	DEIC:PROX-CONTR	work-IMP
'You work	(over) here.'	

(96) guapu wehkapu mõ'mērāka

gua-puweka-pu~bobe-~da-ka1PL.EXCL-CONTRabove-CONTRwork-EXRT-EVID:REAS'We work up there (upstream).'

(97) ĩgũ waharã i yebapusãrẽ baharã ãrikãyũmã

	~igu	waha-~da	i	yeba- pu -~sa-de
	3sg:m	enemy-PL:AN	DEIC:PROX	ground-CONTR-ADD-REF
	baa-~da	~adi-k	a-∼yu-~ba	
	many-PL:AN be-EXIST-EVID:QUC			/FOLK-3PL:AN.IMPERF
	'There were many enemies of him in this (other) land too.'			
(98)	g u a gahip	u waamũrã ãrĩt	oirika	

gua gai-pu wa-~budi ~adi-bidi-ka

1PL.EXCL other-CONTR go-climb be-NEG-EVID:REAS

'We will not move to another place.'

The contrast morpheme is frequently used with deictic expressions and with postpositions indicating location as in (99) and (100).

(99) deyorãsã sipu dohkapu

deyo-~da-~sa si-pu doka-pu hunt-EXRT-ADD DEIC:DIST-CONTR below-CONTR '(You) will also hunt down there.'

(100) weadigu wi'i purupu nîgîkumî

wea-di-gu wii pudu-**pu** clay-CLS:meat-CLS:tubular house after-CONTR ~digi-ku-~bi

stand.up-EVID:NVIS-3SG:M.IMPERF

'The statue is standing behind the house.'

The examples in (101) are excerpts from an explanation, by the speaker, that some Desano people went to live down the (Papuri) river; (101b) is a question from one of the speakers listening to the story, asking if this place down the river referred to the places previously mentioned in the narrative, hence the contrastive function of *-pu*.

(101) a. buaderă ărīmā pare

bua-de-~da~adi-~bapadego.down-PERF-PL:AN.PERFbe-3PL:ANIM.IMPERFthen'They went down (the river), then.'
b. iripure pare

idi- pu -de	pade
DEIC:PROX-CONTR-REF	then

'To these (other) places then?'

This contrastive function can also be seen in the examples in (102). It should be noticed that in (102a) the locative -ge, referring to a specific location, is used; whereas (102b) refers to a different location.

(102) a. wehkuyagere daha

weku-ya-ge-de daa

tapir-CLS:creek-LOC-REF again

'(He left our brothers-in-law) in the Tapir creek again.'

b. sipu mãsege

	ha aa aa
si-p u	~ba-se-ge

DEM:DIST-CONTR river-CLS:bank-LOC

'There, on the other side of the river.'

Other examples of the occurrence of the contrastive morpheme -pu are illustrated

in examples (103)-(104).

(103) sibupure koaru ahpiabu yu'u

si-bu- pu -de		koa-du
DEM:DIST-CLS:basket-CO	NTR-REF:OBJ	gourd-CLS:concave
api-a-bʉ	y uu	
leave-perf-non3.perf	1sg	

'I put the gourd in that (other) one.'

(104) yu'u peamähsä wäipure severianu wäikua

	y uu	pea-~basa		~wai- pu -de
	1sg	firewood-peop	ole	name-CONTR-REF
	severiano	~wai-kʉ-a		
	severiano	name-VBLZ-PE	RF	
	'For the w	hite people, I'n	n called	Severiano.'
(105)	ĩgũ abeya	mũrũkupure yẽ	ãyũmĩ	
	~igu	abe=ya	~buru-	k u-pu- de
	3sg:m	sun=poss	tobacc	o-CLS:tree-CONTR-REF:OBJ
	~yea-~yu-	~bi		
	hold-EVID:	QUOT/FOLK-3S	G:M.IMF	ERF
	'He took t	he tobacco plan	t of the	sun.'
(106)	õã nãhsẽãp	0 u		
	~о-а	~dase-	а-р и	

DEM-PL:AN tukano-PL:AN-CONTR

'These Tukano people.'

4.5.5 The Comitative/Instrumental Case

The case marker ~*beda* <-mērā> is used to code two different relationships between nouns. One is a comitative (accompaniment) relationship between two nouns, as illustrated in (107) and (108).

(107) gahi mũrũyuhkumẽrã ariyũmĩ

gai ~budu-yʉkʉ-~beda adi-~yu-~bi

 $another \ to bacco-tree-{\tt COM/INST} \ come-{\tt EVID:QUOT/FOLK-3SG:M.IMPERF}$

'He came with another tobacco plant.'

(108) îgũ yegumêrā keogārīgũ iyūmī

~igu yee-gu-~beda keo-~gadi-~gu 3SG:M jaguar-CLS:trunk-COM/INST measure-advance-3SG:M i-~yu-~bi do-EVID:QUOT/FOLK-3SG:M.IMPERF

'He came measuring with his sacred cane.'

4.5.6 Solitary: -~pe

The 'solitary' morpheme -pe < -pe' which, in my corpus is attached to particles, codes the meaning *only X*. I call it a 'solitary' suffix in order to remain consistent with the terminology used in the Tukanoan literature. Examples of the use of the solitary marker - -pe is shown in (109).⁸⁴

(109) a. ipētā yu'u māhsīā

	i-∼ pe- ta	y uu	~basi-a
	DEIC:PROX-SOL-REF	1sg	know-perf
	'I know only this.'		
b.	ipē ārīka		
	i-∼ pe	~adi-k	ca
	DEIC:PROX-SOL	be-EV	ID:REASON

⁸⁴. The morpheme -~*pe* has a meaning similar to the solitary morpheme *di'ta* in Wanano. However, in Wanano, the solitary morpheme occurs with other types of noun roots, not only with particles (cf. Stenzel 2004:177).

'This is all.'

c. ipēta ārīka yu'u werenīduaro

i-~ pe- ta	~adi-ka	y uu	wede-~adi-dua-do	
DEIC:PROX-SOL	be-EVID:REASON	1sg	tell-say-des-non3.IMPERF	
'I want to narrate this only.'				

4.5.7 Emphatic: -ta

The morpheme *-ta* is described in Miller (1999:163) as a 'limiter' glossed as 'exactly, precisely, just'. This morpheme can also occur in verbal constructions. It is possible to identify a cognate in other Eastern Tukanoan languages. In Tukano, for example, the same morpheme is described by Ramirez (1997: 222) as a 'specifier' having the same glosses as the ones presented by Miller for Desano. In Wanano, *-ta* is described as a 'referential' morpheme whose function is 'emphasize cross-reference to a previously-mentioned or focal noun;' thus, *-ta* has a meaning analogous to *that very X* or *that same X* (Stenzel 2004:173).

The examples in (110) illustrate the occurrence of the referential *-ta* in sentences excerpted from the same dialogue between two Desanos. In (110a), the morpheme *-ta* is used to emphasize an event that has been previously mentioned in the dialogue, something that could be literally translated as 'I tell you this same thing (again)', a construction commonly found in traditional narratives.⁸⁵ In (110b) the *-ta* indicates the exact location to which they had traveled. In that location, the speaker built a house, and

⁸⁵ A Desano consultant told me that one of the reasons that this construction is commonly used in narratives is 'because it helps to identify the important parts to be remembered.'

when asked if the house was his, the speaker replies with (110c), with the morpheme *-ta* emphasizing that the house was indeed his. Finally, (110d) is talking about a relative who is studying to be a priest and is in his last year in the missionary school; the referential *-ta* emphasizes that there is only a little time left for his relative to complete his studies and become a priest.

(110) a. mũ'ũre irireta ãrīrā yu'u

~b uu -de	idi-de- ta	~adi-da	yʉʉ
you-ref	DEIC:PROX-REF-EMPH	say-NON3.IMPERF	1sg
'I tell you this			

b. mūrīrabu tee oabuta

~bʉdi-da-bu	tee	oa-bu- ta		
climb-perf-non3.perf	until	skunk-CLS:concave-EMPH		
'We went to the Skunk rapids.'				

c. yu'uya wi'ita ãrīmūrību

yuu ya wii**-ta** ~adi-bu

1SG POSS house-EMPH be-NON3.PERF

'It was (used to be) my house.'

d. amērõta duyakaro

~abe-do-ta duya-ka-do

be.small-NOM:ABSTR-EMPH lack-EVID:REAS-NOM3.IMPERF

'Not taking long (to complete)'

The emphatic function of *-ta* can also be attached directly to a noun root, as in

(111) or to a nominal particle, as in (112).

(111) îgũ pĩrũta, ĩgũ gahsiru ĩgĩrũ ãrikãro iya

	~igu	~pid u-ta	gasi-~du	~igi-~du
	3sg:m	snake-EMPH	bark-CLS:concave	prow-CLS:concave
	~adi-~ka-	do		
	be-EVID:R	EASON-NON3.II	MPERF	
	'The snake itself was the prow of the canoe.'			
(112)	ĩta m ũ tãyĩ	ĩmĩ		
	~i -ta		~bʉta-~yu-~bi	
	DEM:PROX	:AN-EMPH	be.in.front.of-EVID:Q	JOT/FOLK-3SG:M.IMPERF
	'This (itse	elf) was in the f	ront.'	

4.5.8 The Additives: $-k\mu$ and $-\sim sa$

There are two 'additive' morphemes that occur with nominals. The first is *-ku*, which I gloss as 'and'. I call this morpheme 'CONJUNCTION'. It functions as a conjunction used to coordinate two or more nouns, as illustrated in (113) and (114).

(113) îgü nômêku pôrâku pârâmêrâku wagā ikū mū'urē ehatuabiriboka ārīyūmī

~igu ~	-dobe-ku	~po-~da	-k u	~pa-~da-~beda- k ʉ
3sg:m f	emale-CONJ	progenee-I	PL:AN - CONJ	grandchild-PL:AN-COM/INST-CONJ
wa-~ga	i-~kʉ	~b uu -de	ea-tua-bidi-l	bo-ka
go-DIM	do-adver	2sg-ref	arrive-lean-1	NEG-DUB-EVID:REAS
~adi-~y	u-~bi			
say-EVII):QUOT/FOLI	K -3 8G:M.IM	PERF	

"When he has a wife, children and grandchildren, it will be difficult for you to live," he said.'

(114) mãrī põrākū ārā o mārī pārāmērākurāge

~badi	~po-~da-kʉ	~adi-a	~0	~badi
1PL.INCL	progenee-PL:AN-CONJ	be-PERF	or	1PL:INCL
~pa-~da-~	-beda-ku-da-ge			
grandchild	1- PL:AN- COM/INST-CON	IJ-ADVER-LOC		

'We, who are parents or (some of us) who have grandchildren (...)'

The additive $-\sim sa$ codes a meaning similar to the English adverbs 'also/too', as in examples (115)-(117). I call this morpheme 'ADDITIVE'.

(115) pebu kuribu yu'usã

pe-b u	kudi-b u	y uu -∼sa		
hear-NON3.PERF	walk-non3.perf	1sg-add		
'I also walked (and) heard (this story).'				

(116) erota waïsa arimuridera arima

edo-ta	~wai-~sa	~adi-~budi-de-~da	~adi-~ba	
DEIC:DIST-EMPH	uncle-ADD	be-grow-PERF-PL:AN.PERF	be-3PL:AN.IMPERF	
'There, my uncles also lived.'				

(117) îpũsã ìgũ mũrũyẽmẽrã ăriyũmĩ

~i-p u-~s a	~igu ~budu-~ye-~beda
DEM:PROX:AN-CONTR-ADD	3SG:M tobacco-CLS:tangled-COM/INST
~adi-~yu-~bi	

be-EVID:QUOT/FOLK-3SG:M.IMPERF

"This other (person) was also with his cigar," he said."

4.5.9 Referential: -de

The morpheme -de <-re> in Desano has multiple functions. It is has been described by Miller as a 'specific object maker' which functions to mark "specific, referential patients of transitive verbs, experiencers, and spatial-temporal expressions (1999:57)." Ramirez (1997:224) claims that the basic functions of the morpheme -de, in Tukano, are syntactic and semantic. Syntactically, -de marks nonsubject arguments. Semantically, it marks the (most) *definite* or *referential* of the nonsubject arguments. I call -de a 'referential' marker, in order to account for its multiple functions: as 'object marker' (nonsubject argument), and as a marker of spatial-temporal expressions. Note on Table 4.9 that -de can occur after any of the morphemes in the slots 2-8. The object marker -de and the spatial-temporal morpheme -de are discussed in separately; however, I treat them as being the same morpheme.

4.5.9.1 The Object Marker: -de

Some nouns that function as objects of transitive and ditransitive verbs are marked by the suffix *-de*.

The Subject has no overt marking. In (118)-(120), -de marks the object of transitive verbs.

(118) îgũ yegure bohkayũmĩ

~igu	yee-gu-de	boka-~yu-~bi
3sg:m	jaguar-CLS:trunk-REF	find-EVID:QUOT/FOLK-3SG:M.IMPERF

'He found the sacred cane.'⁸⁶

(119) wãābu ārīyūmā īgūrē wererā

~wa-bʉ	~adi-~yu-~ba
be.good-PERF-NON3.PERF	say-evid:quot/folk-3pl:an.imperf
~ig u-de	wede-~da
3sg:m-ref	tell-3pl:AN.PERF

"It was good," they told him."

(120) ẽrã mãhsũduarã ĩyũrã ẽrãrẽ

~era ~basu-dua-~da i-~yu-~da

3PL:AN tame-want-PL:AN.PERF do-EVID:QUOT/FOLK-PL:AN.PERF

 $\sim eda-de$

3 PL:AN-REF

'They want to raise them.'

With ditransitives, only the indirect object is marked, as shown in (121).

(121) ĩgũ tĩgõrẽ wa'i o'gu imĩ

~igʉ	~tigo-de	wai	oo-g u	i-~bi
38G:M	sister-REF	fish	give-3SG:M	do-3sg:m.imperf

'He is giving fish to his sister.'

Deictic expressions also receive the referential marker -re when they are coded as

complement (object) of a verb (construction) coding motion, as in (119) and (120).

⁸⁶ The word *yee* 'jaguar/shaman' as a mythical entity can receive the classifier *-gu* 'CLS:trunk' to refer to another mythical entity, the sacred cane.

(119) ĩgũ erore yãmãhãnũgãyũmĩ

~igu	edo-de	~ya-~baa-~dʉga-~yu-~bi
3sg:m	DEIC:DIST-OBJ	see-stop-stand.up-EVID:QUOT/FOLK-3SG:M.IMPERF
'There, he	looked (around	d), stopped and stood up.'
m ũ 'ũãsã õ	gere mãhkãger	e māhārīke
~b uu -a-~s	a ~o-ge-	de
	~igʉ 3sg:M 'There, he mũ'ũãsã õ ~buu-a-~s	 ~igu edo-de 3sG:M DEIC:DIST-OBJ 'There, he looked (around mũ'ũãsã õgere mãhkãgere ~buu-a-~sa ~o-ge-

 $2 \texttt{SG-PL:AN-ADD} \quad \texttt{DEIC:PROX-LOC-REF}$

~baka-ge-**de** ~ba-adi-ke

settlement-LOC-OBJ stop-come-IMP

'You too, come here, to the settlement.'

The referential *-re* is also used to code complements as 'experiencers' of the idea expressed by the verb, as in (121) and (122).

(121) i yu'ure duyaro ãriyumi

iyuu-deduya-do~adi-~yu-~biDEIC:PROX 1SG-REFlack-NON3.PERFsay-EVID:QUOT/FOLK-3SG:M.IMPERF""To me, this is missing," he said.'

(122) mũ'ũ pagu mũ'ũrẽ doregu

~b uu	pa-g u	~b uu -de	dode-g u
2sg	progenitor-3SG:M	2sg-ref	command-3SG:M
_			

'Your father ordered you.'

4.5.9.2 The Spatial-Temporal: -de

The morpheme -de <-re> can also function as a marker of spatial-temporal expressions. In this function, -de generally occurs after the 'locative' marker -ge, as shown in (123)-(125). The expression *tātūs ciglos* in (125) is a borrowing from Spanish *tantos siglos* 'many centuries'.

(123) i ũmũge pũrĩ ãrĩro nẽgere

i	~ u bu-ge	~puri-~adi-do	~de-ge-de	
DEM:PRO	x universe-LOC	be.unclean-be-NON:ABSTR	nothing-LOC-REF	
'This universe was impure in the beginning.'				

⁽¹²⁴⁾ yu'u ye yãrõgere

y uu	ye	~ya-do-ge-de
1sg	shaman	see-NOM:ABSTR-LOC-REF

'My vision of/as shaman.'

```
(125) tātus ciglos purugere ariyūmī īgū
```

tãtus ciglos pʉdʉ-ge-de adi-yu-~bi many centuries after-LOC-REF come-EVID:QUOT/FOLK-3SG:M.IMPERF ~igʉ

3 SG:M

'After many centuries, he came.'

In (126) -*re* occurs after the 'additive' marker -~*sa* (slot 8 in the nominal morphology template); and in (127) it is suffixed directly onto the nominal particle. In these examples, -*re* also functions to mark a spatial reference. However, it should be noticed that in (128), despite occurring after the 'additive' suffix -~*sa*, the referential

marker *-re* functions as a marker of non-subject argument, which indicates that *-re* has a flexible order.

(126) îgũ waharã i yebapusãrẽ baharã ãrikãyũmã

waaa-~da	i	yeba-p u -~sa- de
enmity-PL:AN	DEIC:PROX	ground-CONTR-ADD-REF
~adi-k	a-~yu-~ba	
n be- EXI	ST-EVID:QUOT/I	FOLK-3PL:AN.IMPERF
	waaa-~da enmity-PL:AN ~adi-k N be-EXI	waaa-~da i enmity-PL:AN DEIC:PROX ~adi-ka-~yu-~ba N be-EXIST-EVID:QUOT/I

"There were many enemies of him in this land too," they said."

(127) õrẽ ĩgũ wirinũgã yãyũmĩ

~o-de ~igu widi-~duga-~ya-~yu-~bi

DEIC:PROX-REF 3SG:M leave-to.stand.up-see-EVID:QUOT/FOLK-3SG:M.IMPERF

'(Around) here he got up, went outside, and saw.'

(128) gõãmure ĩgu iriro dopăta ăria ĩgu săre

~goa-~b u-de	~igu	i-di-do	~dopa-ta	~adi-a
bone-IND-REF	3sg:m	do-perf-non3.imperf	COMP-EMPH	be-PERF
igu-~sa-de				
3sg:m-add-ri	ΞF			

'As it was done with God (Jesus), it was (done) with him too.'

4.6 Noun Phrase Structure

In this section, I present a description of the structure of noun phrases in Desano. The noun phrase in Desano is described following the typology proposed in Dryer (2007). It is possible to identify two types of noun phrases in Desano: (i) simple noun phrases, formed by pronouns or nouns plus simple modifiers (demonstratives, numerals, adjectives); (ii) complex noun phrases, formed by possessive or genitive modifiers and relative clauses.

In general, the order of constituients within a noun phrase is: modifier + head noun, which is generally the case in languages with a basic Object-Verb order (Givón 2001a:242; Song 2001).

4.6.1 Simple Noun Phrases and Their Parts

Simple noun phrases are considered to be common in languages, and they refer to noun phrases which are formed by either a single unmodified noun or a pronoun (cf. Dryer 2007). Desano does not have articles. The types of noun phrases illustrated in examples (129)-(131) are quite frequent in the language.

(129) bui duriamĩ

- [bui]_{NP} dudi-a-~bi agouti flee-PERF-3SG:M.IMPERF 'the/a agouti escaped.'
- (130) yiaye pîrûrê kuriamî

[yeaye]_{NP} [~pidu-de]_{NP} kudi-a-~bi

dog snake-REF bite-PERF-3SG:M.IMPERF

'The/a dog bit the/a snake.'

(131) ĩŋũ kãrĩmĩ

[~igu]_{NP} ~kadi-~bi

3SG:M sleep-3SG:M.IMPERF

'He sleeps.'

4.6.1.1 Demonstrative Modifiers

As nouns modifiers, demonstratives function to point out, or focus on the nouns they modify. In sections 4.3.5.3, 4.4.1, and 4.4.4, it was shown that some independent nouns, including pronominal forms, are derived from particles which have deictic and anaphoric functions. A demonstrative pronoun derived from one of these particles can occur by itself as a noun phrase, as in (132)-(134). In these examples, the demonstratives are used anaphorically to refer to a noun that has already been introduced. In (132), *îta* refers to a male person, previously mentioned in the text; whereas in (133), *iripēta* refers to the text itself. In (134), the demonstrative *iri* is overtly marked with the specific class marker *-gu* 'CLS:tree' functioning as anaphoric link to the referent noun *yuhkugu* 'tree', previously introduced in the narrative.

(132) îta mutăyumî

~i-ta	~buta-~yu-~bi	
DEM:PROX:AN-EMPH	be.ahead-EVID:QUOT/FOLK-3SG:M.IMPERF	
'This (one) was going ahead.'		

(133) iripēta

idi-~pe-ta

DEM:PROX:IN-SOL-EMPH

'This (is) all.'

(134) irigu beară inīră iroge doară wereniyūră

idi_m	hea_~da	i_di_~da
lui-ga	UCa- da	1-u1ua

DEM:PROX-CLS:trunk land-PL:AN.PERF do-perf-pl:an.perf ido-ge doa-~da wede-~adi-~yu-~da DEM:PROX-LOC sit-pl:AN.PERF tell-say-EVID:QUOT/FOLK-PL:AN.PERF '(They) landed on this tree, sit and talked.'

As modifiers, demonstratives co-occur with the head nouns, as shown in (135)-(137). Although in these examples the demonstratives are not overtly marked with the noun class features (gender/classifier) of the referent noun, they are still agree with the head noun in animacy.

(135) ĩpũsã ĩgũ mũrũyẽmẽrã

~i-pu-~sa ~igu ~budu-~ye-~beda DEM:PROX:AN-CONTR-ADD 3SG:M tobacco-CLS:tangled-COM/INSTR

'This other (one) too with his tobacco.'

(136) î gâmûmêrâ

~i ~gabu-~beda

brother-COM/INSTR DEM:PROX:AN

'(with) this brother...'

(137) î yebamāhsū i nihkūmūhū ārīyūmī īgū

~i	yeba-~bas u	i		~dik u- ~b uu
DEM:PROX:AN	ground-persor	DEM:PF	ROX:IN	earth-origin
~adi-~yu-~bi			~igu	
be-EVID:QUOT/FOI	LK-3SG:M.IMPE	RF	3sg:m	

'This Person of the Ground was from this planet.'

The demonstrative *gahi* 'other' can occur by itself in the noun phrase, as shown in (138), or it can co-occur with the head noun, as shown in (139). It should be noted that when *gahi* occurs by itself in the noun phrase, it is overtly marked with the class marker of its referent. In (168) the referent noun is a masculine animate noun, anaphorically marked by the 'third person singular masculine' class marker *-gu*. In (139), *gahi* does not have an overt marker.

(138) gahigu arikumī daha

gai-guadi-ku-~bidaaother-3SG:Mcome-ADVERS-3SG:M.IMPERFagain'Another one will come again.'

(139) gahi mãhkãge kurigãgũ ikumĩ

gai	~baka-ge	kudi-~ga-gʉ	i-ku-mĩ
other	settlement-LOC	walk-path-3sg:m	do-fut-3sg.m
'He will v	walk to the other com	munity.'	

4.6.1.2 Quantifier Modifiers

The quantifier expressions that can occur as modifiers in the noun phrase are number and quantity expressions. These quantifiers can also occur by themselves as the head of a noun phrase.

4.6.1.2.1 Number

These quantifiers generally preceed the nouns they modify. There are three simple numeral noun roots that can be identified in Desano, they are: *yuhu* 'one', *pe* 'two', and

ure 'three'. Numeral noun roots are overtly marked for the noun class feature of the head noun only if the head noun is animate, as shown in (140a-b), (141a-b) and (142a-b). If the head noun is inanimate, the numeral modifier is not overtly marked for the class marker feature of the head noun, as shown in (140c-d), (141c-d) and (142c-d).

(140) a. yuhugu mãhsũ

yuu-g u	~basʉ	
one-3SG:M	person	
'one (male) person'		

b. yuhugo nõmeõ

yuu-go	~dobe-o
one-3SG:F	female-38G:F

'one woman'

- c. yuhu gahsiru
 - yuu gasi-du one canoe-CLS:concave
- d. yuhu gãhpību
 - yuu ~gapi-bu
 - one sweet.potato-CLS:basket

'one basket of sweet potatoes'

(141) a. perã kārāyā

~karaya

two-PL:AN chicken

'two chickens'

b. perã nõmẽã

be-a

two-PL:AN female-PL:AN

'two women'

- c. pe wi'iri
 - pe wii-di
 - two house-PL:IN

'two houses'

- d. pe diuri
 - pe diu-di
 - two egg-PL:IN

'two eggs'

(142) a. urerã nômêã

- ude-~da ~dobe-a
- three-PL:AN female-PL:AN

'three woman'

b. urerã yea

ude-~da	ye-a
---------	------

three-PL:AN jaguar-PL:AN

'three jaguars'

- c. ure diuri
 - ude diu-di
 - three egg-PL:IN

'three eggs'

Besides co-occuring with the noun they modify in the noun phrase, the numerals *yuhu* 'one', *pe* 'two' and *ure* 'three' can also occur by themselves as the head of the noun phrase, as shown in examples (143)-(145). In these cases, they have an anaphoric function, and thus are overtly marked with the noun class feature of the noun they refer to.

(143) a. yuhugu

yuu-gʉ

 $one-3 \, \mathrm{SG.M}$

'one/a man'

b. yuhuyeta

yuu-~ye-ta

 $one\text{-}{\rm CLS}\text{:}tangled\text{-}{\rm EMPH}$

'one/an idea'

(144) a. perũ

pe-~da-u

two-PL:AN-M

'two males'

b. peru

pe-~du

two-CLS:day

'two days'

(145) a. urerũ

ure-~da-u

three-PL:AN-M

'three people'

b. urenũ

ude-~du

three-CLS:day

'three days'

The numbers 'four' and 'five' are derived from the complex structures show in (146) and (147), respectively.

(146) wahpiku wapi-ku pair-add 'four'
(147) yuhu mõhõtõ yuu ~boo-~to one hand-CLS:

hand-CLS:palm

'one hand'

As shown in Miller (1999:46), Desano employs a quintenary numeral system. As shown in (147), 'five' refers to one hand. Numbers higher than five are derived by the combination of a hand plus individual numbers, for example, 'six' is *yuhu mõhõtõku yuhuru nĩđrã* (lit. 'one hand plus one other finger'). However, these more complex forms are rarely used, as speakers generally borrow the Portuguese or Spanish numbers.

In Desano, the numeral *yuhu* may have other semantic functions besides

indicating quantity. One of the other functions seems to be for indefinite reference.⁸⁷ In these cases, the noun phrase introduces new participants in discourse in a construction of the type *one* X, as shown in (148).

(148) yu'u yuhugota ãrãbu

у ии	yuu-go-ta	~adi-a-b u
1sg	one-3SG:F-EMPH	be-perf-non3.perf
'I am or	ne female (Desano).'	

Another contruction codes the meanings 'by one self' or 'alone', as shown in the examples (149)-(150).

(149) karu tarisu puriamī yuhuguta

karu	tadi-su	pudi-a-~bi	yuu-g u -ta
Carlos	flute-CLS	blow-perf-3sg.m.imperf	one-3SG:M-EMPH
'Carlos played flute by himself.'			

(150) igo yuhugogā waapo

igo	yuu-go-~ga	wa-a-po
3sg:f	one-3sg:f-DIM	go-perf-3sg:f.perf

'She travelled alone.'

⁸⁷ I follow the analysis proposed by Stenzel (2004:191) for the numeral $\sim ku$ 'one' in Wanano, as this seems to be the same for *yuhu* 'one' in Desano.

Indefinite quantities are expressed by quantifiers coding the meanings 'many/a lot', 'a few' and 'all (of)'. The quantifier *baha* 'many/a lot' can occur either before or after the head noun, as illustrated in (151)-(152), respectively. It also should be noted in these examples that the quantifier word is overtly marked for number and animacy of its referent noun.

(151) baharā ārīkayūmā

baa-~da~adi-ka-~yu-~bamany-PL:ANbe-have-EVID:QUOT/FOLK-3PL:AN.IMPERF'there were many (people)'

(152) wa'i baharā

- wai baa-~da
- fish many-PL:AN

'many fish'

The quantifier word coding the meaning 'a few' is a complex structure derived from *baha* 'many', which takes the negative marker -~*be*, shown in (153) and (154). It should be noticed that in (153), the head of the noun phrase is the pronoun *gua* 'first person plural exclusive'; it precedes the modifier which is overtly marked for the features number and animacy of the referent noun. In (154), the quantifier refers to an inanimate noun, it is not overtly marked for noun class features, and it receives the diminutive morpheme -~*ga*, which, in this case, codes emphasis.

(153) gua bahamérá

1PL:EXCL many-NEG-PL:AN

'a few of us'

(154) bahamega

baa-~be-~ga

many-NEG-DIM

'a few (things)'

The quantifier coding the meaning 'all (of)' is also a complex structure derived from the verbal root *pere*- 'to finish/to end'. It should be noticed that in (155), *pere* codes the verbal meaning 'to end/to finish', but in (156), it forms complex construction with the verb $\tilde{a}r\tilde{i}$ 'be', plus a nominalizer *-ri* to code the quantifier modifier 'all of'.

(155) i ũmũ perebea

i	~ubu	pede-be-a
DEM:PROX:IN	universe	end-NEG-PERF
'This universe does not end.'		

(156) ãrīpereri semanārē

~adi-pede-di semanã-de

be-end-NOM:IN week-REF

'all (of) the week'

Independent quantifier expressions coding 'a lot of/a large amount of' and 'a small amount of/a little' of are constructed with the descriptive verb roots *wua-* 'be big' and $\tilde{a}m\tilde{e}$ - 'be small', which are nominalized, as shown in (157) and (158).

(157) dehko wuaro

deko wua-do

water be.big-NOM:ABSTR

'a large amount of water'

(157) dia ameroga yura nemoadero arabu

dia ~abe-do-~ga yuda-~debo-a-de-do river be.small-NOM:ABSTR-DIM raise-increase-PERF-PERF-NON3.IMPERF '(The water in) the river has raised a little.'

4.6.1.3 Adjectival Modifiers

In Section 3.4.1.3.1 of Chapter 3, it was claimed that Desano does not have an adjective class with grammatical characteristics that make it distinct from the classes of nouns and verbs. Instead, in Desano, adjectival expressions are derived from stative verbs. In this section, I describe the uses of adjectivals in their attributive function (modifying a noun within the noun phrase) and their predicative function (occurrying as a predicate), and use the term 'semantic adjective' to characterize them.⁸⁸

Examples (158)-(161) illustrate the attributive construction. Descriptive verbs used as semantic adjectives are nominalized in order to modify nouns, and they agree with the class marker feature (gender/classifier) of the head noun. For example, in (158) the specific class marker (classifier) -gu 'trunk' of the head noun is overtly marked in the nominalized modifier. In (159), there is gender agreement between the head noun and its modifier. However, in (160), there is overtly marked agreement between the modifier and the head noun. It should be noticed that in (158) and (159), the modifier follows the head

⁸⁸ I follow Dryer (2007:168) who describes the term 'semantic adjective' as an expression to to refer to words that are adjective-like on the base of their meaning, regardless of their grammatical properties.

noun, whereas in (159), the modifier precedes the noun. I hypothesize that when the modifier precedes the head noun, it is not overtly marked for the class marker feature of the head noun, as also shown in (161).

(158) yuhkugu amediguga ara

 $[[yuku-gu]_N$ $[\sim abe-di-gu-\sim ga]_{MOD}]_{NP}$ $\sim adi-a$ tree-CLS:trunkbe.small-NOM:IN-CLS:trunk-DIMbe-PERF'The tree is (very) small.''The tree is (very) small.'

(159) ērā dehko ārīmī ērā pagu wuagu

~eda deko ~adi-~bi [[~eda pa-gu]_N [wua-gu]_{MOD}]_{NP} 3AN:PL half be-3SG:M.IMPERF 3PL:AN genitor-3SG:M be.big-3SG:M 'Among them there was their big father.'

(160) erosā wuari wi'i āriyõ

[edo-~sa]	$[[wua-di]_{N.MOD}[wii]_N]_{NP}$	~adi-yo
DEIC:DIST-ADD	be.big-NOM:IN house	be-EVID:HSAY
'There also there	was a big house.'	

(161) wuari dihtaruge turoge arīyūro yuhkugu

[[wʉa-di] _{MOD}	[dita-~du-ge	tudo-ge] _N] _{NP}
be.big-NOM:IN	lake-CLS:concave-LOC	shore-LOC
~adi-~yu-do	y u ku-gu	

 $be\mbox{-}{\rm EVID:}\mbox{-}{\rm QUOT/FOLK-NON3.}\mbox{-}{\rm PERF} \quad tree\mbox{-}{\rm CLS:}\mbox{trunk}$

'On the shore of a big lake, there was a tree.'

The equivalent of a semantic adjective in a predicative function is expressed by a non-nominalized descriptive verb, as illustrated in examples (162a)-(163a). In (162b) and

(163b), the semantic adjectives are in the attributive function, and thus trigger agreement to the noun they modify. (162a) is used to refer to the temperature in general (e.g. 'it's hot today', '(the soup) is hot'); (162b), in the attributive function, the semantic adjective has been nominalized and is overtly marked with the specific class marker $-\sim du$. In (163a), as the descriptive verb receives the proper verbal morphology, it is the predicate of the sentence coding the 'adjectival' meaning, whereas in (163b) the semantic adjective is nominalized and shows gender agreement with the head of the noun phrase.

(162) a. ahsiārā

asi-~adi-a

be.hot-be-PERF

'(It) is hot.'

b. wārīnu ārā

 $[\sim wa\text{-}di\text{-}\sim du]_{\rm NP} \qquad \sim adi\text{-}a$

be.good-NOM:IN-CLS:day be-PERF

'(It's a) nice day.'

(163) a. ĩgũ turaduamĩ

[~igu] _{NP} [tu	da-dua-~bi] _{VP}
--------------------------	---------------------------

3SG:M be.strong-DES-3SG:M.IMPERF

'He wants to be strong.'

b. îgũ turagu ărīduamī

~igu tuda-gu ~adi-dua-~bi 3sg:M be.strong-3sg:M be-DES-3sg:M.IMPERF

'He wants to be strong.' (lit. 'He wants to be a strong one.')

The examples above show that, in Desano, semantic adjectives are grammatically verbs, having the same morphology (and syntax) of other verbs. Some modifiers are, in fact, relative clause contructions (see section 4.6.2.2).

Finally, an interesting typological characteristic of adjectives found in many indigenous language of South America, which is also present in Desano, is a set of these semantic adjectives which are structurally negative versions of a semantic adjective with the polar opposite meaning (cf. Campbell 2012).⁸⁹ Some examples in Desano are shown in (166)-(167).

(166) Mārī ũmābego ārīmõ

~bari	~ u ba-be-go	~adi-~bo
Maria	be.tall-NEG-3SG:F	be-3SG:F
'Maria is s	short.' (lit. 'Maria is n	ot tall.')

(167) i turabea

i tuda-be-a DEM:PROX be.strong-NEG-PERF 'This is soft.'

4.6.1.4 Locative Adverbial Modifiers

Locative adverbs expressed through locative demonstrative words (like 'here' and 'there' in English) can also function as modifiers of nouns in a noun phrase. This is

⁸⁹ Campbell (2012: 295) provides examples of negative polar oppositive negatives for the languages Tiriyo and Taranoan (Cariban); Nivaclé (Matacoan), and; Chiriguano (Tupían, Tupí-Guaranían branch).

shown in (168) and (169). It should be noticed that the locative demonstrative is overtly marked for for number and animacy agreeing with the head noun.

(168) sõä yēmārā ārāmā

~so-a ~yeba-~da ~adi-a-~ba
DEIC:DIST-PL:AN so.and.so-PL:AN be-PERF-3PL:AN.IMPERF
'Those there are (Tukano)'
(169) õã paia eha

[~o-a pai-a]_{NP} ea-a

DEIC:PROX-PL:AN priest-PL:AN arrive-PERF 'Here the priests arrived.'

4.6.1.5 Interrogative Modifiers

In section 4.4.5 four interrogative nominals in Desano were identified: $n\tilde{o}\tilde{a}$ 'who', y \tilde{e} ' \tilde{e} 'what', $n\tilde{o}$ ' \tilde{o} 'where' and di'i (also $n\tilde{i}$ ' \tilde{i}) 'which'. The pro-forms generally occur alone in the noun phrase, as discussed in section 4.4.5, and illustrated in (170)-(171), using the pro-form $n\tilde{o}\tilde{a}$ 'who'.

(170) nõãpũ buriõmã?

[~doa-pʉ] _{NP}	budi-~oba	
who-contr	be.hard-run	

'Who runs faster?' (lit. 'who other than me runs faster?'

(171) nõãpu sipumāsēge taribuhua mūtārõkuri?

[~doa-pʉ] _{NP}	si-pʉ	~base-ge
who-contr	DEIC:DIST-CONTR	shore-LOC

tadi-bua-~buta-do-ku-di

pass-across-go.ahead-PERF-ADVER-INTER

'Who will cross to the shore (first)?'

The pro-form $y\tilde{e}$ ' \tilde{e} 'what' also occurs alone (as the head) in a noun phrase, with a pronoun function. More commonly, the pro-form $y\tilde{e}$ ' \tilde{e} uses a construction suffixed by the morpheme $n\tilde{o}$ ' \tilde{o} 'where' to express 'doubt'.⁹⁰ Thus, (172) and (173) can be used as interrogatives. The difference between these two is that (172) would allow the answer to be "nothing (is missing)", whereas in (173), it implies that something is missing.

(172) yē'ē duyariro ārīyūmī?

[~yee]_NPduya-di-do~adi-~yu-~biwhatlack-INTER-NON3.PERFsay-EVID:QUOT/FOLK-3SG:M.IMPERF"What is missing?" he said.'

(173) yë'ënõõ duyariro ãrīyũmĩ?

[~yee-~doo]_{NP} duya-di-do ~adi-~yu-~bi

what-where lack-INTER-NON3.PERF say-EVID:QUOT/FOLK-3SG:M.IMPERF ""What is missing?" he said.'

Unlike $n\delta\tilde{a}$ 'who' and $y\tilde{e}$ ' \tilde{e} 'what', the interregotive pro-form *di'i* 'which' generally occurs as a modifier in the noun phrase. When it occurs as modifier of an animate noun, *di'i* does not receive any extra morphology agreeing showing agreement with the head noun, as shown in (174)-(175). However, *di'i* is overtly marked for the specific class (classifier) feature of an inanimate head noun, as shown in (176)-(177).

(174) di'i kurumāhārā ārīrī õā nāhsēāpu

⁹⁰ When *nõõ* is suffixed to *yẽ 'ẽ* it is pronounced [yẽ'ẽnõõ] and it indicates 'uncertainty'.

	[dii] _{MOD}	[kudu-~baa-~	da] _N] _{NP} ~ad	li-di	~ o-a
	which	knot-location	-PL:AN be-	INTER	DEIC:PROX-PL:AN
	~dase-a-p	u			
	Tukano-Pl	L:AN-CONTR			
	'Which cl	an are these oth	ner Tukano f	rom?'	
(175)	di'i diaye	m ũ 'rẽ kũrĩãrĩ			
	[[dii] _{MOD}	[diaye] _N] _{NP}	~b uu -de	~kudi-	a-di
	which	dog	2sg-ref	bite-PE	RF-INTER
	'Which do	og did bite you'),		
(176)	di'iru gah	siru eheayuri?			
	[[dii-du] _M	od [gasi-o	lu] _N] _{NP}	ea-a-~	yu-di

which-CLS:round bark-CLS:round arrive-PERF-EVID:QUOT/FOLK-INTER 'Which canoe has arrived?'

(177) di'ibu puibu mũ'ũ ahpiari?

[[dii-bu] _{MOD}	[pui-bu] _N] _{NP}	~b uu	api-a-di	
which-CLS:concave	basket-CLS:concave	2sg	leave-perf-inter	
'In which basket did you leave (it)?'				

Finally, the pro-form $n\delta'\delta'$ (where' can occur as an independent word when modifying an animate head noun, as in (178); or, more commonly, as a construction suffixed by nominal morphology indicating origin and location - in which case these constructions function as the head noun, as illustrated in (179) and (180).

(178) nõ'õ mũ'ũ koaduhkaru ahpiari

 $\label{eq:model} [[\sim doo]_{MOD} \qquad [\sim b u u]_N]_{NP} \qquad koa-duka-du$

	where	2sg	gourd-fruit-CI	.s:concave
	api-a-di			
	leave-PERF-INTER			
	'Where did you p	ut the bowl?		
(179)	nõ'om ü h ü ârârî îg	ţũ		
	[~doo-~buu] _{NP}	~ada-di	i-~gʉ	
	where-CLS:origin	be-INTER	DEM:PROX-3S	G.MASC
	'Where is he from	1?'		
(180)	nõ'õge ãrãri mũ'ũ	iã wãĩ		
	[~doo-ge] _{NP}	~ada-di	~b uu -a	~wai
	where-LOC	be-inter	2sg-pl:an	uncle

'Whereabouts are you (living), uncle?'

4.6.2 Complex Noun Phrases

According to Dryer (2007:151), complex noun phrases are considered those that contain genitive or possessive modifiers or relative clauses.

4.6.2.1 Genitive (or Possessive) Constructions

The term 'genitive' is used here generally to refer to the type of constructions in which a noun occurs with another noun phrase denoting possession. However, the semantic relation between the genitive noun phrase and the head noun is not limited to possession; hence the term 'genitive construction' is used (cf. Huddleston and Pullum 2009). 4.6.2.1.1 Genitive construction with nominal possessors

In Desano, genitive constructions with nominal possessors are not overtly marked. The possessor and possessed nouns are juxtaposed in Genitive-Noun order, as shown in (181)-(183).

(181) nũhpĩ põrã

~dupi ~po-~da mutum.bird progeny-PL:AN 'mutum bird's squab'

(182) gahki poadari

gaki poa-da-di

monkey hair-CLS:thread-PL:IN

'monkey's hair'

(183) mãhã poari gahsiro

~baa	poa-di	gasi-do	

macaw hair-PL:IN bark-CLS:body.part

'macaw's feather hairdress'

It should be noticed that when the genitive noun has an attributive function,

instead of possessive function, the structure is Noun-Genitive order, as shown in (184a);

or it can have the structure Noun-Class Marker, as in (184b).

(184) a. gere ũrĩnũ

gede ~udi-~du

season pupunha.fruit-CLS:day

'the season of pupunha season' (lit. 'the season is pupunha fruit's')

b. perunũ ãrã

pedu-~dʉ ~ada

caxiri.drink-CLS:day be

'day of caxiri drink' (lit. '(It) is caxiry day.)

4.6.2.1.2 Possessive pronouns

The possessive pronoun markers in Desano have the same form as the personal pronouns discussed in Section 4.4.1 above, repeated here for convenience in Table 4.10.

Table 4.10 Possessive pronouns

	SINGU	GULAR PLURAL		SINGULAR PLURAL		RAL
	MASC	FEM				
1st person	ун	'H	mãrĩ	дна		
			'inclusive'	'exclusive'		
2ND PERSON	mũ	Ϋ́ũ	mนิ	'ã		
3 RD PERSON	Ĩgũ	igo	ĩrã or ẽrã			

Possessives pronouns are independent words which occur before the head noun, as shown in (185). It should be noticed that these pronouns do not undergo any type of phonological change when used in possessive constructions.

(185) a. yu'u pagu

yuu pa-gu

1SG genitor-3SG:M

'my father'

b. mũ'ũ pagusãmãrã

~buu pa-gu-~saba-~da

2SG genitor-3SG:M-vagina-PL:AN

'your ancestry'

c. ĩgũ mãgũ

~igu ~ba-gu

3SG:M child-3SG:M

'his son'

- d. mãrĩ wi'iri
 - ~badi wii-di
 - 1PL:INCL house-PL:IN

'our houses'

e. ẽrã nõmẽã

 $\sim eda \sim dobe-a$

3PL female-PL:AN

'their wives' (lit. their women)

4.6.2.1.3 Alienable and Inalienable Possession

In Desano there is a contrast between alienable and inalienable possession. Kinship terms and body parts are inalienably possessed, that is, where the relationship between possessor and possessed is an inherent or permanent one, as shown in (186a-b). In alienable possession, the relationship is a conventional one, which can be terminated, shown in (187a-b). It should be noticed that, in Desano, the construction with alienable possession has the morpheme *ya* with the possessor noun in (187a-b). This occurs with all instances of alienable possession found in the data.

- (186) a. ĩgũ tĩgõ
 - ~igu ~ti-go 3SG:M sibling-3SG:F 'his sister'
 - b. igo põrã
 - igo po-~da 3sG:F progenee-PL:AN 'her children'

(187) a. mũ'ũ ya wi'i

~buu ya wii

2sg gen house

'your house'

- b. ĩgũ ya mũrũku
 - ~igu ya ~budu-ku
 - 3SG:M GEN tobacco-CLS:tree

'his tobacco plant'

The possessive marker ya is analyzed here as a clitic, as shown in (187) above and (188a-c) below.⁹¹ It can be the root morpheme in derivations such as the ones in (188), in which the class marker -gu 'third person singular masculine' and -go 'third person singular feminine' are attched to the root ya in (188a) and (188b) respectively.

(188) a. guaya kuru

gʉa=ya	kudu
2PL:EXCL=POSS	knot
'our group'	

b. nãhsẽã ya nĩhkũ

~dase-a=ya ~dikʉ Tukano-PL:AN=POSS land

'Tukano's land'

c. ẽrã ya bueri

~eda=ya bue-di

3PL=POSS study-NOM:IN

'their studies'

(189) a. î doto mârî yagu

~i doto ~badi	ya=gʉ
---------------	-------

DEM:PROX:AN doctor 1PL:INCL POSS=3SG:M

'this doctor of ours'

⁹¹ Stenzel (2004:197-198) argues that, in Wanano, the morpheme ya is a root.
b. ĩgũ yago

~igu ya=go

3SG:M POSS=3SG:F

'his wife'

The example in (190) shows contructions with both alienable and inalienable possession.

(190) ĩgũ nêhkũ ya wi'i

~igu ~yeku=ya wii grandfather=POSS 3sg:m house 'his grandfather's house'

4.6.2.1.4 The locative possession: -~baa <baha>

In Desano, the morpheme baha codes a relationship between beings and their place of origin, as shown in (191) and (192).

(191) iri kuru mähārā ārīrī

idi kudu ~baa-~da ~adi-di DEM:PROX knot belong.to-PL:AN.PERF be-PERF '(They) are from this group.'

(192) wīrā sāgabrie māhārā

~wida	são.gabriel	~baa-~da
Desano	São Gabriel	belong.to-PL:AN.PERF
Desenes from S	a Gabriel'	

'Desanos from São Gabriel

4.6.2.2 Relative Clauses Modifiers

In section 4.6.1.3, it was shown that adjectival expressions derived by descriptive and stative verbs, called semantic adjectives, are common modifiers in the noun phrase. Dryer (2007:169) claims that 'semantic adjectives' are not grammatically distinct from relative clauses.

In Desano, relative clauses are nominalized (cf. Kaye 1968, 1971:103-147) constructions as illustrated in the examples (193) and (194), which show the nominalized relative clauses as modifiers of the head noun.

(193) igo mãhĩrãrẽ boego ãrĩmõ

[[igo] _N	[~bai-~da-de	boe-go] _{RC MOD}] _{NP}	~adi-~bo
3sg:f	child-PL:AN-REF:OBJ	study-3SG:F	be-3sg:F
'She is the	one who teaches child	dren.' (= 'she is a teach	ner.')

(194) ĩgũ wa'i wehigu ãrĩmĩ

[[~igu] _N	[wai	wei-gu]] _{RC MOD}	~adi-~bi
3sg:m	fish	kill-3sg:m	be-3sg:M.IMPERF

'The one who kills fish.' (lit. 'He is a fisherman.')

4.7 Summary

In this chapter I have described the nominal morphology of Desano. I described the basic structure of nouns and the different types of nouns, including the ones derived from particles. It was shown in section 4.3 that nouns in Desano are divided into two main types: animate and inanimate. These nouns are subcategorized according to the noun-class markers they take. Animate nouns take general class markers (gender markers) and inanimate nouns take specific markers (classifiers). The general class markers were also described in this chapter, followed by a discussion of how they are used to derive new nouns from other nominal roots or verbal roots.

The chapter also presented a discussion of pronouns and other pro-forms (section 4.4). It was shown that it is possible to identify five pro-forms for the language: personal pronouns, demonstrative pronouns, and interrogative pronouns. Other nominal morphemes were described in section 4.5. Two lexical morphemes were identified: the diminutive morpheme -ga and the augmentative morpheme -guro. There are six grammatical morphemes that attach to nouns. These were discussed in the order in which they appear when suffixed to the root (or stem).

Finally, in section 4.6, it was shown that the structure of noun phrases can be divided in two groups according to the types of modifiers they take. One is the simple noun phrase, formed by a pronoun or noun plus simple modifier (demonstrative and numeral) and the other is complex noun phrase, formed by possessive (or genitive modifiers) or relative clauses.

CHAPTER 5

VERBAL MORPHOLOGY

5.1 Introduction

In this chapter, I describe the verb types and their morphological structure. In Section 5.2, the general structure of verb roots is characterized. Section 5.3 presents a description of the verb classes found in Desano, namely, stative, non-stative, and auxiliary verbs. Section 5.4 describes the types of serial verb constructions. In Section 5.5 the grammatical morphemes that are suffixed onto the verb root are described in the order they appear in the verb template, these include dependent verb roots, negative suffix, aspectual suffixes, markers of modality and evidentiality and agreement markers.

5.2 Structure of Verb Roots

The Verb roots in Desano have the following characteristics:⁹²

- iv. they are bimoraic: [góé] 'to return', [koré] 'to stay';
- v. they are lexically specified as nasal [+nasal] or oral [-nasal]: /~adi/ [ãri] 'to be', /adi/ [ari] 'to come'; and,

⁹² For discussions of tone, bimoraic root structures, and morphemes specified as nasal vs. oral, see Chapter 2 on Phonology.

vi. Verbs are lexically specified for tone (each root requires at least one high tone): /~adi/ [ấrī] 'to be', /~adí/ [ãrī] 'to say'. The great majority of verb roots in Desano is bimoraic and has CVCV shape. However, verb roots of the shape V are also found, for example, /i/ [ii] 'to do'; /u/ [uu] 'to seal'. Unlike nouns, verb roots are not marked for stress. Stress is assigned when additional morphology is suffixed to the root (see Section 2.3.2, Chapter 2).

5.2.1 Derived Verbs

In Desano, verbs can be derived from nominals in two ways. One is derivation with the verbalizer suffix -ku attached to a nominal root; and the other is derivation through noun incorporation.

5.2.1.1 Verbs Derived with the Verbalizer -k#

Some verbs are derived from nominals with the nominalizer suffix -ku, as shown in examples (1) and (2). In (1), the verb *wāiku* 'to name/be called' is derived from the noun *wāī* 'name'; after being verbalized, it receives the aspectual suffix -di, seen in (1). In (2), the verb *diku* 'to become fat' is derived from the noun *di* 'meat', and receives the Subject-verb agreement marker $-m\tilde{i}$ for third person masculine.

(1) ĩgũ yo'gu wãikudi ãrimĩ

~igu yogu wãī-ku-di ~adi-~bi
3SG:M stutter name-VBLZ-IMPERF be-3SG:M.IMPERF
'He was called stutterer.'

(2) ĩgũ dikumĩ

~igu di-ku-~bi

3SG:M meat-VBLZ-3SG:M.IMPERF

'He fattened (up).'

5.2.1.2 Noun Incorporation

In Desano, some verbs stems are derived by a process of noun-verb compounding generally know as noun incorporation (cf. Mithun 1999:44). Examples of noun incorporation are shown in (3) and (4). In (3), the verb stem *duhkawari* 'to split up' is derived from the noun root *duhka* 'fruit' and the verb root *wari* 'to split'. In (4), the verb stem *pagãrĩ* 'to crawl' is derived from the noun root *pa* 'belly' and the verb root *gãrĩ* 'to advance'.

(3) õã wãĩmũ'ũsã duhkawariyũmã

~o-a ~wai-~buu-~sa

 ${\tt DEIC:PROX-PL:AN} \quad uncle-2{\tt SG-ADD}$

duka-wadi-~yu-~ba

fruit-split-evid:quot/folk-3pl:an.imperf

'These nephews (of mine) also split up (in groups).'

(4) pagārīke

pa-~gadi-ke

belly-advance-IMP

'Crawl!' (imperative)

In general the type of noun incorporation found in Desano involves simple compounding of a noun plus a verb, as shown in the examples above. This type of noun incorporation is considering Type I, in Mithun's (1984) typology of noun incorporation.

Miller (1999:110-111) mentions other types of verbal derivations. She mentions, for example, the suffix *-ye* which can be suffixed to a noun root to derive verbs coding the meaning 'give'. Thus the noun root *wãi* 'name' derives the verb *wãiye* 'to name'. Miller also describes a derivational process in which the 'assertive perfect' aspect suffix *-a* is attached to noun roots to derive verbs coding 'making into', for example, *dehko* 'water' derives the verb *dehkoa* 'melt'. I have not found examples of these types of derivation in my data.

5.3 Verb Classes

The Desano verbs can be classified into two major classes: stative and nonstative. The verbs in these classes have similar inflectional morphology when occurring in finite clauses. The inflectional verb paradigm is shown in Table 5.1. This section describes the morphosyntactic characteristics of these two classes of verbs, as well as their semantic subcategories.

5.3.1 Stative Verbs

The subcategories of stative verbs in Desano are: the copula verb *ãrĩ* /~adi/ 'be/exist/live/stay'; the nonexistential verb *mãrĩ* /~badi/; the stative possessive *ohpa* /opa/; locative and postpositional verbs; and descriptive stative verbs (semantic adjectives).

Simple	1	2	3	4	5	6	7
Verb	MAN	ASP	MOD	NEG	ASPECT	CLAUSE	AGREEMENT
Root						MODALITY	MARKERS
(INDEP)		·	•		PERFECT	EVIDENTIAL	'third-
ROOT	DI	EPENDE	ENT	bi(di)	IMPERFECT		person'
	VI	ERB RO	TO			INTERROGATIVE	'non third-
					'participle'		person'
						'ORIENTED'	

Table 5.1 Template of verbal morphology

5.3.1.1 Copula Verb

The copula verb $\tilde{a}r\tilde{t}/\sim$ adi/ is used in constructions that code permanent or temporary states. The predicates in these clauses may function to describe the permanent states or attributes of the subject of the clause. The examples in (5) show that the copula verb is used to code an attribute or permanent state of the subject.

(5) a. ĩgũ colombiamũhũ ărīmĩ

~igu Colombia-~buu ~adi-~bi 3sg:M Colombia-CLS:origin be-3sg:M.IMPERF 'He is Colombian.'

b. yuhkugu wuadigu ara

yuku-gu wua-di-gu ~adi-a tree-CLS:trunk be.big-NOM-CLS:trunk be-PERF 'The tree is big.'

In Desano, copula verbs are also used in constructions with dummy-subject clauses, as shown in (6). These constructions generally describe events referring to natural conditions or weather phenomena (cf. Givón 2001:117).

(6) a. ahsiri ãrã

asi-di	~adi-a
be.hot-NOM	be-PERF
'(It) is hot.'	

b. wārīnu ārā

~wa-di-~du ~adi-a
be.good-NOM-CLS:day be-PERF
'(It) is a good day.'

Besides coding states and attributes, the copula verb *ãrĩ* is also used to indicate permanent or temporary location. In these cases, the copula is usually translated as 'live' or 'stayed' by the speakers.⁹³ The expression 'dos ãyõ mẽõ' in (7b) is a borrowing from Portuguese 'dois anos e meio' (two and a half years).

(7) a. ero ăriyoră eră pare

edo	~adi-~yo-~da	~eda	pade	
DEIC:DIST	be-EVID:HSAY-PL:AN.PERF	3pl	then	
'Then they lived there.'				

⁹³ Stenzel (2004:264) also reports that for Wanano, speakers often gloss the copula as 'to stay' instead of 'to be'.

b. arīkāmūrību yu'u o teresitare dos ayo meo

	~adi-ka-~bʉo	li-b u	унн	~0
	be-EXIST-HAI	3-NON3.PERI	F 1SG	DEIC;PROX
	teresita-de	dos ano me	eio	
	Teresita-REF	two and ha	lf years	
	'I was/stayed	(lived) in Te	erezita for t	wo and half years.'
Anoth	er function of t	he copula <i>ãr</i>	is to indic	cate existence, as shown in (8).
a.	ลิกังนิต่?			
	~adi-~yu-di			
	be-EVID:QUO	T/FOLK-INTI	ER	
	'There were (people there)?'	
b.	ĩg ũ waharã i y	vebap u sãrẽ b	aharā ārīkā	iyūmā
	~ig u waa-~	da i		yeba-p u -~sa-de
	3sg:M enmity	y-PL:AN DEN	I:PROX	land-contr-add-ref
	baa-~ra	~adi-ka-~y	′u-~ba	

many-PL:AN be-EXIST-EVID:QUOT/FOLK-3PL:AN.IMPERF

'There were many of his enemies in this land.'

5.3.1.2 The Nonexistence Verb

(8)

The stative verb *mãrĩ* /~badi/ is an inherently negative stative verb that indicates 'nonexistence', as shown in examples (9) and (10). Cognates of the verb mari/~badi/ are found in other Eastern Tukanoan language and with different glosses.⁹⁴

(9)	iri sibure	e mãrĩrĩmã	
	idi	sibu-de	~badi-di-~ba
	DEM:PRO	x time-REF	not.exist-IMPERF-3PL:AN.IMPERF
	(During)) this time, the	re were no (doctors); they (doctors) didn't exist.'
(10)	nẽ mãrĩy	vũmã nẽ mãrĩy	ũmã yuhudiayeta ãrīkāya
	~de	~badi-~yu-	~ba
	nothing	not.exist-EV	VID:QUOT/FOLK-3PL:AN.IMPERF
	~de	yuu-diaye-ta	a ⁹⁵ ~adi-ka-ya
	nothing	simply	be-EXIST-EVID:RES
	'There w	as nothing; sir	nply nothing existed.'

The stative verb *mari* can also occur with noun incorporation, indicating 'lack of

X/without X', as shown in (11) and (12); marī can also indicate 'a few', as in (13).

(11) ĩgũ surimãrĩ

~igu sudi-~badi

3sg:M clothing-not.exist

'He (is) naked'

(12) karu dimārīgū ārīmī

Carlos di-~badi-gu ~adi-~bi

Carlos meat-not.exist-3sg:M be-3sg:M.IMPERF

⁹⁴ The glossing 'not.exist' is used here following Stenzel's gloss for the same verb in Wanano (2004:264).

⁹⁵ At this point, I do not have a final analysis for the word *yuhudiayeta* glossed by my consultants as 'simply'. Thus, for now, I use his gloss.

'Carlos is skinny.'

(13)	д и а	mõhõmãríkʉ̃rã		
gʉ	a	~boo-~ badi -kʉ-~da		
1P	L:EXCL	hand-not.exist-FRUST-PL:AN.PERF		
'We (are) a few (people).'				

5.3.1.3 Stative Possession: ohpa 'HAVE'

The state of possession is coded by the transitive verb *ohpa* 'to have', as illustrated in the examples (14)-(16). It is used in constructions coding permanent states, as in (14), or in constructions coding temporary states, as in (15). The verb *ohpa* can also be used meaning 'to hold', as in the serial verb construction in (16).⁹⁶

(14) bugu puari ohparimī

bugu	pua-di	opa-di-~bi
anteater	claw-PL:IN	have-IMPERF-3sg:M.IMPERF
'The anteater has big claws.'		

(15) ĩgũ tebori ohpamĩ

(16)

~igu tebo-di opa-~bi
3SG:M be.lazy-NOM have-3SG:M.IMPERF
'He is lazy.' (lit. 'He has laziness.')
yẽã ãĩgãrĩ wi'ige ehomãhsũopayũmã

~yea-~ai-~gadi wii-ge

⁹⁶ According to Givón (2001:134), cross-linguistically, the verb of possession 'to have' is "historically derived from verbs such as 'take', 'grab', 'seize', 'hold' or 'obtain'."

comprise-take-advance house-LOC eo-~basu-**opa**-~yu-~ba feed-tame-have-EVID:QUOT/FOLK-3PL:AN.IMPERF '(They) caught them (the birds), took them home and raised them.'

5.3.1.4 Stative Verbs Indicating Location/Position

The verbs in this subgroup code (temporary or permanent) position or location of an entity, as illustrated in examples (17)-(24). The verb $m\tilde{g}\tilde{t}/\sim$ digi/ 'be standing' in (18) has also a semantically related sense of 'to stay', as in (19) and (20). However, the verb $m\tilde{e}r\tilde{e}/\sim$ bede/ is generally used to code the meaning 'to stay' (living permanently), as illustrated in examples (20) and (21). Another verb coding the meaning of 'to stay' is *duo*, as shown in (22) and (23). Finally, the verb *peya* 'to be on top of' occurs as a dependent verb, as in (24).

(17) îta mutăyumî

~i-ta	~bʉta- ~yu-~bi		
DEM:PROX:AN-EMPH	be.before- EVID:QUOT/FOLK-3SG:M.IMPERF		
'This (male person) went ahead of us.'			

(18) igo dihsiporo nĩgĩmõ

igo	disi-podo	~digi- ∼bo		
3sg:f	mouth-CLS:curve	be.standing/stay-3SG:F.IMPERF		
'She is standing by the door.'				

(19) eroge nĩgĩyũro irigu

edo-ge ~digi-~yu-do

idi-ku

DEM:PROX-CLS:tree

'This hill was standing/stayed/was located over there.'

(20) ũyũyã wi'igoroge ehamẽrẽyũmĩ

~uyu-~ya wii-godo-ge

avocado-CLS:creek house-clear.field-LOC

ea-~bede-~yu-~bi

arrive-stay-EVID:QUOT/FOLK-3SG:M.IMPERF

'He arrived and stayed in the community in the Avocado creek.'

(21) gua piamērēā waka pade

gʉa	pia-~ bede -a		wa-ka	pade		
1PL:EXCL	come-stay-PEF	RF	go-EVID:REAS	then		
'Then we	'Then we came and stayed (here).'					
mũ'ũã õ duhoke						
~b uu	~0	duo-ko	2			
2 PL	DEM:PROX	stay-IN	∕₽			

'You stay here.'

(22)

(23) duhoagu pare mãrī yēku ārībugu ariyumī pare

duo-a-gʉ	pade	~badi ~yekʉ	~adi-bʉ-gʉ
stay-PERF-3SG:M then	1pl	grandfather	be-NON3.PERF-3SG:M
adi-~yu-~bi		pade	
come-EVID:OUOT/FOLK	-38G:M.IN	APERF then	

'Then he came and stayed to be our grandfather.'

(24) ĩgũpu buhapu yãpeyapu ẽrãrẽ

~igu-pu	bua-p u	~ya -peya- p u	~eda-de
3sg:m-contr	pigeon-CONTR	see-be.on.top.of-3SG.M.PERF	3PL-REF:OBJ
'The pigeon obser	ved them from	above.'	

It should be noticed that the stative verbs $m\tilde{e}r\tilde{e}$ /~bede/ 'to stay/ in (20) and (21), and the verb *peya* 'to be on top of' in (24) occur as dependents in serial verb constructions.⁹⁷ Stative verbs coding location/position can also be nominalized, as shown in (25).

(25) ẽrã peyariberori mõmẽyũmã

~eda peya-di-bedo-di

3PL be.on.top.of-NOM:IN-circle-PL:IN

~bobe-~yu-~ba

work-EVID:QUOT/FOLK-3PL:AN.IMPERF

'They worked on the headdresses.'

5.3.1.5 'Adjectival' Stative Verbs (Semantic Adjectives)

As mentioned in Section 4.6.1.3 of Chapter 4, Desano does not have a lexical class of adjectives. It was argued that adjectival meanings are expressed by descriptive stative verbs through nominalization, as shown in (26)-(29), with the adjectival expressions in bold (see also example 25, above).

⁹⁷ See Section 5.4 for a description of serial verb constructions.

(26) yuhkugu wuarigu ara

yuku-guwua-di-gu~adi-atree-CLS:trunkbe.big-NOM:IN-CLS:trunkbe-PERF'The tree is big.'

(27) yuhkugu ameriguga ara

yuku-gu~abe-di-gu-~ga~adi-atree-CLS:trunkbe.short-NOM:IN-CLS:trunk-DIMbe-PERF'The tree is (very) small.'

(28) ērā dehko ārīmī ērā pagu wuagu

~edadeko~adi-~bi~edapa-guwua-gu3AN:PLhalfbe-3sg:M.IMPERF3PL:AN genitor-3sg:Mbe.big-3sg:M'Among them there was their big father.'

(29) ĩgũ yo'gu wãikudi ãrimĩ ĩgũ yo'gu bugu

~igʉ	yogʉ	wãĩ-kʉ-di	~adi-~bi	~igu	yogʉ
3sg:m	stutter	name-VBLZ-IMPERF	be-3sg:m.imperf	3sg:m	stutter
b u -g u					

be.old-3sg:m

'He, the stutterer, was called old stutterer.'

Descriptive stative verbs can also be inflected by regular verbal morphology, as shown in (30)-(32). In (31), the verb $w\tilde{a}$ /~wa/ is inflected with the evidential suffix *-ka* 'reason'. In (31), the verb ~*budi* 'to be hard' forms a serial verb construction with the active verb ~*ya* 'to see'. In (32), the verb *pobe* 'be fast' inflects for negation and evidentiality.

(30) mīnīsturu daporare pare! tſau, wāka m \tilde{u} ' \tilde{u}

minitrodapoda-depadetchau~wa-ka~buuministernow-REF:TEMP thenINTERJbe.good-EVID:REAS2sg'Now you are a (church) minister, then. Golly! You are good.'

(31) ĩgũ buriyãduarimĩ

~igu ~budi-~ya-dua-di-~bi

3SG:M be.hard-see-DES-IMPERF-3SG:M.IMPERF

'He wanted very much to see you.'

- (32) mũ'ũ pobebirika
 - ~buu pobe-bidi-ka

2SG **be.fast-NEG-EVID:REAS**

'You are not fast.'

5.3.2 Non-Stative Verbs

In this section the main semantic subclasses of non-stative verbs are described: *active verbs, verbs of motion,* and *verbs of perception and mental processes.* The lists of verbs presented do not pretend to be exhaustive; rather, they are intended to show the main characteristics of each of these subclasses.

5.3.2.1 Active Verbs

Active verbs describe events that are deliberately started by an active agent as the SUBJECT of the clause (cf. Givón 2001:106). Verbs describing actions may be intransitive

(with no object complement) as shown in (33); transitive (with a direct object

complement) as in (34); and ditransitive (with two object complements) as in (35).

INTRANSITIVE ACTIVE VERBS

(33) a. ĩgũ mõ'mēbirimĩ

[~igu] _{SUBJ}	~ bobe- bidi-~bi
3sg:m	work-neg-3sg:m.imperf
'He doesn't	work.'

b. peebu kuribu yu'usã

pe-b u	kudi-bʉ	[yʉʉ-~sa] _{SUBJ}
listen-NON3.PERF	walk-non3.perf	1sg-add

'I also listened and walked.'

c. diaye turiamĩ

dog bark-PERF-3SG:M.IMPERF

'The dog barked.'

d. ĩgũ ĩmĩpage gõrẽãmĩ

[~igu]_{SUBJ} ~ibi-pa-ge ~**gore-**a-~bi

3SG:M sand-CLS:fine.particles-LOC urinate-PERF-3SG:M.IMPERF

'He urinated on the sand.'

e. ērā doamā

[~eda] _{SUBJ}	doa- a-~ba
3PL	sit-perf-3pl:an.imperf

'They sit.'

TRANSITIVE ACTIVE VERBS

(34) a. karu pĩrũrẽ weheapʉ

	[karu] _{SUBJ}	[~pidu·	-de] _{DIR ОВЈ}	wee-a-p u
	Carlos	snake-I	REF	kill-perf-3sg:m.perf
	'Carlos killed	the snak	ke.'	
b.	yu'u ĩgũrẽ paa	ւ նս		
	[yʉʉ] _{SUBJ}	[~igʉ-c	le] _{DIR OBJ}	pa -a-bʉ
	1SG	3sg:m-	REF	hit-perf-non3.perf
	'I hit him.'			
C.	bua ẽrã nõmẽã	i põrãrẽ	s u yabirip u	
	[bua] _{SUBJ}	[~eda	~dobe-a	~po-~da-de] _{DIR OBJ}
	pigeon	3PL	female-PL:AN	progeny-PL:AN-REF
	suya- bidi-p u			
	like-NEG-3SG:	M.PERF		
	'The pigeon d	idn't lik	e the girls.'	
d.	ĩgũ yegure bol	hkayũmi	ĩ	
	[~igu] _{SUBJ}	[yegʉ-c	de] _{DIR OBJ}	boka- ~yu-~bi
	3sg:m	sacred.	cane-REF	find-evid:quot/folk-3sg:m.imperf
	'He found the	sacred o	cane.'	
e.	guaho dehko a	ahsirimẽ	rã soeamõ wah	arire
	[gua-o] _{SUBJ}	deko	asi-di-~beda	
	Guaho	water	be.hot-NOM-CO	DM/INSTR
	soe- a-~bo		[waadi	-de] _{DIR OBJ}

burn-perf-3sg:f.imperf Wahari-ref

'Guaho burned Wahari with the hot water.'

DITRANSITIVE ACTIVE VERBS

(35) a. ĩgũ guare wehkua ô'mũrĩrãmĩ

[~igu] _{SUBJ}	[gʉa-de] _{OBJ}	[wekʉ-a] _{OBJ}		
3sg:m	1pl:excl-ref	tapir-PL:AN		
oo-~budi-da-~bi				
give-hab-perf-3sg:m.imperf				
'He used to give us tapirs.'				

b. mãrī karure peru o'oamõ

[maria] _{SUBJ}	[carlos-de] _{OBJ}	[pedu] _{OBJ}			
Maria	Carlos-REF	caxiri.drink			
oo-a-~bo					
give-perf-3sg:f.IMperf					
'Maria gave caxiri drink to Carlos.'					

5.3.2.2 Verbs of Motion

This section presents a description of some of the motion verbs found in Desano. These are subdivided in fours subgroups: basic motion verbs, directional motion verbs, and relational motion verbs. Each of these is described in the following subsections.

5.3.2.2.1 Basic motion verbs

wa'a /waa/ 'go' - the verb wa'a codes a translocative motion (a motion away from the speaker), as shown in (36). It is generally used to code the meaning 'to leave', as in (36a), and 'to happen' as in (36d). The verb wa'a [wa'a] is usually reduced to [wa] in fast speech.

(36) a. bahko wa'abirika dapa

baku	waa -bidi-ka	dapa		
boat	go-NEG-EVID:REAS	still/yet		
'The boat hasn't left yet.'				

b. igo yuhugogã wa'apo

igo	yuu-go-~ga	waa-po
3sg:f	one-3SG:F-DIM	go-3sg:f.perf
'She w	vent by herself.'	

c. tee mãrige kauchero wa'arã

tee ~badi-ge kausero waa-~da

until 2PL:INCL-LOC rubber.tapper go-PL:AN.PERF

'The rubber tappers went all the way to (find) us.'

- d. õpã wa'aburo wãĩ
 - ~opa **waa-bu-**do ~wai
 - thus go-PERF-NON.3 uncle

'(It) happened like that, uncle.'

ari /adi/ 'come' - the verb *ari* codes a cislocative motion (motion towards the speaker), as shown in (37).

(37) a. ariya yu'u abewige

	adi- ya		yuu	abe-wi-ge	
	come-]	EVID:RES	1sg	sun-CLS:house	-LOC
	'I com	e from the hous	se of the Sun'.		
b.	ĩgũ ari	yũmĩ pare			
	~igʉ	adi- ~yu-~bi			pade
	3sg:m	come-EVID:QU	jot/folk -3 sg:1	M.IMPERF	then
	'Then	he came.'			
c.	mũ'ũ a	inin			
	~b uu	adi-di			
	2sg	come-INTER			

'Are you coming?'

Examples of other verbs of basic motion that occur often in the texts are given verbs are given in (38). They are *uba* 'to run' (38a), *kudi* 'to walk' (38b), *nũrũ* 'to chase/to follow' (38c) and *wu* 'to fly' (38d). The verb *wu* 'to fly' is generally used to describe the action of flying in general. However, another verb *peya* 'to fly', which does not occur as often as *wu*, can be used only with birds, as in (38e).

(38) a. wagãĩnĩgã ũmãyõrã
wa-~gai-~di-~ga ~uba-~yo-~da
go-scream-DIR:away-MOV:advance run-EVID:HSAY-3PL:AN.PERF
'(They) ran screaming.'
b. mũrã mãmãrõ gere ũrĩnũrẽ kurisʉayũmã

~bu-~da ~baba-do gede ~udi-~du-re

elder-PL:AN be.new-NOM season pupunha.fruit-CLS:day-REF

kudi-sʉa-~yu-~ba

walk-like-evid:quot/folk-3pl:an.imperf

'The elders liked to walk (roam around) during the pupunha season.'

c. ĩgũ nũrũsiayũmĩ ĩgũ yebamãhsũ

~igu ~**dudu**-sia-~yu-~bi

3SG:M chase/follow-go.behind-EVID:QUOT/FOLK-3SG:M.IMPERF

~igu yeba-~basu

3SG:M ground-person

'He followed the Person of the Ground.'

d. yuhunu yahiyéhkôméra mîmî wubirara iyôra

yuu-~dʉ	yai-~yeko-~beda	~bibi
one-CLS:day	heron-grandmother-COM/INSTR	hummingbird
wu- bida-~da	i-~yo-~da	

fly-play-3pl:an.perf do-evid:hsay-3pl:an.perf

'One day, the tuiuiu bird and the hummingbird were playing flying.'98

e. daharo payogãrīpu

daa-do	payo-~gadi-pu
be.slow-nom.abstr	fly-advance-3SG:M.PERF
'He (the tuiuiú) flew slov	vly.'

⁹⁸ 'Tuiuiú' is the local name for the *Jabiru mycteria*.

5.3.2.2.2 Directional motion verbs

A subgroup of motion verbs are the ones that code directions (or spatial orientations), for example, *baha* 'go uphill/upriver'; *bua* 'to go downhill/downriver'; and *yuri* 'to go down'. Cognates of these directional verbs are found in other Eastern Tukanoan languages. As pointed out by Stenzel (2004:284), the Eastern Tukanoan groups travel extensively by the rivers, and the identification of their communities is related to their position on the river and relation to other communities, which make these verbs extremely useful. Most of the traditional Desano communities are located in the headwater of the main streams in the Vaupés, Papurí, and Tiquié rivers, and travelling on foot is also common; thus, these verbs can be used to refer to motion done on land or on the river. The examples in (39) and (40) show the distinction between 'go uphill/upriver' and 'go downhill/downriver'. In (41), the verb *yuri* 'to go down' has a more generic sense. Although it can be used to in to refer to 'go down uphill/downhill' as shown in (41a); it can also be used in constructions to indicate a movement from a higher place to a lower place, as in (41b).

(39) a. õrē māhānūgāmūrību pare

~o-de	~baa-~duga-~budi-bu	pade
DEIC:PROX-REF	go.uphill/upriver-INCEP-HAB-NON3.PERF	then
'Then we moved	up here.'	

b. yee mãhãehayũpu

ye **~baa-**ea-~yu-pʉ shaman **go.uphill/upriver**-arrive-EVID:QUOT/FOLK-3SG:M.PERF 'The shaman arrived.' (lit. 'The shaman went upriver). (40) a. buanügāderā ārīmā ērā

(41)

	bua- ∼d u ga	a-de-~da				~adi-~ba	~eda
	go.downri	iver-ince	P-PARTIC-	3PL:AN	PERF	be-3pl:an.imperf	3PL
	'They beg	an to go c	lown (to a	nother	village)		
b.	buaderã ãr	ĩmã					
	bua -de-~d	a			~adi-~	ba	
	go.downri	iver-PART	TIC-3PL:AN	N.PERF	be-3PL	:AN.IMPERF	
	'They wen	it down (t	o another	village).'		
a.	ero ẽrã yui	bīnā āririr	eră ărīmã	ãrĩkʉyã	ã		
	edo	~eda	yudi-di-	~da			
	DEIC:DIST	3PL	go.down	-DIR:to	wards-3	PL:AN.PERF	
	~adi-de-~d	la		~adi-~	ba		
	be-PARTIC	-3PL:AN.P	ERF	be-3PL	:AN.IMF	PERF	
	'They went down (moved) and used to live there.'						

b. ĩgũ mãgũgã ĩgũrẽ ẽrã ya wi'i wahpikuturi ãrĩrõge yuridihia

~igu ~ba-gu-~ga	~igu-de	~eda	ya	
3SG:M progeny-3SG:M-DIM	3sg:m-ref	3PL	POSS	
wii-wapi-k u -tudi	~adi-do-ge	yud	i-di-a	
house-double-ADD-CLS:stacked be-NOM-LOC go.down-DIR:toward-PERF				
'His little son fell off the fourth floor of their house.'				

5.2.2.2.3 Relational motion verbs

This subset of motion verbs indicates relation in respect to a reference point which has been contextually established in discourse. Some of the relational verbs commonly found in Desano are: *pia* 'to move out/into', *wia* 'to move forward' (using the hands), and *wiri* 'to move outward', as shown in (42)-(44).

(42) a. ãĩômãã piari ariribu

~ai-~oba-a	pia -adi-di-bʉ
take-wrap-PERF	MOV.out.into-come-DIREC:towards-NON3.PERF
'He put (the dead mor	nkey) on his back and left (the forest) coming.

b. eroge ãrīmāhsībiri gua piamērēā waaka

gua	pia-~bede-a	waa-ka
1pl.excl	MOV.out.into-stay-PERF	go-EVID:REAS
'We moved fr	om there (to here).'	

(43) a. ero wiamühüyümī

edo **wia-**~buu-~yu-~bi

DEIC:DIST MOV.forward-lift-evid:quot/folk-3sg:m.imperf

'There he handed it (the tobacco plant) in.'

b. ĩgũ pagure wianũmĩ

~igu pa-gu-de wia-~yu-~bi
3sg:M genitor-3sg:M-REF MOV.forward-EVID:QUOT/FOLK-3sg:M.IMPERF
'He gave it (the tobacco) to his father.'

(44) a. õrê ĩgũ wirinũgãyãyũmĩ

~o-de ~igu

DEIC:PROX-REF 3SG:M

widi-~duga-~ya-~yu-~bi

MOV.outward-INCEP-see-EVID:QUOT/FOLK-3SG:M.IMPERF

'He came outside (of the longhouse) and looked around.'

b. ĩgũpu mãsãtĩgũ wiriduagu iriyũmĩ

~igu-pu ~basa-~tigu widi-dua-gu
3SG:M-CONTR people-sibling-3SG:M MOV.outward-DES-3SG:M.IMPERF
i-di-~yu-~bi
do-DIR:towards-EVID:QUOT/FOLK-3SG:M.IMPERF
'He wanted to be born as the people's brother.' ('He wanted to be
everyone's brother').

5.3.2.3 Verbs of Placement

The only verb of placement occurring as an independent verb (i.e., not suffixed to another verb root) found in my data corpus is *api* 'to leave', used with the semantics of 'placement' as shown in (45a-b). Other verbs of placement such as $m\tilde{e}r\tilde{e}$ 'to fall on top', and *tu* 'to put' are dependent verbs and occur in serial verb constructions, discussed in Section 5.4.

(45) a. iyêrê yu'u mũ'ãrê ahpiduaka
i-~ye-de yuu ~buu-a-de api-dua-ka
DEM:PROX-CLS:topic-REF 1SG 2SG-PL:AN-REF leave-DES-EVID:REAS
'I want to leave this subject to you.'

b.	wehkuyage ahpidi	ārīmī n ū gūdehkoge	
	wek u -ya-ge	api-di	~adi-~bi
	tapir-CLS:creek-LOC	e leave-IMPERF	be-3sg:m.imperf
	~du-ku-deko-ge		
	forest-CLS:tree-half	-LOC	

'He left them in Tapir Creek in the middle of the forest.'

5.3.2.4 Verbs of Perception and Mental Processes

These subclasses of nonstative verbs are described here when they occur as independent roots. They can also occur in dependent root position coding modality. However, I limit the discussion here to their main functions as independent verb roots.

5.3.2.4.1 Verbs of perception

The verbs of perceptions are $y\tilde{a}$ 'to see', pe'pi 'to feel' and pe 'to hear'. The verb $y\tilde{a}$ can be transitive, as in (46a), or intransitive, as in (46b). The verb pe'pi 'to feel' in (47a) can also be used to code the meaning 'to think' (a mental process), as in (47b). The verb pe 'to hear' used as a transitive verb in (48a), can also be intransitive, coding the meaning 'to listen' as in (48b).

(46) a. yāmēdihumī yu'ure

~ya-~be-diu-~bi yuu-de
see-throw-MOV.downwards-3SG:M.IMPERF 1SG-REF
'He looked at me.'

b. yãgũ iyũmĩ

~ya-gui-~yu-~bisee-3SG:M.IMPERFdo-EVID:QUOT/FOLK-3SG:M.IMPERF'He was looking.'

(47) a. wãrõ pe'piaka

~wa-do pepi-a-ka

be.good-NON3.IMPERF feel-PERF-EVID:REAS

'(Thus) I felt good.'

- b. yu'u pe'piabu
 - yuu pepi-a-bu
 - 1SG feel-PERF-NON3.PERF

'I thought.'

(48) a. ipēta yu'u pekarabu

i-~pe-ta	yu'u	pe-ka-da-bʉ
DEM:PROX	1sg	hear-EXIST-FRUST-NON3.PERF
'I heard only t	his.'	

b. peebu kuribu yu'usã

pe-bu	kudi-b u	y uu -∼sa
hear-NON3.PERF	walk-NON3.PERF	1sg-add
'I also listened and v	valked.'	

5.3.2.4.2 Verbs of mental processes

The verbs denoting mental process which occur currently in Desano texts are $m\tilde{a}s\tilde{i}$ 'to know', $k\tilde{e}$ 'to dream' and *beye* 'to explain'. These verbs can be transitive as in (49a, 50a, 51a), or intransitive as in (49b, 50b, 51b).

(49) a. yu'u ipēta māhsīka

y u 'u	i-~pe-ta	~ basi- ka				
1sg	DEM:PROX-SOL-EMPH	know-EVID:REAS				
'I know only this.'						

b. mähsidera arima era

~basi- de-∼da	~adi-~ba	~eda
know-PARTIC-PL:AN.PERF	be-3pl:AN.IMPERF	3pl
'They have known.'		

(50) a. igo yagumērā kēāpo

igo	ya-g u -~beda	~ ke- a-po					
3sg:f	poss-3sg:m-com/instr	dream-perf-3sg:f.perf					
'She dreamt about her man.'							

b. igo kego imo

igo ~ke-go i-~bo 3sg:F dream-3sg.F.IMPERF do-3sg:F.IMPERF 'She is dreaming.'

(51) a. irire beyeweadoaderā ārīmā

idi-de	beye- wea-doa-de-~da
DEM:PROX-REF	explain-extract-sit-PARTIC-3PL.PERF

~adi-~ba

be-3PL:AN.IMPERF

'They sat explaining these things.'

b. beyeweagārīgū iyūmī

beye-wea-~gadi-gui-~yu-~biexplain-extract-advance-3SG:Mdo-EVID:QUOT/FOLK-3SG:M.IMPERF'He was explaining and (continuing) analyzing again.'

5.3.4 Auxiliary Verbs

There are three auxiliary verbs in Desano: i 'do', wa'a 'go' and $\tilde{a}r\tilde{i}$ 'be'. Each of these auxiliary verbs has distinctive functions, as described in the following subsections.

5.3.4.1 The Auxiliary Verb i 'do'

The verb i 'do' is generally used in construction in which the main verb is marked for imperfect aspect, as in the examples (52a-b), which are constructions equivalent to 'progressive' constructions. In (53), the verbe i 'do' is the main verb of the clause, i.e., not in an auxiliary function.

(52) a. ĩgũ mõmẽgũ imĩ

~igu ~bobe-gu i-~bi 3sg:M work-3sg:M.IMPERF do-3sg:M.IMPERF 'He is working.' b. bugu sĩrĩgũ mĩ

bu-gu ~sidi-gu i-~bi

elder-3sg:m die-3sg:m.IMPERF do-3sg:m.IMPERF 'The old man is dying.'

(53) a. ērā iderā

- ~eda i-de-~da
- 3PL do-partic-pl:an.perf

'They have done (it).'

b. yu'u mũ'ũre õpã iaya ã'rībirika

yuu ~buu-de ~opa i-a-ya ~adi-bidi-ka 1sg 2sg-ref so/thus **do**-perf-evid:res say-neg-evid:reas 'I can't say you did so.'

In the perfect constructions in (54a-b), the verb *i* follows a series of verb

constructions with no overt marker for perfect/imperfect aspect. In these cases, the events described by the main verbs are interpreted as perfect, which is marked by the subject agreement marker $-m\tilde{t}/-\sim$ bi/ in the examples in (54).

(54) a. erosārē pāmūgora ibiriyūmī

	edo-~sa-de	~pabu-goda
	DEIC:DIST-ADD-REF	ferment-reach
	i-bidi-~yu-~bi	
	do-NEG-EVID:QUOT/	FOLK-3SG:M.IMPERF
	'He didn't create (e	verything) there.'
b.	yuhun u ta mãgãrīpãi	n ũ mãgãrĩ ibiriyũmĩ ĩg ũ
	yuu-d u -ta	~ba-~gadi-~pabʉ
	one-CLS:day-EMPH	go.uphill-advance-ferment

~ba-~gadi	i-bidi-~yu-~bi	~igu			
go.uphill-advance	do-neg-evid:quot/folk-3sg:m.imperf	3sg:m			
'He didn't come at once, and he didn't come creating at once.'					

The auxiliary verb *i* 'do' can also occur in serial verb constructions, as in (55a-b).

~igu i-~tabu-a-~bi

3SG:M do-help-PERF-3SG.M.IMPERF

'He helped.'

b. ikurimūrīabu yu'usā surara wa'agu

i-kudi-~budi-a-bu yu'u-~sa surara waa-gu do-walk-HAB-PERF-NON3.PERF 1sg-ADD soldier go-3SG:M.IMPERF 'I used to be a soldier.' (lit. 'I walked (lived) being a soldier.')

5.3.4.2 The Auxiliary Verb wa'a 'go'

While the auxiliary *i* 'to do' is used with both perfect and imperfect constructions; the auxiliary verb *wa'a* 'to go' is used to form constructions coding the perfect aspect, as shown in (56). In the examples in (56a, b, c), the verb *wa'a* 'go' verb as an auxiliary codes 'change of state'.

(56) a. igo karia wa'amõ

igo	kadi-a	waa-~bo
3sg:f	sleep-PERF	go-3sg:f.imperf
'She f	ell asleep.'	

b. duhpu turia wa'abu

	d u pu	tudi-a		waa-bi	ŧ					
	branch	break-I	PERF	go-NO	N 3 .PERF	,				
	'The br	anch bi	roke.'							
C.	ohteri s	ĩrĩa wa	'abʉ							
	ote-di		~sidi-a		waa-bŧ	ŧ				
	sow-NO	ЭМ	die-PEF	2F	go-NOI	N 3 .PE	ERF			
	'Plantat	tion die	ed.'							
d.	mẽsirin	ũ kã wa	amã ba	harã						
	~be-sid	i-∼d u ga	a			waa	~ba		baa-~da	
	throw-c	lo.repea	atedly-11	NCEP		go-3	BPL:AN	IMPERF	many-3PL.A	N
	'Many	(people	e) left.'							

5.3.4.3 The Auxiliary Verb arī 'be'

The verb \tilde{art} 'be' is also used to code perfectivity. It is used in constructions in which the main verb is in its participle form, which is coded by the participial morpheme *-de*, as shown in the examples in (57). When glossing sentences that have this type of structure (i.e., main verb the 'participle' suffix *-de* and auxiliary \tilde{art}), my consultants would often translate them as describing an event that combines past and present. For example, sentence (57c) was translated as 'they also lived and live there'. Thus, I treat these constructions and coding (present) perfect aspect.

(57) a. buanügāderā ārīmā ērā

```
bua-~dʉga-de-~da ~adi-~ba ~eda
```

go.downriver-INCEP-PARTIC-3PL:AN.PERF be-3PL:AN.IMPERF 3PL

'They began to go down (to another village).' (repeated from 40)

b. ero ērā yuririrā ārīderā ārīmā ārīkuyāā

edo	~eda	yudi-di-~	-da
DEIC:DIST	3PL	go.down-DIR:towards-3PL:AN.PERF	
~adi -de- ~o	la		~adi-~ba
be-partic-pl:an.perf		ERF	be-3PL:AN.IMPERF

'They went down (moved) and used to live there.' (repeated from 41)

c. gua pagusã eroduhuaderã ãrimã

guapa-gu-~saedodua-de-~da2PL:EXCLgenitor-3SG:M-ADDDEIC:DISTstay-PARTIC-PL:AN.PERF~adi-~ba

be-3pl:An.IMPERF

'Our fathers too have lived there.'

5.4 Serial Verb Constructions

Serial verb constructions (SVC) are defined as "a sequence of verbs which act together as a single predicate, without any overt marker of coordination, subordination or syntactic dependency of any other sort" (Aikhenvald 2006:1). Thus, the concepts coded by each verb in the construction are described as one single event. Aikhenvald uses four parameters to establish a typology of SVCs: 'composition' (symmetry), 'contiguity', 'wordhood of components', and 'marking'. 'Symmetrical' constructions are formed by verbs from the same semantic or grammatical classes (open/unrestricted verbs), where each of them has equal status in the construction as a whole. 'Asymmetrical' constructions are formed by verbs from the 'open/unrestricted' (for example, activity verbs) class plus verbs from the 'closed/restricted' (for example, motion/placement verbs) class. In a continuous SVC, the verbs are placed next to each other without any element intervening between them; the non-continuous SVC can have intervening elements. The verbs in a SVC may form an independent phonological or grammatical word (incorporating) or may not (non-incorporating). Finally, marking of grammatical categories (person/number of the subject, tense, etc.) may occur only once (single marking) or categories may be marked on every component of the construction (concordant marking).

5.4.1 Serial Verb Constructions in Desano

As in other Eastern Tukanoan languages, SVCs are a highly productive morphological process in Desano. The SVCs in Desano can be characterized as being 'continuous' and 'incorporating'; showing both symmetrical and asymmetrical constructions, and having both single and concordant marking. Concordance marking is not very common.⁹⁹

The symmetrical and asymmetrical constructions are illustrated in (58) and (59), respectively.

(58) SYMMETRICAL

yuhunu yahiyehkomera mimi wubirara iyora

⁹⁹ For analysis of serial verb constructions in other Eastern Tukanoan languages, see Stenzel (2007) for Wanano and Piratapuyo; and Gomez-Imbert (2007) for Barasano and Tatuyo.
yuu-~d u	yai-~y	reko-~beda	~bibi	
one-CLS:day heron-		grandmother-COM/INSTR	hummingbird	
wu-bida- ∼da		i-~yo-~da		
fly-play-3pl:an.perf		do-evid:hsay-3pl:an.perf		
'One day, the tuit	iiu bird	and the hummingbird were fly	ing and playing. ¹⁰⁰	

(59) ASYMMETRICAL

eroge arīmāhsībiri gua piamērēa waaka

gʉa	pia-~bede-a	waa-ka
1pl.excl	MOV:out.into-stay-PERF	go-EVID:REAS
'We move	d from there (to here).' (repe	eated from 42b)

Concordant marking does not occur often; however, instance of this type of marking can be found, as shown in (60). Each verb is marked for person/number of the same subject.

(60) peebu kuribu yu'usã

pe-bukudi-bu[yuu-~sa]_{SUBJ}listen-NON3.PERFwalk-NON3.PERF1sG-ADD'I also listened and walked.' (repeated from 33b)

5.4.1.1 The Structure of SVCs in Desano

SVCs in Desano share the structural characteristics found in other languages in the area: they can have up to four verb roots (although the most common SVCs have two or three verb roots); and they can be analyzed as having two basic positions in the SVC

¹⁰⁰ 'Tuiuiú' is the local name for the Jabiru mycteria. (repeated from 38b)

template, which can be simple or complex (cf. Ospina Bozzi and Gomez-Imbert, to appear).

The structures in (61) illustrate the types of SVCs found in Desano.¹⁰¹ In (61a), the simple structure is formed by two verb roots, each occupying one of the two positions of the SVC template, as illustrated with the examples in (62), with the structure inside the brackets.

(61)		position 1	POSITION 2	
	a.	[R ₁	R ₂]	simple SVC structure
	b.	[R ₁	$[R_2 \ R_3]]$	
	C.	$[[R_1 \ R_2]$	R ₃]	
	d.	$[[[R_1 \ R_2] \ R_3]$	R_4]	complex SVC structures
	e.	$[[R_1 \ [R_2 \ R_3]]$	R ₄]	
	f.	$[[R_1 \ R_2]$	$[R_3 R_4]]$	
(62)	a.	ĩgũ yegu mẽrã ke	ogãñg ũ	
		~igu ye-gu-~be	eda	[keo-~gadi]-gu
		3sg:m jaguar-cl	s:trunklike-COM/IN	JSTR measure-advance-3SG:M
		He (was) approad	ching with the jagu	ar cane ('sacred cane').

b. ĩgũpu buhapu yãpeyapu ẽrãrẽ

~igu-pu	bua-p u	[~ ya-peya] -pʉ	~eda-de
3sg:m-contr	pigeon-CONTR	see-be.on.top.of-3SG:M	3PL-REF
(1	• •

'The pigeon observed them from above.' (repeated from 24)

¹⁰¹ Following Ospina Bozzi and Gomez-Imbert (to appear) analysis of SVCs in Yuhup (Makuan), Barasano and Tatuyo (Eastern Tukanoan).

(63)
$$[R_1 \qquad [R_2 \ R_3]]$$

beyeweagārigū iyūmī daha

[beye-[wea-~gadi]]-gui-~yu-~bidaaexplain-extract-advance-3SG:Mdo-EVID:QUOT/FOLK-3SG:M.IMPERFagain'He was explaining and (continuing) analyzing again.' (repeated from 51b)

$$(64) [[R_1 \ R_2] R_3]$$

õrē īgū wirinūgāyāyumī

~o-de ~igu

DEIC:PROX-REF 3SG:M

[[widi-~duga]-~ya]-~yu-~bi

MOV.outward-begin-see-EVID:QUOT/FOLK-3SG:M.IMPERF

'He began to come outside and looked around.' (repeated from 44a)

 $(65) \quad [[[R_1 \ R_2] \ R_3] \qquad \qquad R_4]$

ĩgũsã yākurimāhīdihia goepu

~igu-~sa[[[~ya-kudi]-~bai]-dii]-agoe-pu3SG:M-ADDsee-walk-turn-descend-PERF return-3SG:M.PERF

'He also walked looking back going down (the river), and returned (home).

 $(66) \quad [[R_1 \ [R_2 \ R_3]] \qquad \qquad R_4]$

ĩgũ kuriyãpeoyuria wa'ayũmĩ

~igu [[kudi[-~ya-peo]]-yudi]-a wa'a-~yu-~bi

3SG:M walk-see-end-go.down-perf go-EVID:QUOT/FOLK-3SG:M.IMPERF

'He walked, finished seeing (everything) and went down.'

 $(67) \quad [[R_1 \ R_2] [R_3 \ R_4]]$

pămũpahayããpeoyũmĩ

[[~pabu-paa][-~ya-peo]]-~yu-~bi

ferment-float-see-end-EVID:QUOT/FOLK-3SG:M.IMPERF

'(He) floated (on the river) transforming (and) finished seeing (everything).'

5.4.1.2 Semantics of Verb Roots in SVCs

The SVCs in Desano consist of two types of verbs which occur in dependent position (i.e., after the first verb in the construction): motion verbs, coding displacement (i.e., 'to go', 'to come', 'to move uphill/upriver', etc.); location/position (i.e., 'to be on top of', 'to be behind', 'to be before', etc.); and other verbs coding either stative or agentive events (i.e., 'to walk', 'to see', 'to explain', etc.).

In Desano, dependent verbs in SVCs code the following semantic notions: 'adverbial' notion (manner); notion of 'cause-effect'; 'aspectual' notion; and (deontic) modality.

5.4.1.2.1 Dependent roots coding 'adverbial' notions

In SVCs, adverbial notions of 'manner' are expressed by intransitive motion verbs (cf. Aikhenvald 2006). These intransitive motion verbs, when used as a dependent verb in a SVC, indicate *how* the action/event expressed by the independent verb is performed. In Desano, the most common motion verbs used in this function are *ari* 'to come' and *wa'a* 'to go'. In the examples in (68), the verb *ari* is used as a dependent verb indicating that the event described by the independent verb (i.e., the verb in initial position) occurs

simultaneously with the cislocative motion (i.e., moving towards the speaker or a referent point).

(68) a. iariyūmī īgū

i-adi-~yu-~bi ~igu do-come-EVID:QUOT/FOLK-3SG:M.IMPERF 3SG:M 'He came doing (things).'

b. aiomaa piari ariribu

~ai-~oba-a pia-adi-di-bu

take-wrap-PERF MOV.out.into-come-DIREC:towards-NON3.PERF

'He put (it) on his back and left (the forest) coming.' (repeated from 42a)

The examples in (69) show constructions with the verb wa'a' to go' coding a 'translocative' motion (i.e., motion away from the speaker or a referent point in the discourse). It should be noticed that wa'a [wa'a] is phonologically reduced to [wa] when occurring as a dependent verb.

(69) a. duhawarā mūnīriaribu daha

dua- waa- ~da	~budi-di-adi-bu	daa
return-go-EXRT	MOV:go.uphill-IMPERF-come-NON3.PERF	again
'Let's go back, an	d go up again.'	

b. duhkawake nihku

duka-**waa**-ke ~dikʉ split-**go**-IMP land 'Go dividing the land.' Other verb roots of motion can be used to code different adverbial notions, as shown in (70). For example, the verb root *yudi* 'to go down' in (70a) indicates how the Subject of the sentence is returning to his place of origin. In (70b), the dependent verb *widi* 'to move outward' describes how the Creator of the world came out from one of the holes by the river creating new things in the world. Finally, in (70c), the verb $m\tilde{e}$ 'to throw' indicates that the monkey was in a stationary position in the tree while looking towards the hunter on the ground.

(70) a. ĩgũ mãīyuria ehayũmĩ

~igu ~bai-yudi-a

3SG:M turn.around-go.down-PERF

ea-~yu-~bi

arrive-EVID:QUOT/FOLK-3SG:M.IMPERF

'He turned around going down (the river), and arrived.'

b. õgere peragobege pămũwirigu arigu pare

~o-ge-de	peda-gobe-ge	~pab u-widi -g u
DEIC:PROX-LOC	port-hole-LOC	ferment-MOV:outward-3SG:M
adi-g u	pade	
come-3sg:m	then	

'Then, coming to create (while) leaving from that hole in the port.'

c. yămēdihumī yu'ure ārīmī

~yaa-~be(de)-diu-~bi yuu-de
see-stay-MOV:downwards-3SG:M.IMPERF 1SG:M-REF
'(The monkey) looked at me (from above).'

5.4.1.2.2 Dependent roots coding aspect

SVCs are often used to provide aspectual information regarding an activity or state. Some common aspectual notions coded by SVCs are: indicating the beginning or end of an event; indicating the regular performance of the action; indicates its duration; and emphasizing if the event is completed, among others. The examples in (71) show some of the aspectual notions that are coded by SVCs. In (71a), the verb $n\bar{u}g\bar{a}$ 'to begin' codes 'INCEPTIVE' aspect. The verb $m\bar{u}r\bar{i}$ 'to grow' in (71b) is commonly used as a dependent verb coding 'HABITUAL' aspect in SVCs. In (71c), the verb *peo* 'to end' codes the total completion of the event; and in (71d), the verb $s\bar{u}$ 'to finish' indicates the termination of an event.

(71) a. imīsõkāyāge īgū dui wāiku gohanūgādi

i-~biso-ka-ya-ge			dui	
DEM:PROX-squirrel-CLS:triangular-CLS:creek-LOC			Luis	
~wai-kʉ	goa-~ duga- di			
name-VBLZ	write-begin/INCEPT-IMPERF			
'From squirrel creek, he, named Luis, began to write.' (repeated from 55b)				

b. ārīkāmūrību yu'u õ teresitare dos āyo meo

~adi-ka-~ bud	i-bʉ	y uu	~0
be-EXIST-grow	w/HAB-NON3.PERF	1sg	DEIC:PROX
teresita-de	dos ano meio		
Teresita-REF	two and half years		

'I lived in Terezita for two and half years.' (repeated from 8b)

c. ĩgũ kuriyāpeoyuria wa'ayũmĩ

~igu kudi-~ya-peo-yudi-a
3SG:M walk-see-end/TOTAL-go.down-PERF
wa'a-~yu-~bi
go-EVID:QUOT/FOLK-3SG:M.IMPERF
'He walked, finished seeing (all) and went down.' (repeated from 65).
d. iriyẽ dukãsũbu

idi-~yedu-~ku-~su-bupadeDEM:PROX-CLS:topicput.down-FRUSTR-finish/COMPL-NON3.PERFthen'Then he abandoned this project.''

5.6.1.2.3 Dependent roots coding modality

Dependent verbs in SVCs are also used to code the notion of deontic modality (i.e., wishes, ability). These deontic notions are expressed by dependent verbs of mental process, as shown in (72a) with the verb *dua* 'to want' coding 'DESIDERATIVE' mode. The verb *māhsī* 'to know' is generally used to code 'ABILITY', as shown in (72b); but it can also be used to code 'permission' as in (72c).

(72) a. ĩgũ buriyāduarimī

~igu ~budi-~ya-**dua**-di-~bi

3SG:M be.hard-see-DES-IMPERF-3SG:M.IMPERF

'He wanted very much to see you.' (repeated from 31)

b. eroge ārīmāhsībiri gua piamērēā wa'aka pare

edo-ge ~adi-~basi-bidi gua pia-~bede-a

DEIC:DIST-LOC be-know-NEG 2PL:EXCL MOV:out.into-stay-PERF
waa-ka pade
go-EVID:REAS then
'Not being able to live there, we moved (here).
c. ĩgũ ômũhũ ãrã ã'rĩmãhsĩmĩ
~igu ~o-~buu ~adi-a ~adi-~basi-~bi
3SG:M DEIC:PROX-CLS:origin be-PERF say-know-3SG:M.IMPERF

'He can say he is from here.'

5.5 Verb Morphology

In the previous sections, different types of verbs were described, including a description of SVCs and how dependent verb roots can be used to code adverbial, aspectual and deontic modal notions. This section presents a description of the grammatical morphemes that can attach to independent verb roots and SVCs, as shown on Table 5.1.

The first three positions ('manner', 'aspect', 'mode') are coded by dependent verb roots in serial verb constructions, as described in Section 5.4.1.2.

5.5.1 Morphemes Coding Manner

Adverbial-type notions of manner are commonly expressed by verb of motion (see Section 5.4.1.2.1). In these expressions, the verb of motion indicates 'how' the action/event is performed. Some common verbs of motion used dependently to code adverbial notions of manner are listed in (73).

(73)	a.	/-~ be /	/tu-~ be -yoa/
		'to throw'	lean-throw-suspend
			'pushing one's into (something)'
	b.	/-tu/	/pua-tu-~ba-~ga/
		'to lean/put against	stumble-lean-MOV:quickly-MOV:advancing
			'stumble leaning'
	C.	/-widi/	/~pab u-widi /
		'to move outward'	ferment-MOV:outward
			'to go out transforming'
	d.	/-sidi/	/pa- sidi- ~bʉta/
		'to do repeatedly'	hit-do.repeatedly-be.in.front
			'go ahead protecting'
	e.	/-pi/	/~bo'be -pi /
		'to drop/release/quit'	/work-quit/
			'quit working'

5.5.2 Morphemes Coding Aspect

The second slot in the verb morphology template can be occupied by dependent verb roots coding aspect (see Section 5.4.1.2.2). These roots code aspects of both active and stative verbs, as shown in (74). Some common aspectual notions coded by dependent roots in Desano are: inception (74a), duration (74b-d), completion (74e), and habituality (74f).

(74)	a.	/-~duga/	/were-~adi-~duga/
		'begin/inception'	tell-say-INCEP
			'start narrating'
	b.	/-~dugu/	~ai-~b uu -~dugu
		'stand/do continuously'	take-lift-CONT
			'take it over'
	C.	/-doa/	~gʉya -doa
		'sit/do for a period of time	think-DUR
			'meditate'
	d.	/-~gado/	~abu-~gado
		'continue/do continuously'	fix-cont
			'getting better'
	e.	/-peo/	kudi-~ya- peo
		'end/do exhaustively	walk-see-EXRT
			'walk and see exhaustively'
	f.	/-~bʉdi/	oo-∼ bʉdi
		'go upriver/do it habitually'	give-HAB
			'be used to give'

5.5.3 Morphemes Coding Deontic Modality

The third slot in the verb morphology template is occupied by dependent verb roots coding deontic modality. As was discussed in Section 5.4.1.2.3, deontic modality is expressed by verbs of mental process, such as *dua* 'to want', and *mãhsĩ* 'to know', which

are used to code 'DESIDERATIVE', as in (75a), and 'ABILITY', as in (75b-c), modes. See also full sentence examples in (72).

(75)	a.	/-dua/	ba -dua
		'eat/desiderative	eat-DEON
			'want to eat'
	b.	/~basi/	~a'di-~basi
		'know/permission'	say-DEON
			'can say'
	C.	/~basi/	~adi -~basi
		'know/ability'	be-know
			'be able'

5.5.4 Morphemes Coding Negation

The negative morpheme *-biri* /-bidi/ occupies the fourth position in the verb template. In the absence of dependent verbs in positions 1-3, the negative suffix can be suffixed directly to the independent verb root. The negative *-biri* can be attached to any type of verb: descriptive (76a), stative (76b), activity (76c), motion (76d), perception (76e), or mental process (76f).

٢

(76) a. igo műkűbirigo

igo ~buku-bidi-go 3SG:F be.happy-NEG-3SG:F 'She is not happy' b. gua gahipu wa'amuara aribirika

	gua	gai-pʉ		waa-~bʉa-~c	la
	2PL.EXCL	EXCL other-CONTR go-go.downriver-3PL:AN.PE		iver-3pl:AN.PERF	
	be~adi- bidi- ka				
	be-neg-predic				
'We are not going to move.'					
C.	ĩgũ mõmẽb	oirimĩ			
	~igu	bobe- bidi -	~bi		
	3sg:m	work-NEG-	-3sg:m	IMPERF	
	'He doesn'	't work'			
d.	. ühtāmūgere wabita guya ārīka				
	~ʉtabu-ge-	·de	wa-bi	di-ta	gʉya-~adi-ka
	rapids-LOC	-REF	go-NE	G-ADMON	be.dangerous-be-EVID:REAS
	'Don't go i	in the rapid	s becau	ise it is danger	ous.'
e.	mãrĩ pagus	sāmārā mār	î yâbiri	bu	
	~badi	pa-g u	~s	aba-~da	~badi
	1PL.INCL	genitor-3so	∃:M va	gina-3PL:AN	1pl.incl
	~ya-bidi-b	u			
	see-NEG-N	ON3.PERF			
	'We didn't	see our and	cestors	(grandparents).'
f.	ĩg ũ mãhsĩb	oirimĩ			
	~igu	~basi -bidi -	~bi		
	3sg:M	know-NEG	-3sg:m	I.IMPERF	

'He doesn't know.'

In constructions with auxiliary verbs, the negative morpheme is attached to the auxiliary, and not to the main verb, as shown in (77).

(77) a. erosārē pāmūgora ibiriyūmī

edo-~sa-de ~pabu-goda DEIC:DIST-ADD-REF ferment-reach

i-bidi-~yu-~bi

do-neg-evid:quot/folk-3sg:m.imperf

'He didn't create (everything) there.' (repeated from 53a)

b. mãhārõ nīmāpū wuaro wa'abiro ārību īgūrē

~baa-do	~diba-p u	wʉa-do	waa-bi-do
QUANT-NOM	poison-CONTR	be.big-NOM	go-neg-non3.imperf
~adi-bʉ	~igu-de		
be-NON3.PERF	3sg:m-ref	7	

'There wasn't enough poison to (kill) him.'

Miller (1999:136) postulates that the negative morpheme *-biri* becomes *-bea* before morphemes she identifies as 'present tense' and 'visual evidence'. In my data corpus, it is possible to see a tendency of the morpheme *-biri* to be realized as *-bi*, as in example (77b) above. For the moment, the occurrence of *-be* is considered here as the reduced *-bi* which may go under vowel lowering when occur in near a non-high vowel. However, this is a topic for further investigation.

5.5.5 Morphemes Coding Grammatical Aspect

Section 5.4.1.2.2, showed that dependent roots in serial verb construction can give aspectual information of the event expressed by the independent verb root. There are also two grammatical aspect markers that occur in the fifth position in the verb template: -a'perfect' and -di 'imperfect'. These aspect markers allow for perfective and imperfective interpretations of the situation (i.e., state, motion, mental process, etc.) described by the clause. A third morpheme that occurs in this slot is -de 'participle', which may co-occur with either -a or -di. The perfect aspect morpheme -a is always followed by the participial -de, as shown in (78a-b). The examples in (79) show the use of the imperfect aspect marker -di, which may be followed by the participle marker -de, as shown in (79c). Examples with the participial morphemes without preceding grammatical aspect morphemes are shown in (80).

(78) a. erogeta i wīrāya kõmõādero ārību

edo-ge-tai~wida=yaDEIC:DIST-LOC-EMPHDEM:ANAPHDesano=POSS~kobo-a-de-do~adi-bustraggle-PERF-PARTIC-NON3.IMPERFbe-NON3.PERF'There, this (traditional knowledge) of the Desanos was abandoned.'

b. yuriaderā ārīmā

yudi-a-de-~da ~adi-~ba go.down-PERF-PARTIC-3PL:AN.PERF be-3PL:AN.IMPERF 'They went down (the river to another village).' (79) a. ĩgũ yo'gu wãikudi ãrīmĩ

~igu yogu wãĩ-ku-di ~adi-~bi
3SG:M stutter name-VBLZ-IMPERF be-3SG:M.IMPERF
'He was called stutterer.' (repeated from 1)

b. ĩgũ yo'gu wãikudi ãrīmĩ ĩgũ yo'gu bugu

elder-3sg:M

'He, the stutterer, was called old stutterer.' (repeated from 29)

c. ērā ārīrīrē

~eda ~adi-di-de

3PL say-IMPERF-PARTIC

'They (have) said.'

(80) a. buanügāderā ārīmā ērā

bua-~duga-de-~da~adi-~ba~edago.downriver-INCEP-PARTIC-3PL:AN.PERFbe-3PL:AN.IMPERF3PL'They began to go down (to another village).' (repeated from 40a)

b. buaderā ārīmā

bua-**de**-~da ~adi-~ba go.downriver-**PARTIC**-3PL:AN.PERF be-3PL:AN.IMPERF 'They went down (to another village).' (repeated from 40b)

5.5.6 Morphemes Coding Clause Modality

Morphemes that are used to mark clause modality and occupy position six in the verb template are discussed in this section. Table 5.2 summarizes the morphemes according to the subcategories to which they belong. The subcategories are labeled following Stenzel (2004:331). The subcategories correspond to the three types of sentences in which these morphemes occur: statements - which can be realis (marked with evidentials) and irrealis (predictive or speculative); interrogatives; and 'oriented' (commands and permission).

Table 5.2 Clause modality	Table	5.2	Clause	modality
---------------------------	-------	-----	--------	----------

CLAUSE MODALITY				
STATEM	STATEMENTS			
REALIS	IRREALIS			
EVIDENTIALS		/ -ke / IMP		
\oslash VIS	/-ka/ predict	/ -ta / ADMON		
/-ku/ NVIS/AUD	/-bo/ specul	/ -~da/ EXRT		
/-~yo/ HSAY		/ -ku / ADVER		
/-~yu/ QUOT/FOLK				
/-ya/ RESUL				
/-ka/ REAS				
INTERROGATIVES				
/-di/ INTER				

5.5.6.1 Realis Statements: Evidentials

The markers of evidentiality occur in the sixth position in the verb template. This section presents a description of the evidential system of Desano according to the current typology of evidentiality. The analysis presented here is not intended to be exhaustive; instead, it aims to show the complexity of the system.

5.5.6.1.1 Evidentials in cross-linguistic perspective

The study of evidentials in Native American languages is relatively new. In his description of Kwakiutl (Kwak'wala), a Wakashan language of Vancouver Island, Frans Boas (1911) described a set of unusual suffixes that expresses 'the *source of subjective knowledge* - as by hearsay, or by a dream'' (p. 443). Other early descriptions of evidentials also come from other Native (North) American languages: Hannis, a Coos language of the coast of Oregon (described by Leo Fratchtenberg in 1922); Wintu, a Wintuan language of California (described by Dorothy D. Lee in 1938), and Pawnee, a Caddoan language spoken in North Dakota and Oklahoma (described by Douglas Parks in 1976) (cf. Campbell 1997, Mithum 1999).

The term 'evidentials' was first used by Roman Jacobson in (1957 [1971]). He describes it as a speaker's report of an event on the basis of someone else's report (quotative, i.e. hearsay evidence), of a dream (revelative evidence), of a guess (presumptive evidence), or of his own previous experience (memory evidence).

According to Anderson (1986) and Portner (2009), the terms 'evidentials' and 'evidentiality' can be defined as following:

- *Evidentiality* is defined as a broad semantic category which is found in some form in every language, used to express speaker's assessment for what she says.
- *Evidential* is defined as a syntactically and semantically constrained grammatical marker which is used to express evidentiality.

English also codes evidentiality in some modal verbs, e.g. *Mary must sing* (inferred from the fact that the rest of her family sings, so probably she also sings). Thus, in English, evidentiality is *lexically* marked, as shown in (81), where the words in bold form the expression coding evidentiality.

- (81) a. I heard the rain falling.
 - b. Someone told me it is raining.

Desano (and all Eastern Tukanoan languages) is a language in which evidentiality is *grammatically* marked by evidential suffixes, as shown in (82).

(82) a. kārāyā uhegū ikumī

~karayaue-~gui-ku-~bichickenkill-3sg:Mdo-EVID:NVIS-3sg:M.IMPERF'He is killing the chicken.' [I can hear the chickens]

b. yuhusibu õã painõmēã pai waayõrã

yuu-sibu	pai-~dobe-a	pai	waa-~yo-~da
one-while	priest-female-PL:AN	priest	go-evid:rep-3pl:an.perf
'Once, when t	the missionaries where	travelli	ng' [I heard from others]

According to the analysis presented here, Desano has five evidentials that are overtly marked with a suffix on the verb, plus an unmarked (visual) evidential, as listed in Table 5.3.

5.5.6.1.2.1 The distinction of information source. Each of the evidential markers listed in Table 5.3 indicates a type (source) of information in a proposition. These evidentials fall within the set of evidential meanings found cross-linguistically (cf. Aikhenvald 2004, Anderson 1986, Bybee 1985, de Haan 1999, Willet 1988). Aikhenvald (2004) lists six semantic parameters used with languages with grammatical evidentiality: 'visual', 'sensory', 'inference', 'assumption', 'hearsay', and 'quotative'. Aikhenvald points out in her typology, that "no systems have been found with all six types expressed" (p. 367).

Table	5.3	Desano	eviden	tials
			• • • • • • • • •	

EVIDENTIAL	GLOSS
-ku	auditory
-~yo	reported
-~yu	quotative/folklore
-уа	inference: results
-ka	inference: reasoning
Ø	visual

DIRECT			Indirect				
	Attested		F	Reported		Infe	erring
Visual	Auditory	Other sensory	Second- hand	Third- hand	Folklore	Results	Reasoning
Ø	-k	ι (U	-~y0	-	~уи	-ya	-ka
	EVID	:NVIS	EVID:HSAY	EVID:Q	UOT/FOLK	EVID:RES	EVID:REAS

Table 5.4 Types of Evidence (adapted from Willet 1988:57)

The characterization of types of evidentials adopted to describe the evidentials in Desano follows the framework proposed in Willet (1988) for the categorization of types of evidentials, as shown in Table 5.4. The Desano evidential markers have been added to illustrate how they fit in Willet's typology.

5.5.6.1.3 Describing the evidential markers of Desano

5.5.6.1.3.1 Direct visual evidence. In Desano, visual evidence is unmarked.¹⁰² Thus, sentences with no overt marker of evidentiality imply the speaker has witnessed the event. Sentence (83) implies that the speaker saw the person chopping firewood.

- (83) ĩgũ pea tabegu imĩ
 - ~igu pea tabe-gu i-~bi

¹⁰² This has also been claimed by Kaye (1970).

3SG:M firewood chopp-3SG:M. do-3SG:M.IMPERF

'He is chopping the firewood.'

5.5.6.1.3.2 Nonvisual evidential. The morpheme -ku is used to code *direct attested* evidential covering information acquired by auditory and other senses (taste, smell, touch). I call -ku 'nonvisual', indicating information based on direct evidence other than sight, as illustrated in (84) and (85) in two different contexts.

Auditory context I: Speaker is inside home and hears the noise of firewood being chopped.

(84) ĩgũ pea tabegu ikumĩ

~igu pea tabe-gu i-ku-~bi 3SG:M firewood chopp-3SG:M do-EVID:NVIS-3SG:M.IMPERF 'He is chopping the firewood.'

Auditory context II: Speaker is inside home and feels a heat (from a fire) and hears noise of things burning. He doesn't see the fire.

- (85) ērā soera soekumā
 - ~eda soe-ku-~ba
 - 3PL burn-EVID:NVIS-3PL:AN.IMPERF

'They are burning (something).'

5.5.6.1.3.3 Reported: Hearsay evidential. The morpheme $-\gamma yo$ is used to code *indirect reported* evidential indicating that the information was acquired from some other person who might have (directly) witnessed the event described. I call $-\gamma yo$ 'hearsay' to indicate information obtained through a third party, as illustrated in (86) and (87) in two different contexts.

Reported context I: Speaker is talking about a school of a nearby village. The speaker has not seen the school, but was told about it from someone who had seen it and told him about it.

(86) yuhu eskola iayõrã ĩrã

yuu eskola i-a-~yo-~da ~ida one school do-PERF-EVID:HSAY-3PL:AN.PERF 3PL 'They built a school.'

Reported context I: *Speaker is telling an anecdote he heard from his father, which he believes to be true.*

(87) yuhusibu õã painõmẽã pai wayõrã

yuu-sibu pai-~dobe-a pai wa-~yo-~da one-while priest-female-PL:AN priest go-EVID:HSAY-3PL:AN.PERF 'Once, when the missionaries where travelling...'

5.5.6.1.3.4 Reporterd: Quotative/folklore evidential. The morpheme -~yu also codes *indirect reported* evidential generally appearing in traditional narratives (folklore). It has a 'near-quotation' function; however, no particular referent for the original speech can be identified as the source of information. This marker is also used in situations where the speaker has heard about the event from someone who has not directly witnessed the event. This marker is used for rumors or news that cannot be 'directly' verified, as shown in (88) and (89) in two different contexts.

Reported 'folklore' context: *a line from a tradition narrative about the Desano mythology*.

(88) wirinügäyä ĩgũ mũrũ uhupuri iyũmĩ

widi-~duga-~ya ~igu ~budu

leave-stand.up-see 3SG:M tobacco

uu-pudi-i-~yu-~bi

suck-blow-do-evid:Hsay-3sg:M.IMPERF

'He left from home looking around (and) smoking tobacco.'

Reported third-hand context: Same as (87) above, but this time the speaker was

told about the school from someone who ha not seen the school.

(89) yuhu eskola iayõrã ĩrã

yuu eskola i-a-~yu-~da ~ida one school do-PERF-EVID:HSAY-3PL:AN.PERF 3PL '(I've been told that) they built a school.'

5.5.6.1.3.5 Inferred: Result evidential. The morpheme -ya codes indirect

inferring evidential. It is used in utterances that express the speaker's conclusion about an event or state based on some type of (indirect) observed results (i.e., the speaker observes results, rather than the actual event causing the result), illustrated in different contexts in (90) and (91). I call this evidential 'RESULT' for short.

Inferred results context I: Speaker infers based on observable evidence (i.e. by seeing the plants and the ground outside are wet) that it rained during the night.

(90) dopa yāmīrē dehko mērāyā

dopa ~yabi-de deko ~beda-ya today night-REF water fall-EVID:RES 'It rained last night.' Inferred results context II: Speaker meets with a relative many years since they have last seen each other. The speaker's nephew looks old and tired.

(90) mõmēya mũ'ũ wãĩmũ

~bobe-ya ~buu ~waibu
work-EVID:RES 28G nephew
'You've worked (hard), nephew.'

5.5.6.1.3.6 Inferred: Reason evidential. The morpheme *-ka* is also used to code *indirect inferring* evidential. It is based on the speaker's own previous experience, resulting in reasoning suppositions or on his cultural/historical/physical knowledge, resulting in assertions of facts. I call this evidential 'assertion', and its occurrence is shown in different context in (91) and (92).

Reasoning context I: Speaker infers the situation based on the 'reasoning' that his brother is usually working when the speaker returns home.

(91) yu'u goekũ mõmẽgũ yãduaka

yuu goe-~ku ~bobe-gu ~yaa-dua-ka

1sg return-when work-IMPERF see-DES-EVID:REAS

'I hope to see you working when I return.'

Reasoning context II: It is a general knowledge in the Vaupés Region that the rapids are dangerous, thus, a sentence like (92) is generally used to describe some of the dangerous rapids.

(92)	ũ htāmūgere wabita g u ya ārīka				
	~utabu-ge-de	wa-bidi-ta	guya-~adi- ka		
	rapids-LOC-REF	go-NEG-EMPH	be.dangerous-be-EVID:REAS		

'Don't go on the rapids because it is dangerous.'

A detailed examination of the morphosyntax and semantics of evidential markers is likely to yield reliable information on the status of evidentials as a grammaticalized evidential system. This is an area of ongoing investigation by the author.

The relationship between evidentials and epistemic values is, according to the claim made by Aikhenvald (2004), that "evidentiality markers occur in the mood and modality slot in verbal word, and are thus mutually exclusive with conditional, imperative, interrogative markers and so on" (p.11). It seems to be the case that, in Desano, evidentials can be analyzed as a (special kind of) epistemic modals, i.e, evidentials contribute to truth conditions. Finally, according to the current typology of evidentiality (Aikhenvald 2004), Desano is unusual in which it expresses the six semantic parameters *-- visual, sensory, inference, assumption, hearsay*, and *quotative*, a condition Aikhenvald claimed does not occur in the languages surveyed.

5.5.6.2 Irrealis Statements

There are two morphemes conveying that nonfactuality of an event: -ka 'prediction', as shown in (93) and -bo 'speculation' as shown in (94).

It should be noted that the speculative morpheme is always followed by the prediction morpheme.

(93) a. gua gahipu waamura aribirika

guagai-puwa-~budi~adi-bidi-ka1PL.EXCLother-CONTRgo-go.uphillbe-NEG-PREDIC'We will not move to another place.'

b. yu'u wārõ wa'auka ārīyūmī

(94)

	y uu	~wa-do	wa-(g)ʉ-ka		
	1sg	be.good-NOM	go-3sg:m-p	REDIC	
	~adi-~	-yu-~bi			
	say-EV	VID:QUOT/FOLK•	-3sg:m.imper	F	
	ʻʻI wi	ll be fine," he s	aid.'		
a.	ĩgũ nõ	ómēk u põrāk u p	ārāmērāk u w	a'gā ikū mū'ūrē ehatuab	iriboka ãrīyũmĩ
	~igʉ	~dobe-kʉ	~po-~da-kʉ		
	3sg:m	female-conj	progenee-PL	AN-CONJ	
	~pa-~	da-~beda-kʉ		waa-~ga	i-ku
	grande	child-pl:an-co	m/inst-conj	go-MOV:advance	do-ADVERS
	~b uu -de ea-tua-bidi- bo -ka				
	2sg-ref arrive-lean-NEG-SPECUL-PREDIC				
	~adi-~yu-~bi				
	say-EVID:QUOT/FOLK-3SG:M.IMPERF				
	"When he has a wife, children and grandchildren, it will be difficult for				

you to live," he said.'

b. wuatariro wa'aboka

wʉa-tari-ro waa-**bo**-ka

be.big-be.long-NOM go-spec-predic

'(The story) will be too long.'

5.5.6.2.1 Interrogative

The morpheme *-ri* /-di/ is used to code questions, as shown in (95). Although the interrogative morpheme is not followed by personal ending morphemes for animates, it may occur with the nonthird person morpheme referring to an inanimate entity, as shown in (96).

(95) a. mũ'ũ ariri ~buu adi**-di**

2sg come-inter

'Are you coming?'

b. di'i kurumāhārā ārīrī õā nāhsēāpu

	dii	kudu-~baa-~da	~adi- di	~ o-a
	which	knot-location-PL:AN	be-inter	DEIC:PROX-PL:AN
	~dase	-a-p u		
	Tukan	0-PL:AN-CONTR		
	'Whic	h clan are these other	Fukano from?	2
(96)	a.	yẽ'ẽ duyariro ãrīyũm	7?	
	~yee	duya- di-do	~adi	-~yu-~bi
	what	lack-INTER-NON3.IM	PEF say-I	evid:quot/folk -3 sg:m.imperf
	"Wha	at is missing?" he said.	,	

The interrogative morpheme may occur also with evidential markers (97a);

irrealis morphemes coding 'speculation' and 'prediction' as shown in (97b) and (97c), respectively. It should be noted that the morpheme *-ka* 'prediction' is realized as [-ku] when followed by the interrogative marker.

(97) a. di'iru gahsiru eheayuri?

dii-dugasi-duea-a-~yu-diwhich-CLS:roundcanoe-CLS:roundarrive-PERF-EVID:QUOT/FOLK-INTER'Which canoe has arrived?'

b. nõã ērārē siubiragu ārīkuri?

~doai-~da-desiu-bida-gu~adi-ka-diwhoDEM:PROX-PL:AN-REFcall-play-3SG:Mbe-PREDIC-INTER'Who will be the captain of our team?'

c. mãrī irire ãīwāgānugābiribokuri

~badi idi-de ~ai-~waga-~duga-bidi-bo-ka-di
1PL.INCL DEM:ANAPH-REF take-emerge-begin-NEG-SPEC-PREDIC-INTER
'(Why) don't we restart this (project)?'

5.5.6.3 'Oriented' Modality

The markers of oriented modality are: *-ke* 'imperative'; *-ta* 'admonitive'; *-da* 'exhortative' and *-ku* 'adversative'. The label 'oriented' modality is used here following Stenzel's (2004:386) adaptation of the terminology 'speaker-oriented modality' proposed by Bybee et al. (1994), which claims that these subgroup of modality 'do not report the existence of conditions on the agent, but rather allow the speaker to impose such conditions on the addressee (Bybee, et al:179). Thus, in Desano, this subgroup of modality is coded by different morphemes.

The imperative morpheme *-ke* is used quite often to code demands or requests as shown in (98a-c). The admonitive is used to code 'warnings' for events that might have

negative consequences; for example, navigating through some rapids in the river (98d), or warning someone not to go to a party (where there will be too much drinking), as shown in (98e). The 'adversative' oriented modality is shown in (98f). The only exhortative example found in the data is shown in (98g), coding request.

(98) a. mũ'ũã õ duhoke

~b uu	~0	duo -ke
$2 \mathrm{PL}$	DEM:PROX	stay-IMP

'You stay here.' (repeated from 22)

b. duhkawa'ake nihku

duka-waa-ke \sim diku

split-go-IMP land

'Go dividing the land.'

c. arike

adi**-ke**

come-IMP

'Come.'

d. ühtämügere wabita guya ärika

~utabu-ge-de	wa-bidi- ta	gʉya-~adi-ka			
rapids-LOC-REF	go-NEG-ADMON	be.dangerous-be-EVID:REAS			
'Don't go on the rapids because it is dangerous.' (repeated from 75d)					

e. mũ'ũ bohsenũrẽ wa'abita

~buu bose-~du-de waa-bi(di)-ta

2SG party-CLS:day-REF go-NEG-ADMON

'Don't go to the party.'

	Don't go to the party.			
g.	ĩgũ nõmẽku põrãku pārāmērāku wa'gā ikũ mũ'ũrẽ ehatuabiriboka ārīyūmī			
	~igu ~dobe-ku ~po-~da-ku			
	3SG:M female-CONJ progenee-PL:AN-CONJ			
	~pa-~da-~beda-k	u	waa-~ga	i-ku
	grandchild-PL:AN-COM/INST-CONJ go-MOV:advance do-ADVERS ~b uu -de ea-tua-bidi-bo-ka			
	2sg-ref arrive-lean-Neg-specul-predic			
	~adi-~yu-~bi			
	say-EVID:QUOT/FOLK-3SG:M.IMPERF			
	"When he has a wife, children and grandchildren, it will be difficult for			e difficult for
	you to live," he said.' (repeated from 94) duhawarā mūrīrīaribu daha			
f.				
	dua-waa-~ da	~bʉdi-di-adi-bʉ		daa
	return-go-EXRT MOV:go.uphill-IMPERF-come-NON3.PERF again			again -

'Let's go back, and go up again.' (repeated from 68)

5.5.7 Agreement Markers

The last position in the verb morphology template is for the category of personal pronominal agreement; these are morphemes that agree with the subject in gender and number. Besides coding person and number of the subject, these personal endings also code perfect and imperfect aspect, as shown in Table 5.5. These morphemes are obligatorily marked on the verb root in noninterrogative and nonoriented utterances.

The examples in (99) show verb inflection with personal agreement in the perfect aspect and with the unmarked visual evidential. In (99a) the personal ending *-bu* 'non-third person perfect' agrees with the first person singular subject *yu'u*. In (99b), *-pu* agrees with the 'third person singular masculine' subject *Carlos*, and also describes a perfective event. The personal ending *-po*, in (99c) shows agreement with the 'third person singular feminine' subject *igo* 'she'; in (99d), the morpheme *-~da* marks agreement with the third person subject *~eda*.

Table 5.5	Markers	of personal	agreement

VERB	PERSONAL PRONOMINAL		
Root	(AGREEMENT) SUFFIXES		
ROOT _{INDEP}	PERFECT	IMPERFECT	
(+ROOT _{DEP})	/-bʉ/ [-bʉ]	/do/ [-ro]	
	NON3	NON3	
	/-pʉ/ [-pʉ]	/-~bi/ [-mĩ]	
	3sg:m	3sg:m	
	/-po/ [-po]	/-~bo/ [-mõ]	
	3sg:f	3sg:f	
	/-~da/ [-rã]	/-~ba/ [-mã]	
	3PL	3PL	

(99) a. pebu kuribu yuusã

	ре- b ʉ		kudi- bu	yuu- ~sa
	listen-	NON 3. PERF	walk- NON3.PERF	1sg-add
	'I also listen and walked.'			
b.	o. karu pīrūrē weheapu			
	karu	~pidu-de	wee-a-pu	
	Carlo	s snake-REF	kill-stat- 3sg:m.per	F
	'Carlos killed the snake.'			
c.	z. igo yuhugogā wa'apo			
	igo	yuu-go-~ga	waa-po	
	3sg:f	one-3SG:F-DIN	Mgo- 3sg:f.perf	
	'She went alone.'			

d. ero ãrīyõrã ērã pare

edo ~adi-~yo-~da ~eda pade

DEIC:DIST be-EVID:HSAY-3PL.PERF 3PL then

'Then, they lived there.' (repeated from 7a)

Examples of subject agreement for each personal ending coding imperfective aspect are given in (100).

(100) a. eroge nĩgĩyũro irigu

do-ge ~digi-~yu-do DEM:DIST-LOC be.standing/stay-EVID:QUOT/FOLK-NON3.IMPERF i-ku DEM:PROX-CLS:tree 'This hill was standing/stay over there.' (repeated from 19)

b. ĩgũ mõmẽgũ imĩ

	~igu	~bobe-gu		i -~bi	
	3sg:m	work-3SG:M.IM	IPERF	do-3sG:	M.IMPERF
	'He is working.' (repeated from 52a)				
c.	igo	dihsiporo nĩgĩn	nõ		
	igo	disi-podo		~digi -~l	00
	3sg:f	mouth-CLS:cur	ve	be.stand	ling/stay- 3sg:F.IMPERF
	'She is standing by the door.' (repeated from 18)				
d.	ĩgũ waharã i yebapusãrẽ baharã ãrĩkãyũmã				
	~igu	waa-~da	i	<u>y</u>	yeba-p u -~sa-de
	3sg:m	enmity-PL:AN	DEM:PF	rox 1	and-CONTR-ADD-REF
	baa-~d	la ~adi-ka	ı-∼yu-~	ba	
	many-l	PL:AN be- EXIS	T-EVID	:QUOT/FC	DLK- 3pl:an.imperf

'There were many of his enemies in this land.' (repeated from 8b)

5.6 Summary

This chapter outlined the main characteristics of the morphology of Desano verbs. It identified three main classes of verbs: stative, non-stative verbs, and auxiliary verbs. The class of stative verbs is composed of copula verbs, verbs coding non-existence, possessive verbs, verbs indicating location/position and 'adjectival' stative verbs. The class of non-stative verbs is composed of active verbs, verbs of motion, verbs of placement and verbs of perception and mental processes. Three auxiliary verbs were identified: *i* 'do', *wa'a* 'go' and *ārī* 'be'. It was shown that serial verb construction is a productive morphological process. Dependent roots in these constructions function to code 'adverbial' notions of manner, aspect, and modality.

The grammatical morphemes that occur in the verb were also described, focusing on the description of the highly complex evidential system and other markers of clause modality. The personal ending morphemes, which occur in the last slot of the verb template, were described as markers of subject-verb agreement.

CONCLUSION

This dissertation provided a typologically informed grammatical description of Desano. This chapter presents an overview of Desano phonology. In Chapter 1, I presented a description about the Desano people and their language, as well as a history of the research. Desano is an endangered Eastern Tukanoan language with some 300 fluent speakers.

In the second chapter, I described the phonology of the language. Desano phonemic inventory is comprised of 11 consonants, and six vowels, and their variants. I showed that the status of the glottal segments 2 and h as full consonant segments is debatable. The chapter also presents a description of the suprasegmentals of Desano, starting with a description of the syllable structure. Stress and tone were treated as a mixed system creating an interaction of *lexical tone* and *metrical tone*. The other suprasegmental trait presented was nasalization. I provided a description of the system and its basic process of nasal spreading. Three morphophonemic processes were described: vowel deletion, vowel fusion, and syllable reduction. The chapter ends with some information on the Desano orthography which I adopt throughout the dissertation.

In Chapter 3, I defined the notion of 'word' in Desano according to phonological and grammatical criteria. There are only two open classes of words can be identified in the language: nouns and verbs. Desano does not have adjectives. Adjectival and adverbial notions are derived from nominal and verbal roots.
The nominal morphology of Desano was described in Chapter 4. It presented the basic structure of nouns and the different types of nouns, including the ones derived from particles. It was shown that nouns in Desano are divided into two main types: animate and inanimate. These nouns are subcategorized according to the noun-class markers they take. Animate nouns take general class markers (gender markers) and inanimate nouns take specific markers (classifiers). The general class markers were also described in this chapter, followed by a discussion of how they are used to derive new nouns from other nominal roots or verbal roots. The chapter also presented a discussion of pronouns and other pro-forms. It was shown that it is possible to identify five pro-forms for the language: personal pronouns, demonstrative pronouns, and interrogative pronouns. Each nominal morpheme was also described. Two lexical morphemes were identified: the diminutive morpheme $-\alpha ga$ and the augmentative morpheme -guro. There are six grammatical morphemes that attach to nouns. These were discussed in the order in which they appear when suffixed to the root (or stem). The chapter ends with a description of the structure of noun phrases. It was shown that noun phrases can be divided in two groups according to the types of modifiers they take. One is the simple noun phrase, formed by a pronoun or noun plus simple modifier (demonstrative and numeral) and the other is complex noun phrase, formed by possessive (or genitive modifiers) or relative clauses.

Chapter 5 is the last chapter. It outlined the main characteristics of the morphology of Desano verbs. It described three main classes of verbs: stative, nonstative verbs and auxiliary verbs. The class of stative verbs is composed of copula verbs, verbs coding non-existence, possessive verbs, verbs indicating location/position and 'adjectival' stative verbs. The class of non-stative verbs is composed of active verbs, verbs of motion, verbs of placement and verbs of perception and mental processes. Three auxiliary verbs were identified: *i* 'do', *wa'a* 'go' and *ãrī* 'be'. It was shown that serial verb construction is a productive morphological process. Dependent roots in these constructions function to code 'adverbial' notions of manner, aspect and modality. The grammatical morphemes that occur in the verb were also described, focusing on the description of the highly complex evidential system and other markers of clause modality. The personal ending morphemes, which occur in the last slot of the verb template, were described as markers of subject-verb agreement.

The present description is an on-going project. Although many of the most interesting typological traits have been described here, there are many aspects of the grammar are currently under investigation. For example, the description of sentence and argument structures, and some discourse-pragmatic considerations. The language has much to contribute to linguistics. Linguistic investigations of Tukanoan languages have been important for linguistic typology. Finally, it is important to mention that this work is a result of a collaborative language documentation project is the product of a language documentation project that also aims to provide material to aid in the Desano educational projects.

APPENDIX A

SURVEY OF NUMBER OF SPEAKERS AND LANGUAGES

SPOKEN IN THE DESANO COMMUNITIES

Abbreviations: BAR Barasano

DES	Desano
KUB	Kubeo
HUP	Hupda

MAKMakunaPORPortugueseSPASpanishTARTariano

TUKTukanoTUYTuyucaWANWananoWERWerekena

Community: Sa	ao Luís, Tiquié F	River					
Ethinic group	No. of people	Languages spoken by the individuals					
MALES		DES	HUP	POR	TUK	TAR	WAN
Desano	2	~		~	~		
	2			✓	✓		
Hupda	3		 ✓ 	✓	✓		
	4		 ✓ 		 ✓ 		
FEM	ALES						
Tukano	1			 ✓ 	 ✓ 		
Piratapuyo	1			 ✓ 	 ✓ 		
Tariana	1			 ✓ 	 ✓ 	~	1
Hupda	2		 ✓ 		~		
	4		1	1	1		

Community: Ur	ubú Lago, Tiquié	River				
Ethinic group No. of people Languages spoken by the individual						
MA	LES	DES	HUP	POR	TUK	
Desano	6	1		~	1	
Karapanã	2			~	1	
FEM	ALES					
Desano	2	~		~	~	
Miriti-Tapuya	4				~	
Tukano	1			~	1	
Hupda	1		1		~	

Community: Sa	anta Rosa, Tiquie	é River					
Ethinic group	No. of people	Languages	s spoken l	by the in	dividual	S	
МА	LES	DES	HUP	POR	TUK	TAR	WAN
Desano	8	✓		✓	✓		
Tukano	1	✓		~	 ✓ 		
Hupda	4		✓		 ✓ 		
	1		✓	~	 ✓ 		
FEM	FEMALES						
Desano	1	1			 ✓ 		
Tukano	4				 ✓ 		
	2			~	 ✓ 		
Tariana	1			~	 ✓ 	~	~
Hupda	2		 ✓ 		✓		
	4		~	1	1		

Community: Sa	anto Antônio, Ti	quié River						
Ethinic group	No. of people	Languages spoken by the individuals						
MALES		DES	POR	TUK	TUY	SPA		
Desano	3	~	✓	✓		~		
	3	~	✓	√				
	2		✓	 ✓ 				
Tukano	1		✓	~				
FEM	ALES							
Tukano	4		✓	~				
	1		1	1	1			
	1		1	1				

Community: Piracema/ Umari River /Tiquié River								
Ethinic group	No. of	Languag	Languages spoken by the individuals					
	people							
MALE	S	DES	HUP	POR	TUK	TUY	SPA	
Desano	10	✓		√	~			
FEMALES								
Desano	2	 ✓ 		√	√			
Tukano	1	✓		√	√			
	2			√	√		4.1	
Тиуиса	1	~		~	~	~	~	
Hupda	2		~		~			

Community: Sa	ão João Batista, Tiqu	ié River					
Ethinic group	number of people	Languages spoken by the individuals					
MALES		DES	HUP	POR	TUK		
Desano	6	~		×	1		
Tukano	5			~	1		
FE	MALES			II			
Desano	1	~		×	1		
Tukano	1	~		~	1		
	7			~	✓		
Tariana	1			1	~		

Community: Tucandira, Umari River / Tiquié River							
Ethinic group	number of people	Langua	Languages spoken by the individuals				
N	IALES	DES	HUP	POR	TUK	TUY	
Desano	3	✓ ✓ ✓ ✓					
FE	MALES						
Tukano	1	~		~	~		
Тиуиса	1	~		~	~	~	

Ethinic	No. of	Lang	uages	spoker	by the	e indivi	duals				
group	people			opener							
MAL	ES	BAR	DES	HUP	KUB	MAK	POR	TAR	TUK	TUY	WER
Desano	9		\checkmark				✓		✓		
	3		\checkmark				\checkmark		✓		
	2	24	√				✓		✓	✓	
	1	~	\checkmark			√	✓		✓	✓	
	1		\checkmark		100	~	\checkmark		✓	✓	
	1		\checkmark		~	~				0	~
	1		\checkmark		~		\checkmark		✓	1.	
	1		\checkmark			0	\checkmark		✓	1	~
Tukano	2		\checkmark				✓		✓		
	8						✓		✓		
Kubeo	1						✓		√		
Karapanã	1		1.1				✓		✓	1.	
Tuyuca	1		~				1		1	1	
FEMA	LES						1.1	0.00			
Desano	5		1.1				~		~		
	4		\checkmark				~		1		10.1
	1		~				~		1		~
Tukano	4						~		~		
	1		~				~		~		
	1		1				1		1	1	
Makuna	1					~			1		
	1		1.5				1		1	< ?	
Tuyuca	1		1						1	1	
2	1		~				1		1	1	
	1						~		1	1	
	1				14				~	1	
Kubeo	1				~		~		~	88	
	1				~		~				
Miriti-	1						~		~		
Tapuya											
Tariana	1							~			
Hupda	2			1					1		

APPENDIX B

TEXT 1 - THE HUNTER AND THE MONKEY

1 yuhunũ yu'u yẽhku mũrõ Kãdĩ mũrũ wa'ipuriwa'abu

yuu-~dʉ yʉu ~yekʉ ~bʉdʉ kadi ~bʉdʉ one-CLS:day 1SG grandfather deceased Candido deceased wai-pudi-waa-bʉ

hunt-blow-go-NON3.PERF

'One day, my deceased grandfather Candido went hunting (with a blowgun).'

2 ĩgũ nũhkũge ĩgũ waakũ ôã sẽrã mãhãrã ãrīmã ã'rīmĩ

~igʉ	~dugu-ge	~igu	waa-~k u	~ o-a	~seda
3sg.m	forest-LOC	3sg.m	go-when	DEIC:PROX-PL.AN	woolly.monkey
~baa-~	da	~adi-~ba	a∼adi-~bi		
many-	PL.AN	be-3PL.II	MPERF	say-3sg.m.imperf	

'When he went into the forest, he used to see many woolly monkeys, he said.'

3 õãrē purira ãrīdurinīgū yāābu ã'rīmī

~o-a-de	pudi-da	~adi-dudi-~di-~gʉ	~yaa-bʉ
DEIC:PROX-PL.AN-REF	blow-per	F be-hide- IMPERF - 3SG.M	see-NON3.PERF

 $\sim adi-\sim bi$

say-3SG.M.IMPERF

'He used to stay put (hiding) to watch them and catch them, he said.'

4 mühtaraga arīma a'rīmī

~buta-~da-~ga ~adi-~ba ~adi-~bi be.small-PL:AN-DIM be-3PL.AN.IMPERF say-3SG.M.IMPERF 'The monkeys were very small, he said.'

5 mühtaraga tariamürima a'rimi

~buta-~da-~gatadi-a~budi-~ba~adi-~bibe.small-PL.AN-DIMtransmit-PERFclimb-3PL.AN.IMPERFsay-3SG.M.IMPERFThe little monkeys jumped around (from branch to branch), he said.'

6 ērā dehko ārīmī ērā pagu wuagu ā'rīmī

~eda deko ~adi-~bi ~eda paa-gʉ wʉa-gʉ 3PL half be-3SG.M.IMPERF 3PL genitor-3SG.M be.big-3SG.M ~adi-~bi

say-3SG.M.IMPERF

'Among then, there was the father, who was big, he said.'

7 ĩgũ ero paagãrîkũ ĩã ĩgũrẽ mĩũbeobu ã'rĩmĩ tsu tsu ibu ã'rĩmĩ

~igʉ	edo	paa-~gad	i-g u	i-a	~igu-de
3sg.m	DEIC:DIST	belly-adv	vance-3SG.M	do-PL.AN	3SG.M-REF
~biu-b	eo-bʉ	\sim	adi-~bi	i-b u	~adi-~bi
breast.	feed-CEL-NON3	.PERF Sa	ay-3sg.m.imp	erf do- non3.p	ERF say-3SG.M.IMPERF
'When	the monkey ca	ime crawli	ing there, he	(my dad) said <i>t</i>	<i>chu tchu</i> , he said.'

8 yāmē dihiumī yu'ure ā'rīmī

~yaa-be-diu-~bi yuu-de ~adi-~bi see-CEL-downwards-3SG.IMPERF 1SG-REF say-3SG.M.IMPERF 'The monkey stared at him, he said.'

9 ĩgũ yãmẽ dihiukũ yããwããrõ purita iribu ã'rīmĩ

~ig u ~yaa-be-	diu-~kʉ	~yaa-~wa-do	pudi-ta			
3SG.M see-CEL-	downwards-when	see-go-NON3.IMPERF	blow-EMPH			
idi-b u	~adi-~	bi				
DEM:PROX-CLS.	concave say-3s	say-3SG.M.IMPERF				
He said that when he say the monkey staring,'						

10 Îgü gâmîrôgã arônôhôgã ĩgũrẽ purimẽ siugura arīribu a'rīmĩ

~igu ~gabi-do-~ga	l	~ado-do-ga	~ig u -de
3SG.M ear-CLS:body.part-DIM		DEIC:PROX-CLS:body.part	3sg.m-ref
pudi-~be-siu-gʉda	~adi-di-bu	~adi-~bi	

blow-CEL-pierce-hit be-IMPERF-NON3.PERF say-3SG.M.IMPERF 'He said that when he blew the blowgun, he wanted to hit right on the monkey's little ear.'

11 be'e eropa ãrīgū īgū nīmāmērā purikū īgū yuridihia wa'amī ã'rīmī

bee	edopa	~adi-~gʉ	~igu	~diba=beda		pudi-~kʉ
by.the.way	thus	say-3SG.M	3sg.m	poison=com/I	NSTR	blow-when
~ig u yudi-di	ii-a		waa-~l	bi	~adi-~	bi
3SG.M go.dow	vn-desc	end-PERF	go-380	.M.IMPERF	say-3s	G.M.IMPERF
'By the way, h	ne was t	blowing the blo	w gun v	with poison and	l the mo	onkey fell off,
he said.'						

12 ĩgũ yuridihia yã'ã iri buhuwahkare ãĩ ãĩkarabu ãrĩmĩ

~igʉ	yudi-dii-a	~yaa	idi	buu-waka-de
3sg.m	go.down-descend-PERF	see	DEM:PROX	***-skewer-REF
~ai	~ai-kada-bʉ	~adi-~	bi	
take	take-loosen-NON3.PERF	say-3s	G.M.IMPERF	
'He sai	d the monkey fell down and h	e went	to take the skew	wers that hit the
monke	y.'			

13 māhārõ nīmāpū wuaro waabiro ārību īgūrē ā'rīmī

~baa-do	~diba-pʉ	w u a-do	waa-bi-do
many-NOM.ABSTR	poison-CONT	TR be.big-NOM.ABSTR	go-NEG-NON3.PERF

~adi-bu ~igu-de	~adi-~bi
-----------------	----------

be-non3.PERF 3SG.M-REF say-3SG.M.IMPERF

'However, the poison wasn't enough (to kill) him (the monkey).'

- eropi îgũre seire ãi ãiômãbu ã'rimi
 edopi ~igu-de sei-de ~ai ~ai-~oba-bu
 thus 3sG.M-REF monkey-REF take take-run-NON3.IMPERF
 ~adi-~bi
 say-3sG.M.IMPERF
 Thus, he took the monkey and put it on his back.'
- 15 ãĩômãã piariariribu ã'rīmĩ

16

~ai~oba-a	pia-di-adi-	di-b u				
take-wrap-PERF	MOV:out.in	to-DIR	:towards-c	ome-DIR	towards-NON3.IM	PERF
~adi-~bi						
say-3SG.M.IMPERF						
'Putting the monkey	on his back,	he retu	urned, he s	aid.'		
yu'u piarikũ ĩgũ yếĩsi	ru mãhsã wa	aari ãri	ັ ຫາັ aິ'ຕັຫາ ັ ງ	pare		
y uu pia-di-~k u	~ig	u ∼y	vei-sidu ¹⁰³		~basa-waa-di	
1sg MOV:out.into-	when 3so	.M be	ugly-dece	ased	know-go-IMPERF	,
~adi-~bi	~adi-~bi		pade			

¹⁰³ siru is a Tukano word that means 'deceased,' the word in Desano is buru.

'He said that when he was returning, the monkey came back to life.'

mähsäpeemähsituaha yuure äröyöra puibugora yuure kürikärämi ä'rimi
 ~basa-pee-~masi-tua-da yuu-de ado-goda

~basa-pee-~masi-tua-da yuu-de ado-goda
know-listen-know-lean-PERF 1sG-REF DEIC-CLS:shell
pui-bu-goda

basket-CLS:concave-CLS:shell

'He said that he noticed the monkey was moving on his back, then the monkey bit him on his buttocks.'

- 18 agua mũũ yếisiru yuure kũrikohoaka ãri igũrẽ mẽpikābu ã'rimi ~buu ~yei ~budu yuu-de ~kudi-koa-ka agua ~adi ouch 2sg be.ugly deceased 1sg-REF bite-throw.away-EXIST be ~igu-de ~bepi-ka-bu ~adi-~bi throw.away-EXIST-NON3.PERFsay-3SG.M.IMPERF 3SG.M-REF "Ouch! You bit me damn monkey!" He said throwing the monkey from his back.'
- 19 ĩgũ mẽhpĩkũ ĩgũpũ paamũrĩa waamĩ a'rīmĩ

~igu ~bepi-~ku ~igu-pu paa-~budi-a wa-~bi 3SG.M throw.away-when 3SG.M-CONTR hit-climb-PERF go-3SG.M ~adi-~bi 'He said that when he threw the monkey, the monkey climbed on the trees.'

- 20 yë'ëdaro ika musiru îgurë ari îgurë yuhkugu ta'abe îgurë pabu ~yee-da-do i-ka ~buu ~budu ~igu-de ~adi be.bad-PERF-NOM.ASBTR do-exist 2SG deceased 3sg.M-REF be ~igu-de yuku-gu taa-be ~igu-de paa-bu 3SG.M-REF tree-CLS:trunk cut-CEL 3SG.M-REF hit-NON3.PERF "You did bad," the deceased (man) told the monkey; "I got a piece of wood and hit him," he said.'
- 21 ĩgũ dihpuru pawabu ĩgũre pare ã'rĩmĩ

~igu dipu-du paa-wa-bu ~igu-de pade ~adi-~bi
3SG.M head-CLS:concave hit-go-NON3.PERF 3SG.M-REF then say-3SG.M.IMPERF
'He said that he hit the monkey on the head.'

22 yée arima mariwaimura baaduaku namékama irasa ari werenimi

~yee-~adi-~ba	~badi	wai-~bʉdamũ	dã	baa-dʉa-~kʉ
be.bad-be-3PL.AN.IMPERF	1pl	fish-deceased		eat-DES-when
~gabe -~ka-~ba	i-~da-	~sa	~adi	wede-~adi-~bi
want-PREDIC-3PL.AN.IMPERF	DEM:PI	ROX-PL.AN-ADI	be	tell-say-3SG.M.IMPERF
'The animals are evil when w	ve try to	eat them; the a	nimals	also get revenge, he
narrated.				

APPENDIX C

TEXT 2 - COLLECTING MATERIALS TO MAKE ORNAMENTS

1 ēropīrā mūrā māmāro gere ūrīnūrē kurisuayumā

~edopida ~bʉ-~da ~baba-do gede ~ʉdi-~dʉ-de thus elder-PL.AN be.new-NOM.ASBSTR season pupunha.fruit-CLS:day-REF kudi-sʉa-~yu-~ba walk-like-EVID:QUOT/FOLK-3PL.AN.IMPERF

'Thus, the elders like to go walking during the pupunha fruit season.'

2 waitu ãĩrã nũhpĩ põrã yẽãrã karamũhũ põrã omõãborebu põrã yẽãrã ~ai-~da waitu ~dupi ~po-~da ~yea-~da wild.fruit take-NON3.PERF mutum.bird progenee-PL.AN hold-3PL.PERF kada-bu ~po-~da 0 ~boa-bode-bu jacu.bird-CLS.concave progenee-PL.AN salt-CLS.buddle-CLS.concave or ~po-~da ~yea-~da progenee-PL.AN hold-3PL.PERF

'(They) caught *waitu* fruit, caught babies of mutum birds, babies of jacu birds end babies of jacamim birds.

3 ērā māhsūduarā īyūrā ērārē

~eda ~basu-dua-~da i-~yu-~da ~eda-de 3PL tame-DES-PL.AN.PERF do-EVID:QUOT/FOLK 3PL-REF 'They want to tame them.'

4 yẽããĩgãn wi'ige eho mãhsũ opayũmã

~yea-~ai-~gadi wii-ge

comprise-take-advance house-LOC

eo-~basu-opa-~yu-~ba

 $feed-tame-have-{\tt EVID.QUOT/FOLK-3PL.AN.IMPERF}$

'They took the caught the birds and brought them home to raise them.'

5 goe ũmũgũge waayũmã daha ũmũgũge mũrĩã ũmũ porare ũmũ surire taayũmã ~ubu-gu-ge waa-~yu-~ba daa goe return japu.bird-CLS.trunk-LOC go-EVID:QUOT/FOLK again ~ubu-gu-ge ~budi-a ~ubu ~po-~da-de climb-PERF japu.bird progenee-PL.AN-REF japu.bird-cls.trunk-LOC ~ubu sudi-de taa-~yu-~ba japu.bird nest-REF cut-EVID.QUOT/FOLK-3PL.AN.IMPERF 'They returned again looking for trees where the japu birds breed to collect the nests with the baby birds.'

7 taaãīdihari ehomāhsū īyūmā

taa-~ai-dia-di eo~basu i-~yu-~ba cut-take-descend-IMPERF feed-tame do-EVID:QUOT/FOLK-3PL.AN.IMPERF 'They cut off the nests, took the baby birds, climbed down and raised the baby birds.'

- 8 ērā peyari beruri īmõrā ārīrā
 ~eda peya-di bedu-di ~ibo-~da ~adi-~da
 3PL headdress-PL.IN circle-PL.IN obtain-PL.AN.PERF be-PL.AN.PERF
 'They (the elders) caught them (the birds) to make headdresses.'
- 9 ērā mērā ērā poari mērā āī ērāpeyari berori mõmēyūmā
 ~eda=~beda ~eda poa-di=~beda ~ai ~eda peya-di
 3PL=COM/INSTR 3PL hair-PL.IN=COM/INSTR take 3PL be.on.top.of-NOM
 bedo-di ~bobe-~yu-~ba
 circle-PL.IN work-EVID:QUOT/FOLK-3PL.AN.IMPERF
 'With these birds, they took their feathers and worked on the headdresses.'
- mãhã poarire mãhã poari gahsiro waya
 ~baa poa-di-de ~baa poa-di gasi-do waa-ya
 macaw hair-PL.IN-REF macaw hair-PL.IN bark-CLS:body.part go-EVID:RES
 'The macaw feathers formed the headdresses of macaw feather.'

11	gahkia dopa ārīrā āīyūmā daha i poadari gahki poadari īmõrā ārīrā					
	gaki-a	dopa	~adi-~da	~ai-~	-yu-~ba	daa
	monkey-P	L.AN like	be-PL.AN.F	PERF take-	evid:quot/folk -3 pl.imperf	again
	i	poa-da-di		gaki	poa-da-di	
	DEM:PROX	hair-CLS:th	read-PL.IN	monkey	hair-CLS:thread-PL.IN	
	~ibo-~da		~adi-~da	ì		
	obtain-PL.	AN.PERF	be-PL.AN	J.PERF		

'Again, they took any monkey to make headdresses of monkey's hair.'

REFERENCES

AIKHENVALD, ALEXANDRA Y. 1996. Areal diffusion in North-West Amazonia: the case of Tariana. *Anthropological Linguistics* 38.73-116.

AIKHENVALD, ALEXANDRA Y. 1999. Areal diffusion and language contact in the Içana-Vaupés basin, north-west Amazonia. *The Amazonian Languages*, ed. by Alexandra Y. Aikhenvald and R. M. W. Dixon, 385-416. Cambridge: Cambridge University Press.

AIKHENVALD, ALEXANDRA Y. 2000. *Classifiers: A Typology of Noun Categorization Devices*. Oxford: Oxford University Press.

AIKHENVALD, ALEXANDRA Y. 2001. Areal Diffusion, genetic inheritance, and problems of subgrouping: a North Arawak case study. *Areal Diffusion and Genetic Inheritance*, ed. by Alexandra Y. Aikhenvald and R. M. W Dixon, 167-94. Oxford: Oxford University Press.

AIKHENVALD, ALEXANDRA Y. 2002a. *Language Contact in Amazonia*. New York: Oxford University Press.

AIKHENVALD, ALEXANDRA Y. 2002b. Typological parameters for the study of clitics, with special reference to Tariana. *Word: A Cross-Linguistic Typology*, ed. by R. M. W. Dixon and Alexandra Y. Aikhenvald, 42-78. Cambridge: Cambridge University Press.

AIKHENVALD, ALEXANDRA Y. 2003a. A Grammar of Tariana. Cambridge Grammatical Descriptions. Cambridge: Cambridge University Press.

AIKHENVALD, ALEXANDRA Y. 2003b. Mechanisms of change in areal diffusion: new morphology and language contact. *Journal of Linguistics* 39.1-29.

AIKHENVALD, ALEXANDRA Y. 2004. Evidentiality. Oxford University Press.

AIKHENVALD, ALEXANDRA Y. 2006. Serial Verb Constructions in Typological Perspective. *Serial Verb Constructions: A Cross-linguistic Typology*, ed. by Alexandra Y. Aikhenvald, 1-68. Oxford University Press.

ALEMÁN M., TULIO; Reinaldo López H.; and Marion Miller. 2000. *Wirã ya, peamasa ya wererituri* (Desano-Español) (Diccionário bilingüe de 896 palabras). Bogota: Editorial Alberto Lleras Camargo.

ANDERSON, LLOYD. 1986. Evidentials, Paths of Change, and Mental Maps: Typologically Regular Asymmetries. *Evidentiality: The Linguistic Coding of Epistemology*, ed. by Wallace Chafe and Johanna Nichols, 273-312. Norwood, NJ: Ablex.

ARDILA, OLGA. 1989. Diversidad lingüística y multilingüismo en los grupos tucano del Vaupés. *Forma y Función 4*, Bogotá: Universidad Nacional de Colombia.

ARDILA, OLGA. 1993. La subfamilia lingüística tucano-oriental: estado actual y perspectivas de investigación. *Biblioteca Uricoechea* 11.219-233. Bogotá: Instituto Caro y Cuervo.

ARDILA, OLGA. 2000. Resena Bibliografica del Desano. *Lenguas Indigenas de Colombia: Una Vision Descriptiva*, ed. by M. S. Gonzales de Perez and M. L. Rodrigues Montes, 489-91. Santafe de Bogota: Instituto Cairo y Cuervo.

ÅRHEM, KAJ. 1981. Makuna social organization. A study in descent, alliance and the formation of corporate group in the North-Western Amazon. *Acta Universitatis Upsaliensis, Uppsala Studies in Cultural Anthropology 4*. Stockholm: Almqvist & Wiksell.

ÅRHEM, KAJ. 1989. The Makú, the Makuna and the Guiana systems: transformations of social structure in northern Lowland South America. *Ethnos* 54:5-22.

ÅRHEM, KAJ. 1996. The cosmic food web: human-nature relatedness in the Northwest Amazon. Nature and society: Anthropological perspectives, ed. by Philippe Descola and Gísli Pálsson, 185-204. London: Routledge.

BARNES, JANET. 1996. Autosegments with three-way lexical contrasts in Tuyuca. *International Journal of American Linguistics* 62.31-58.

BARNES, JANET. 1999. Tucano. *The Amazonian Languages*, ed. by R. M. W. Dixon and Alexandra Y. Aikhenvald, 207-26. Cambridge: Cambridge University Press.

BARNES, JANET. 2006. Tucanoan Languages. *Encyclopedia of Language and Linguistics*, ed. by Keith Brown, 130-42. Second edition, vol. 13. Oxford: Elsevier.

BEKSTA, KAZYS JURGIS. 1988. *A Maloca Tukano-Dessana e seu Simbolismo*. Manaus: Secretaria de Educação e Cultura, SEDUC.

BECKMAN, MARY E. 1986. Stress, and Non-Stress Accent. *Netherlands Phonetic Archives* 7. Dordrecht: Foris.

BICKEL, BALTHASAR and JOHANNA NICHOLS. 2007. Inflectional morphology. *Language Typology and Syntactic Description*, ed. by Timothy Shopen, 169-239. Second edition, Volume III (Grammatical Categories and the Lexicon). Cambridge: Cambridge University Press.

BIVIN, WILLIAM E. 1986. *The Nasal Harmonies of Twelve South American Languages*. The University of Texas at Arlington: M.A. thesis.

BLEVINS, JULIETTE. 1995. The syllable in phonological theory. *The Handbook of Phonological Theory*, ed. by John Goldsmith, 206-44. Oxford: Blackwell.

BOAS, FRANZ. 1911. Kwakiutl. *Handbook of American Indian Languages*, ed. by Franz Boas, 423-557. Washington: Government Printing Office.

BRUZZI ALVES DA SILVA, A. 1977. *A Civilização Indígena do Uaupés*. Second edition. São Paulo: Missão Salesiana do Rio Negro.

BUCHILLET, DOMINIQUE. 1983. *Maladie et Memoire des Origines chez les Desana du Vaupes: Conceptions de la Maladie et de la Therapeutique d'une Societe Amazonienne*. Universidade de Paris X, Nanterre: PhD Dissertation.

BUCHILLET, DOMINIQUE. 1987. 'personne n'est la pour ecouter': Les Conditions de Mise en Forme des Incantations Therapeutiques chez Desana du Uaupes Brasilien. *Ameridia* 12.7-32.

BUCHILLET, DOMINIQUE. 1988. Interpretação da doença e simbolismo ecológico entre os indíos Desana. *Boletim do Museu Paraense Emilio Goeldi, Serie Antropologia* 4, volume 1.27-42.

BUCHILLET, DOMINIQUE. 1990a. Los Poderes del Hablar: Terapia y aggression chamanica entre los indios Desana del Vaupes brasileiro. *Las Culturas Nativas Latinoamericanas a Trave de su Discurso*, ed. by Ellen Basso and Joel Shezner, 319-54. Quito: Abya-Yala.

BUCHILLET, DOMINIQUE. 1990b. Os Índios da Região do Alto Rio Negro. História, etnografia e situação das terras. *Laudo antropológico remetido à Procuradoria Geral da República*. Brasília, Ms.

BUCHILLET, DOMINIQUE. 1992. Nobody is there to hear: Desana therapeutic incantations. *Portals of Power*, ed. by E. Jean M. Langdon and G. Baer, 211-30. Albuquerque: University of New Mexico.

BUCHILLET, DOMINIQUE. 1995a. Perles de verre, parules de blancs et 'pots de paludisme': Epidemiolgie et representations de maladies infectueuses des Desana (region du haut Rio Negro, Bresil). *Journal de la Societe des Americanistes de Paris*, 81.181-206.

BUCHILLET, DOMINIQUE. 1995b. Contas de vidro, enfeites de branco e 'potes de malaria': epidemiologia e representacoes de doencas infecciosas entre os Desana. *Serie Antropologia*, 187. Brasilia: Universidade Nacional de Brasilia.

BYBEE, JOAN. 1985. *Morphology: A Study of the Relation Between Meaning and Form.* Amsterdam: John Benjamins.

BYBEE, JOAN, REVERE PERKINS and WILLIAM PAGLIUCA (eds.). 1994. *The Evolution of Grammar: Tense, Aspect and Modality in the Languages of the World.* Chicago: University of Chicago Press.

CAMPBELL, LYLE. 1997. *American Indian Languages: The Historical Linguistics of Native America*. Oxford: Oxford University Press.

CAMPBELL, LYLE. 2012a. Classification of the Indigenous Languages of South America. *The Indigenous Languages of South America: A Comprehensive Guide*, ed. by Lyle Campbell and Verónica Grondona, 59-166. (<u>The World of Linguistics, vol. 2.</u>) Berlin: Mouton de Gruyter.

CAMPBELL, LYLE. 2012b. Typological characteristics of South American indigenous languages. *The Indigenous Languages of South America: A Comprehensive Guide*, ed. by Lyle Campbell and Verónica Grondona, 259-330. (<u>The World of Linguistics, vol. 2.</u>) Berlin: Mouton de Gruyter.

CASTILHO, ALVARO MAIA et al. 2002. Umuko Mah'sã, Ya Boeri. Manaus: SEDUC-AM.

CHERNELA, JANET M. 1982. Estrutura social do Uaupés brasileiro. *Anuário Antropológico* 81.59-69. Rio de Janeiro.

CHERNELA, JANET. 1989. Marriage, Language, and History Among Eastern Tukanoan Speaking Peoples of the Northwest Amazon. *The Latin American Anthropology Review*, 1(2):36-42.

CHERNELA, JANET. 1993. *The Wanano Indians of the Brazilian Amazon: A Sense of Space*. Austin: University of Texas Press.

CHERNELA, JANET. 2004. The Politics of Language Acquisition: Language Learning as Social Modeling in the Northwest Amazon. *Women and Language* 27(1):13-20.

CHOMSKY, NOAM. 1965. Aspects of the Theory of Syntax. Cambridge: MIT Press.

CHOMSKY, NOAM and MORRIS HALLE. 1968. *The Sound Pattern of English*. New York: Harper and Row.

CORBETT, GREVILLE G. 2007. Gender and noun classes. *Language Typology and Syntactic Description*, ed. by Timothy Shopen, 241-279. Second edition, Volume III (Grammatical Categories and the Lexicon). Cambridge: Cambridge University Press.

CONSELHO INDIGENISTA MISSIONÁRIO (CIMI) <u>http://www.cimi.org.br/site/pt-br/</u> (date visitied: August 3, 2011).

CORREA RUBIO, FRANÇOIS. 1984. Amazonía colombiana: organización social en el noroeste del Amazonas. *Revista Colombiana de Antropología* 25.183-208.

CORREA RUBIO, FRANÇOIS. 1997. Identidad social y ejercicio lingüístico en la región del Vaupés colombiano. *Lenguas Amerindias: Condiciones Sociolingüísticas en Colombia*, ed. by Ximena Pachón and François Correa, 443-92. Bogotá: Instituto Lingüístico de Verano.

CRAIG, COLETTE. 1992. Classifiers in a functional perspective. Layered Structure and Reference in a Functional Perspective: Papers from the Functional Grammar Conference in Copenhagen, 1990, ed. by Michael Fortescue, Peter Harder and Lars Kristoffersen, 277-301. Amsterdam: John Benjamins.

CROFT, WILLIAM. 2003. *Typology and Universals*. Cambridge: Cambridge University Press.

DA SILVA, ALCIONILIO BRUZI ALVES. 1961. *Ethno-Linguístico-Musical das Tribos dos Rios Vaupés, Içana e Cavaburi.* Sao Paulo.

DE HAAN, FERDINAND. 1996. The Category of Evidentiality. MS.

DE HAAN, FERDINAND. 1999. Evidentiality and Epistemic Modality: Setting Boundaries. *Southwest Journal of Linguistics* 18.83-101.

DIXON, R. M. W. 1977. Where Have All the Adjectives Gone? *Studies in Language* 1.1980. [Reprinted as pp. 1-62 of Dixon 1982].

DIXON, R. M. W. 1982. Where Have All the Adjectives Gone? And Other Essays in Semantics and Syntax. Berlin: Mouton Publishers.

DIXON, R. M. W. 2006. Adjective classes in typological perspective. *Adjective Classes: A Cross-Linguistic Typology*, ed. by R. M. W. Dixon and A. Y. Aikhenvald, 1-49. Cambridge: Cambridge University Press. pp. 1-49.

DIXON, R. M. W. and ALEXANDRA Y. AIKHENVALD. 1999. *The Amazonian Languages*. Cambridge: Cambridge University Press.

DIXON, R. M. W. and ALEXANDRA Y. AIKHENVALD. 2002. *Word: A Cross-Linguistic Typology*. Cambridge: Cambridge University Press.

DOBROVOLSKY, MICHAEL and EWA CZAYKOWSKA-HIGGINS. 2001. Phonology: The Function and Patterning of Sounds. *Contemporary Linguistics: An Introduction*, ed. by William O'Grady, John Archibald, Mark Aronoff, and Janie Rees-Miller, 63-130. Fourth edition. Boston: Bedford/St.Martin's.

DRYER, MATTHEW S. 2007. Noun phrase structure. *Language Typology and Syntactic Description*, ed. by Timothy Shopen, 151-205. Second edition, volume II (Complex Constructions). Cambridge: Cambridge University Press.

EPPS, PATIENCE. 2005. Areal diffusion and the development of evidentiality: Evidence from Hup. *Studies in Language* 29.617-50.

EPPS, PATIENCE. 2007. The Vaupés melting pot: Tucanoan influence on Hup. *Grammars in Contact: A Cross-Linguistic Typology*, ed. by Alexandra Aikhenvald and R. M. W. Dixon, 267-89. (Explorations in Linguistic Typology 4). Oxford: Oxford University Press.

FERNANDES, AMERICO C. and DURVALINO M. FERNANDES. 1996. *Mitologia Sagrada dos Desana-Wari Dihputiro Porã*. Coleção Narradores Indigenas do Rio Negro. São Gabriel da Cachoeira: UNIRT/FOIRN.

FOIRN [Federação das Organizações Indígenas do Rio Negro] and ISA [Instituto Socioambiental]. 2000. *Povos Indígenas do Alto e Médio Rio Negro: Uma Introdução à Diversidade Cultural e Ambiental do Noroeste da Amazônia Brasileira*. São Paulo: Instituto Socioambiental/Federação das Organizações Indígenas do Rio Negro/Ministério da Educação.

FRANCHETTO, BRUNA and ELSA GOMEZ-IMBERT. 2003. Review of 'The Amazonian Languages,' ed. by R. M. W. Dixon and Alexandra Y. Aikhenvald. *International Journal of American Linguistics* 69.232-8.

GIANCONNE, PE. ANTONIO. 1949. Os Tucanos e Outras Tribos do Rio Vaupés, Afluente do Rio Negro-Amazonas. Sao Paulo.

GILDEA, SPIKE. 2012. Linguistic Studies in the Cariban Family. *The Indigenous Languages of South America: A Comprehensive Guide*, ed. by Lyle Campbell and Verónica Grondona, 441-94. (The World of Linguistics, vol. 2), Berlin: Mouton de Gruyter.

GIVÓN, TALMY. 2001. Syntax: An Introduction, Volume I. John Benjamins Publishing Company. Amsterdam/Philadelphia.

GOLDMAN, IRVING. 1948. Tribes of the Vaupés-Caquetá region. *Handbook of South American Indians*, ed. by Julian H. Stewart, 763-98. (The tropical tribes, vol. 3). Washington: Bureau of American Ethnology.

GOMEZ-IMBERT, ELSA. 1982. De la Forme et du Sens Dans la Classification Nominale en Tatuyo (Langue Tukano Orientale d'AmazonieColombienne). Doctorat de 3^e cycle. Paris-Sorbonne: PhD dissertation.

GOMEZ-IMBERT, ELSA. 1991. Force des langues vernaculaires en situation d'exogamie linguistique: le cas du Vaupés colombien. *Cahier des Sciences Humaines* 27.535- 59.

GOMEZ-IMBERT, ELSA. 1993. Problemas en torno a la comparación de las lenguas tucano orientales. *Estado Actual de la Clasificación de las Lenguas Indígenas de Colombia*, ed. by María Luisa Rodríguez de Montes, 235-67. Bogotá: Instituto Caro y Cuervo.

GOMEZ-IMBERT, ELSA. 1996. When animals become "rounded" and "feminine:" conceptual categories and linguistic classification in a multilingual setting. *Rethinking Linguistic Relativity*, ed. by John J. Gumperz and Stephen C. Levinson, 438-69. Cambridge: Cambridge University Press.

GOMEZ-IMBERT, ELSA. 1997. Morphologie et Phonologie Barasana: Approche Non Linéaire. Université Paris 8. Saint-Denis: PhD Dissertation.

GOMEZ-IMBERT, ELSA. 1999. Variations tonales sur fond d'exogamie linguistique. *Cahiers de Grammaire* 24.67-94. Toulouse: Université Toulouse-Le Mirail.

GOMEZ-IMBERT, ELSA. 2005. Fonología de dos idiomas tucano del Pirá-Parana: Barasana y Tatuyo. *Amerindia* 29/30.43-80.

GOMEZ-IMBERT, ELSA. 2007a. Construciones serials en tatuyo y barasana (familia tukano): hacia una tipología de la serialización verban. *Lenguas Indígenas de América del Sur: Estudios Descriptivo-Tipológiocos y Sus Contribuciones para la Lingüística Teórica*, ed. by A. Romero-Figueroa, A. Fernández Garay y A. Corbera Mori, 172-89. Caracas: Ediciones UCAB.

GOMEZ-IMBERT, ELSA. 2007b. Tukanoan nominal classification. *Language Endangerment and Endangered Languages*, ed. by Leo Wetzels, 401-28. (Indigenous Languages of Latin America 5). The Netherlands: CNWS.

GOMEZ-IMBERT, ELSA. 2004. Fonología de dos idiomas Tukano orientales. *Amerindia* 29.43-96. Paris: CELIA.

GOMEZ-IMBERT Elsa. 2011. La famille tukano. *Dictionnaire des Langues du Monde*. Paris: Presses Universitaires de France, 1454-60.

GOMEZ-IMBERT, ELSA, and MICHAEL KENSTOWICZ. 2000. Barasana Tone and Accent. *International Journal of American Linguistics* 66.419-63.

GONZÁLEZ ÑÁÑEZ, OMAR. 2004. Multilingüismo, etnias y culturas indígenas en el "Noroeste Amazónico" del Estado Amazonas. *Cuadernos Lengua y Habla* 3, ed. by Carmen Luisa Domínguez, Lourdes Pietrosemolli y Alexandra Alvarez, 360-70. (Estudios Lingüísticos en homenaje a Paola Bentivoglio). Centro de Investigación y Atención Lingüística, Mérida: Universidad de Los Andes.

GONZÁLEZ DE PÉREZ, MARÍA STELLA and MARÍA LUISA RODRÍGUEZ DE MONTES (eds.). 2000. *Lenguas indígenas de Colombia: una visión descriptiva*. Santa Fé de Bogotá: Instituto Caro y Cuervo. GUSSENHOVEN, CARLOS. 2004. *The Phonology of Tone and Intonation*. Research Surveys in Linguistics. Cambridge: Cambridge University Press.

HASPELMATH, MARTIN. 2002. Understanding Morphology. New York: Arnold Publishers.

HAYES, BRUCE. 1995. *Metrical Stress Theory: Principle and Case Studies*. The University of Chicago Press.

HUBER, RANDALL Q. and ROBERT B. REED. 1992. *Vocabulario Comparativo: Palabras Selectas de Lenguas Indigenas de Colombia*. Santa Fé de Bogotá, Colombia. Asociación Instituto Linguistico de Verano.

HUDDLESTON, RODNEY and GEOFFREY K. PULLUM. 2009. A Student's Introduction to English Grammar. Cambridge: Cambridge University Press.

HUGH-JONES, CHRISTINE. 1977. Skin and soul: the round and the straight. Social time and social space in Pira Paraná society. *Actas del 42 Congreso Internacional de Americanistas* 2.185-204. Paris.

HUGH-JONES, CHRISTINE. 1979. From the Milk River: Spatial and Temporal Processes in Northwest Amazonia. Cambridge: Cambridge University Press.

HUGH-JONES, CHRISTINE. 1988. Conceptos de espacio y tiempo entre los indígenas de Pirá-Paraná (Vaupés-Colombia). *Hombre y Ambiente: El Punto de Vista Indígena* 5.5-56. Quito.

HUGH-JONES, STEPHEN. 1993. Clear descent or ambiguous houses? A re-examination of Tukanoan social organization. *L'Homme* 95-120.

HUGH-JONES, STEPHEN.1994a. 'Food' and 'drugs' in Northwest Amazonia. *Tropical Forests, People and Food: Biocultural Interactions and Applications to Development*, ed. by C. M. Hladik, A. Hladik, O. Linares, H. Pagezy, A. Semple, and M. Headly, 533-48. (Man in the Biosphere Series, volume 13). Paris: Parthenon Press/UNESCO.

HUGH-JONES, STEPHEN. 1994b. Shamans, prophets, priests and pastors. *Shamanism, Colonialism and the State*, ed. by C. Humphrey and N. Thomas, 32-75. Ann Arbor: University of Michigan Press.

HUGH-JONES, STEPHEN. 1995. Back to front and inside out: the androgynous house in Northwest Amazonia. *About the House: Lévi-Strauss and Beyond*, ed. by J. Carsten and S. Hugh-Jones. Cambridge: Cambridge University Press.

HUGH-JONES, STEPHEN. 2002. Nomes secretos e riqueza visível: nominação no noroeste amazônico. *Mana* 8.45-68. Rio de Janeiro.

HULST, H. V. D. and N. SMITH (eds.). 1988. *Autosegmental Studies on Pitch-Accent*. Dordrecht: Foris.

HYMAN, LARRY M. 2007. How (not) to do Phonological Typology: The Case of Pitch Accent. *Paper Presented at the Seventh Biannual Meeting of the Association for Linguistics Typology (ALT)*. Paris, September 27. (handout).

HYMAN, LARRY M. 2009. The Representation of Tone. UC Berkeley Phonology Lab Annual Report. ms.

INSTITUTO SOCIOAMBIENTAL (ISA) <u>http://pib.socioambiental.org/en/c/quadro-geral</u> (date visitided: August 2011).

ITÔ, JUNKO. 1986. *Syllable Theory in Prosodic Phonology*. Doctoral dissertation, University of Massachusetts, Amherst. New York: Garland Press, 1988.

JACKSON, JEAN E. 1974. Language identity of the Colombian Vaupés Indians. *Explorations in the Ethnography of Speaking*, ed. by Richard Bauman and Joel Sherzer. London/New York: Cambridge University Press.

JACKSON, JEAN E. 1976. *Vaupés Marriage: A Network System in the Northwest Amazon*, ed. by C.A. Smith, 65-93. (Regional analysis, volume 2: Social systems). New York: Academic Press.

JACKSON, JEAN E. 1983. *The Fish People: Linguistic Exogamy and Tukanoan Identity in Northwest Amazonia*. Cambridge: Cambridge University Press.

JACKSON, JEAN E. 1984. Vaupés marriage practices. *Marriage practices in Lowland South America*, ed. by Kenneth M. Kensinger, 156-79. Urbana: University of Illinois Press.

JACKSON, JEAN E. 1988. Gender relations in the central Northwest Amazon. *Antropologica* 70.17-68. Caracas.

JACKSON, JEAN E. 1991a. Being and becoming an Indian in the Vaupés. *Nation-States and Indians in Latin-America*, ed. by Greg Urban and Joel Sherzer, 131-55. Austin: University of Texas Press.

JACKSON, JEAN E. 1995. Culture, genuine and spurious: the politics of Indianness in the Vaupes, Colombia. *American Ethnologist* 22.3-27.

JAKOBSON, ROMAN. 1957 [1971]. Shifters, verbal categories, and the Russian verb. *Roman Jakobson: Selected Writings II*, 130-47. The Hague: Mouton.

KAUFMAN, TERRENCE. 1967. *Macro-Mayan Vocabulary Survey Questionnaire*. Unpublished manuscript.

KAUFMAN, TERRENCE and BRENT BERLIN. 1987. *South American Indian Languages Documentation Project Questionnaire*. Pittsburgh/Berkeley: University of Pittsburgh and University of California at Berkeley. Ms.

KAYE, JONATHAN D. 1965. *Two Concepts of Desano Phonology*. Columbia University, New York: M.A. thesis.

KAYE, JONATHAN D. 1968. Nominalized relative clauses in Desano. *Canadian Journal of Linguistics* 14.40-57.

KAYE, JONATHAN D. 1970. The Desano Verb: Problems in Semantics, Syntax and Phonology. Columbia University: PhD dissertation.

KAYE, JONATHAN D. 1971. Nasal harmony in Desano. Linguistic Inquiry 2.37-56.

KENSTOWICZ, MICHAEL. 1994. *Phonology in Generative Grammar*. Blackwell Publishers.

KOCH-GRÜNBERG, THEODOR. 1906. Die Indianerstämme am oberen Rio Negro und Yapura und ihre sprachliche Zugehörigkeit. Zeitschrift für Ethnologie 38.166-205.

KOCH-GRÜNBERG, THEODOR. 1913. Die Betoya-Sprachen Nordwest-brasiliens und der angrezenden Gebiete. *Anthropos* 7.944-77.

KOCH-GRÜNBERG, THEODOR. 1914. Die Betoya-Sprachen Nordwest-brasiliens und der angrezenden Gebiete. *Anthropos* 9.151-95; 569-89; 812-32.

KOCH-GRÜNBERG, THEODOR. 1915. Die Betoya-Sprachen Nordwest-brasiliens und der angrezenden Gebiete. *Anthropos* 10.114-58.

KOCH-GRÜNBERG, THEODOR. 1916. Die Betoya-Sprachen Nordwest-brasiliens und der angrezenden Gebiete. *Anthropos* 11. 421-49.

KOCH-GRÜNBERG, THEODOR. 1923 [1909-10]. Zwei bei den Indianern Nordwest-Brasilens. Stuttgart: Strecker und Schroeder. (Fifth edition, original 1909-1910 Zwei Jahre under den Indianern.)

KOCH-GRÜNBERG, THEODOR. 2005 [1909]. *Dois Anos Entre os Indígenas*. Manaus: EDUA.

LADD, D. ROBERT. 1996. Intonational Phonology. Cambridge: Cambridge University Press.

LADEFOGED, PETER. 2001. A Course in Phonetics. Heinle & Heinle.

LADEFOGED, PETER. 2003. *Phonetic Data Analysis: An Introduction to Fieldwork and Instrumental Techniques.* Blackwell Publishing.

LADEFOGED, PETER and IAN MADDIESON. 1996. *The Sounds of the World's Languages*. Blackwell Publishing.

LANA, FELICIANO. 1986. Ill Ventre Dell'Universo, Sellerio Editore Palermo.

LANA, FELICIANO. 2002. *A Origem da Noite, ou Como as Mulheres Roubaram as Flautas Sagradas.* (Série Autores Indígenas). Rio de Janeiro: FUNARTE/ Editora da Universidade do Amazonas.

LANA, FIRMIANO A. and LUIS G. LANA. 1995 [1982]. Antes o Mundo Não Existia, São João Batista do Rio Tiquié/ São Gabriel da Cachoeira:UNIRT/FOIRN.

LEHISTE, ILSE. 1970. Suprasegmentals. Cambridge: MIT Press

LINDBLOM, BJÖRN. 1986. Phonetic Universals in Vowel Systems. *Experimental Phonology*, ed. by John J. Ohala and Jeri J. Jaeger, 13-44. Academic Press.

LYONS, JOHN. 1968. *Introduction to Theoretical Linguistics*. Cambridge: Cambridge University Press.

MACAULAY, MONICA and JOE SALMONS. 1995. The Phonology of Glottalization in Mixtec. *International Journal of American Linguistics* 61(1).38-61.

MALONE, TERRELL. 1987. *Proto-Tucanoan and Tucanoan Genetic Relationships*. Colombia: Instituto Lingüistico de Verano. Ms.

MEIRA, MÁRCIO. 1991. *Laudo antropológico Área Indígena Baixo Rio Negro*. Belém: Museu Paraense Emílio Goeldi.

MEIRA, MÁRCIO. 1993. O Tempo dos Patrões: Extrativismo da Piaçava entre os Índios do Rio Xié (Alto Rio Negro). Campinas: UNICAMP. M.A. thesis.

MEIRA, SERGIO and ELSA GOMEZ-IMBERT. 2005. Review of 'Language contact in Amazonia,' by Alexandra Y. Aikhenvald. *Studies in Language* 29.168-79.

MENENDEZ, LARISSA L. 2005. *Iconografias do Invisível: A Arte de Feliciano e Luis Lana*. Dissertação de Mestrado do programa de pós-graduação em Ciências Sociais da Pontificia Universidade Católica de São Paulo.

MENENDEZ, LARISSA L. 2006. Dois artistas, um mito. *Artigo Apresentado no II Encontro de Historia da Arte*, IFCH-Unicamp, 27 a 29 de Marco de 2006, Campinas, SP.

MENENDEZ, LARISSA L. 2009. *Iconografias do Invisível: A Arte de Feliciano e Luis Lana*. São Paulo: Annablume, Fapesp.

MICHAEL, LEV. 2010. The interaction of stress and tone in the prosodic system of Iquito (Zaparoan). UC Berkeley Phonology Lab Annual Report. Ms.

MICHAEL, LEV. 2011. On the description and analysis of mixed tone-stress systems. *Paper Presented at the Workshop on How to Study a Tone Language: From the First Elicitation to the Latest Software*. University of California, Berkeley, 18-20 February 2011. Organizers: Larry Hyman (host), Steven Bird, Mark Donohue, Mark Liberman.

MILLER, MARION. 1976. Fonologia del Desano. *Sistemas Fonologicos de Idiomas Colombianos* 3, ed. by Timothy Sudo et al., 105-11. Bogota: Ministerio de Gobierno.

MILLER, MARION. 1999. *Desano Grammar*. (Studies in the Languages of Colombia 6). Summer Institute of Linguistics: University of Texas at Austin.

MILLER, JAMES and MARION MILLER. 1973. Desano. *Aspectos de la Cultura Material de Grupos Etnicos de Colombia* 1, ed. by Stanley Schauer, 89-100. Summer Institute of Linguistics. Bogota: Ministerio de Gobierno.

MITHUN, MARIANNE. 1984. The evolution of noun incorporation. Language 60.847-894.

MITHUN, MARIANNE. 1999. *The Languages of Native North America*. (Cambridge Language Surveys). Cambridge: Cambridge University Press.

MOUNTAIN, KATHY. 1978. Listas de palabras Swadesh y Rowe. *Articulos en Linguistica y Campos Afines*, ed. by Martha Anderson and Bonnie Brobston, 1-57. Summer Institute of Linguistics. Bogota: Ministerio de Gobierno.

MOORE, DENNY. 2006. Brazil: Language Situation. *Encyclopedia of Language and Linguistics*, ed. by Keith Brown, 117-27. St. Louis: Elsevier.

OSPINA BOZZI, ANA MARÍA and ELSA GOMEZ-IMBERT. *to appear*. Predicados complejos en el Noroeste Amazónico: el caso de yuhup, tatuyo y barasana. *Cultural and Linguistic Interaction in the Upper Rio Negro Region (Amazonia)*, ed. by Patience Epps and Kristine Stenzel. Rio de Janeiro: Museu do Índio.

PORTNER, PAUL. 2009. *Modality*. (Oxford Surveys in Semantics and Pragmatics). Oxford: Oxford University Press.

PROULX, PAUL. 2003. Review of 'Desano Grammar' by Marion Miller (1999). *International Journal of American Linguistics* 69(1).100-101.

RAMIREZ, HENRI. 1997. *A fala Tukano dos Ye'pâ-Masa: Tomo I: Gramática*. Manaus: Inspetoria Salesiana Missionária da Amazônia: Brazil.

REICHEL-DOLMATOFF, GERARDO. 1968. *Desana: Simbolismo de los Indios Tukanos del Vaupes*. Bogota: Universidad de los Andes. Departamento de Antropologia.

REICHEL-DOLMATOFF, GERARDO. 1971. *Amazonian Cosmos: The Sexual and Religious Symbolism of the Tukanoan Indians*. Chicago: University of Chicago Press.

REICHEL-DOLMATOFF, GERARDO. 1973. Desana: Le Symbolisme Universel des Indiens Tukano du Vaupes. Paris: NRF/Gallimard.

REICHEL-DOLMATOFF, GERARDO. 1976a. Cosmology as ecological analysis: a view from the rain forest. *Man* 11(3).307-18.

REICHEL-DOLMATOFF, GERARDO. 1976b. Desana curing spells: an analysis of some shamanic methaphors. *Journal of Latin American Lore* 2(2). 157-219.

REICHEL-DOLMATOFF, GERARDO. 1978. Desana animal categories, food restrictions, and the concept of color energies. *Journal of Latin American Lore* 4(2). 243-91.

REICHEL-DOLMATOFF, GERARDO. 1979. Desana shamans' rock crystals and the hexagonal universe. *Journal of Latin American Lore* 5(1). 117-28.

REICHEL-DOLMATOFF, GERARDO. 1981. Brain and mind in Desana shamanism. *Journal* of Latin American Lore 7(1). 73-98.

REICHEL-DOLMATOFF, GERARDO. 1987. Algunos conceptos de geografia chamanistica de los indios Desana de Colombia. *Myth and the Imaginary in the New World*, ed. by Edmundo Magana and P. Mason, 75-92. (Latin American Studies 34). Amsterdam: Foris Publication.

REICHEL-DOLMATOFF, GERARDO. 1989. Desana texts and contexts. *Acta Ethnologica et Linguistica*, 62. (Series Americana 12). Vienna.

REICHEL-DOLMATOFF, GERARDO. 1990. Algunos conceptos de los indios Desana del Vaupes sobre manejo ecologico. *La Selva Humanizada: Ecologia Alternativa en el Tropico Humido Colombiano*, ed. by François Correa, 35-41. Bogota: Instituto Colombiano de Antropologia.

REICHEL-DOLMATOFF, GERARDO. 1996. Desana. *Encyclopedia of World Cultures*. *Encyclopedia.com*. 7 Dec. 2010 http://www.encyclopedia.com

RIVET, PAUL; P. KOK and C. TASTEVIN. 1924. Nouvelle contribution à l'étude de la langue Makú. *International Journal of American Linguistics* 3. 133-92.

RODRIGUES, ARYON D. 2002. *Línguas Brasileiras: Para o Conhecimento das Línguas Indígenas*. São Paulo: Edições Loyola.

RODRIGUES, ARYON D. 2005. Sobre as línguas indígenas e sua pesquisa no Brasil. *Ciência e Cultura* 57. 35-8.

SANTOS-GRANERO, FERNANDO. 2002. The Arawakan matrix: ethos, language and history in native South America. *Comparative Arawakan Histories*, ed. by Fernando Santos Granero and Jonathan D. Hill, 25-50. Urbana: University of Illinois Press.

SCHACHTER, PAUL and TIMOTHY SHOPEN. 2007. Parts-of-speech systems. *Language Typology and Syntactic Description*, ed. by Timothy Shopen, 1-60. Second edition, volume I (Clause structure). Cambridge: Cambridge University Press.

SEIFART, FRANK. 2005. The Structure and Use of Shaped-Based Noun Classes in Miraña (Northwest Amazon), Nijmegem: MPI Series in Psycholinguistics.

SILVA, WILSON. *to appear*. The status of the glottal stop and glottal fricative in Desano. *Laryngeal Features in the Languages of the Americas*, ed. by Leo Wetzels, Elsa Gomez-Imbert and Ana Maria Ospina. Brill Publishers.

SILVA, WILSON. 2008. Acoustic analysis of voiceless obstruents and nasal harmony in Desano. *Amerindia* 32.301-19.

SORENSEN, ARTHUR PETER, JR. 1967. Multilingualism in the Northwest Amazon. *American Anthropologist* 69.670-84.

SORENSEN, ARTHUR PETER, JR. 1969. *Morphology of Tucano*. Columbia University: PhD dissertation.

SORENSEN, ARTHUR PETER, JR. 1985. An emerging Tukanoan linguistic regionality: policy pressures. *South American Indian Languages: Retrospect and Prospect*, ed. by Harriet E. Manelis Klein and Louisa R. Stark, 140-56. Austin: University of Texas Press.

STENZEL, KRISTINE. 2004. A Reference Grammar of Wanano. University of Colorado at Bolder: PhD dissertation.

STENZEL, KRISTINE. 2005a. Multilingualism in the Northwest Amazon, revisited. *Memorias del Congreso de Idiomas Indígenas de Latinoamérica-II*, 27-29 de octubre de 2005. University of Texas at Austin.

STENZEL, KRISTINE. 2005b. Review of 'Language Contact in Amazonia,' by Alexandra Y. Aikhenvald. *International Journal of American Linguistics* 71. 505-7.

STENZEL, KRISTINE. 2006. Study on endangered languages and their oral traditions in amazonia. *Lenguas y Tradiciones Orales de la Amazônia: ¿Diversidad en Peligro?* Havana: UNESCO, Casa de las Américas. pp. 71-121.

STENZEL, KRISTINE. 2007. Glottalization and other suprasegmental features in Wanano. *International Journal of American Linguistics* 73(3). 331-66.

STENZEL, KRISTINE. 2008. Kotiria 'differential object marking' in cross-linguistic perspective. *Amerindia*, ed. by Ana Carla Bruno, Frantomé. Pacheco, Francisco. Queixalós and Leo Wetzels, 153-182. (La structure des langues amazoniennes).

STENZEL, KRISTINE, and ELSA GOMEZ-IMBERT. 2008. Contato linguístico e mudança linguística no noroeste amazônico: o caso do Kotiria (Wanano). *Revista da ABRALIN* 8(2).71-100.

STRADELLI, ERMANO. 1910. Pequnos vocabulários, grupo de línguas tucanas. *3a Reunião do Congresso Científico Latino-Americano*, 254-317. Rio de Janeiro.

TORAMU and GUAHARI (Wenceslau Galvão and Raimundo Galvão). 2004. *Livro dos Antigos Desana – Guahari Diputiro Pora*. (Coleção narradores indígenas do Rio Negro, vol. 7). São Gabriel da Cachoeira. ONIMRP/FOIRN.

WALLACE, ALFRED RUSSEL. 1858. A Narrative of Travels on the Amazon and Rio Negro: With an Account of Native Tribes. Ward, Lock and Co., London.

WALTZ, NATHAN and ALVA WHEELER. 1972. Proto-Tucanoan. *Comparative Studies in Amerindian Languages*, ed. by Esther Matteson, 119-49. The Hague: Mouton.

WICHMANN, SOREN. 2002. Book notice on Marion Miller, Desano Grammar. Studies in the Languages of Columbia, 6. SIL International and The University of Texas at Arlington (1999). *Language* 78.4: 804-5.

WILLET, THOMAS. 1988. A cross-linguistic survey of evidentiality. *Studies in Language* 12(1). 51-97.

WRIGHT, ROBIN M. 1987. História indígena: Alto Rio Negro. Levantamento da Situação dos Povos Indígenas do Brasil. (volume I: Noroeste Amazônico). São Paulo: CEDI.

WRIGHT, ROBIN M. 1991. Indian slavery in Northwest Amazon. Séries Antropológicas 7(2). Museu Emílio Goeldi. Belem: Pará.

WRIGHT, ROBIN M. 2002. Prophetic traditions among the Baniwa and other Arawakan peoples of the Northwest Amazon. *Comparative Arawakan Histories*, ed. by Fernando Santos-Granero and Jonathan D. Hill, 269-93. Urbana: University of Illinois Press.